Doctoral theses at NTNU, 2022:367

Marco Simonetto

Design and management of Assembly Systems 4.0

NTNU

Norwegian University of Science and Technology Thesis for the Degree of Philosophiae Doctor Faculty of Engineering Department of Mechanical and Industrial Engineering



Norwegian University of Science and Technology

Marco Simonetto

Design and management of Assembly Systems 4.0

Thesis for the Degree of Philosophiae Doctor

Trondheim, December 2022

Norwegian University of Science and Technology Faculty of Engineering Department of Mechanical and Industrial Engineering



Norwegian University of Science and Technology

NTNU

Norwegian University of Science and Technology

Thesis for the Degree of Philosophiae Doctor

Faculty of Engineering Department of Mechanical and Industrial Engineering

© Marco Simonetto

ISBN 978-82-326-5608-0 (printed ver.) ISBN 978-82-326-6498-6 (electronic ver.) ISSN 1503-8181 (printed ver.) ISSN 2703-8084 (online ver.)

Doctoral theses at NTNU, 2022:367

Printed by NTNU Grafisk senter

Acknowledgements

I owe my deepest gratitude to my supervisor, Professor Fabio Sgarbossa, for giving me the opportunity to perform this PhD and for his great support, encouragement and fruitful discussions during my PhD studies. He provided me the intellectual freedom to work on challenging and exciting topics with wholehearted support. I cannot thank him enough for what he has done for me so far.

I would like to thank my cosupervisor, Erlend Alfnes, for his support and help during my doctoral period.

Furthermore, I would like to thank NTNU and the Department of Mechanical and Industrial Engineering for financing this research study. The opportunity to work full time in research has been a unique experience. I was fortunate to have been given resources and opportunities to develop and foster my research competence.

I would also like to express my gratitude to Professor Jesus Jimenez, for giving me the opportunity to visit Texas State University and collaborate with his research group. Moreover, I would like to thank Abhimanyu Sharotry for all the help during my three-month stay in Texas and for all the good times we had together and with the twins Shashwait and Chaitanya.

During this PhD study, I have had the best colleagues one could hope for: Mirco, Maggie, Vivek, Mina, Giuseppe, Cosmin, Swapnil, Olumide, Jo, Aili, Nicolo', Giulio, Alessandra and Jasashwi. Thank you all for all the lovely memories.

I would also like to express my profound gratitude to Ana, mi novia, for supporting me, believing in me, cheering me and her patience throughout this journey.

Finally, but most importantly, I would like to express my deepest gratitude to my mom, Lidia, my sister, Anna and my brother, Mattia. Thank you!

Trondheim, December 2022

Marco Simonetto

Summary

In recent years, companies have faced great customer demand change involving, for example, a change in colour (aesthetic design), shape (measurement) or technical characteristics of some components (functionality) (Piller, 2005) of the products that companies assemble in their assembly systems (ASs) (Battaïa et al., 2018). Hence, if companies decide to accept those requests from their customers, they must be ready to produce more complex and diverse products (Otto and Li, 2020). The greater the complexity and diversity of the products that companies need to assemble, the greater the chance that ASs will fail to achieve their goals. Here the goal would be to have high efficiency, maximising throughput with minimum resources, to have high flexibility, rapid changes in the production volume and type of products to be assembled, to minimise the work in progress (WIP), assemble high-quality products, minimise the inventory level, utilise the workforce effectively and to minimise the disruptions in production (Bukchin and Masin, 2004; Hu et al., 2011; Vallandingham et al., 2017) and, thus, satisfying the demand. Therefore, companies are required to find ways that can enable them to handle this complex and diverse demand while achieving and maintaining their AS goals. One way might be the implementation of Industry 4.0 (14.0) technologies (e.g., Internet of Things (IoT) technologies, data analysis, cloud computing, collaborative robots (cobots), augmented reality (AR), virtual reality (VR) and mobile robots, etc.) in the ASs, creating the so-called Assembly System 4.0 (AS4.0) (Bortolini et al., 2017; Dolgui et al., 2022). Although I4.0 technologies promise increased flexibility, better quality and improved productivity of ASs (Zhong et al., 2017), if they are not implemented correctly, they may cause risk for the ASs performance. This especially happens when the companies do not know how to design and manage AS4.0. In fact, although companies should know how to design and manage a traditional AS where no I4.0 technologies are implemented (Battini et al., 2011), there are no clear roadmap on how to design and manage AS4.0 (Dolgui et al., 2022). Therefore, the main goal of the present research study is to support and create new knowledge for the academy and managers and practitioners who would like to design and manage AS4.0.

We start the current research by conducting a systematic literature review (SLR). The SLR work give the opportunity of not only understanding the state of the art of AS4.0, but also the opportunity of identifying the nine future research opportunities concern the introduction of

methods ("a method is a systematic approach to achieve a specific result or goal and offers a description in a cohesive and (scientific) consistent way of the approach that leads to the desired result/ goal." (Verbrugge, 2019)) and models ("a model is the presentation in schematic form, often in a simplified way, of an existing or future state or situation" (Verbrugge, 2019)) for the design and management of AS4.0. Examples of methods are descriptive, rational, and experimental (e.g., observations, framework, and decision support systems), while example of models are graphs, flow charts, 3D models, diagrams, and equations. The nine future research opportunities are models of dynamic reconfigurable AS4.0; models to support the selection of a suitable level of automation; models for dynamic assignment of technologies on AS4.0 configurations; methods for ergo-efficient workplace design; models of new feeding policies; models for real-time multiobjective balancing of assembly line subsystem (ALS) and the scheduling of assembly line feeding subsystem (ALFS); smart and real-time methods for sequencing of ALS and routing of ALFS; methods for efficient control of AS4.0; and models and methods for maintenance of technologies (Dolgui et al., 2022). Based on three of those future research opportunities (i.e., models of dynamic reconfigurable AS4.0, methods for ergo-efficient workplace design and methods for efficient control of AS4.0), we derive the below three research questions:

- RQ1: How can AS4.0 configurations be affected by Industry 4.0 technologies?
- RQ2: How can ergo-efficient AS4.0 workplaces be designed?
- RQ3: How can material management be controlled in AS4.0?

To answer these research questions, the below methods are applied:

- RQ1: exploratory research and experimental research
- RQ2: experimental research
- RQ3: simulation modelling

The main outcomes are derived by answering the three research questions by using those methods:

• RQ1:

Factors for the modelling of dynamic reconfigurable configurations of an Assembly Line Subsystem and the impact of Industry 4.0 technologies on these factors

- Identification and definition of the factors relevant for modelling dynamic reconfigurable configurations of an ALS and identification of how Industry 4.0 technologies impact those factors.
- o Investigation of the impact of augmented reality on the identified factors.
- RQ2:

A framework to design the workplace of an Assembly Line Subsystem by using Virtual Reality and a motion capture system

- Identification of the steps to design an ergo-efficient workplace of an ALS by using VR and a motion capture (mocap) system.
- Identification of how the age of human workers can be included in the design of a workplace of an ALS.
- RQ3:

A Decision Support System to select the best material management solution among traditional Kanban, electronic Kanban, and Digital Twin

- Development of a digital twin (DT)-based model for the control of the material management in an ALFS.
- Identification of the best solution among traditional Kanban, electronic
 Kanban and DT to control the material management in an ALFS.

Finally, by the outcome of the SLR and research questions, we reach our main goals that are: "support and create new knowledge for academy and managers and practitioners who would like to design and manage AS4.0". In fact, by the SLR work, we aim to present the state of the art and possible future research opportunities related to the design and management of AS4.0. Also, we aim to support companies implementing I4.0 technologies in their ASs, showing them the decisions that they have to make, level by level (strategic, tactical and operational), for each technology if they want to achieve and maintain the goals of the ASs. In answering RQ1, we aim to provide companies information during the design phase of their

ASs, explaining to them the impact that each I4.0 technology has on those factors relevant to the decision of which AS configuration to choose. With the answer to RQ2, we aim to encourage companies to use VR when designing the workplaces in their ASs. In fact, they can use our five-step methodological framework, which considers the age of the operators in designing ergo-efficient AS workplaces, here by using VR together with a motion capture system. Finally, based on the answer to RQ3, we aim to support companies in deciding which solution to choose to control the material management activity in their ASs. In fact, we provide a decision support system (DSS) by which companies can choose the most convenient material management solution, here based on investment and operational costs, among traditional Kanban, e-Kanban and DT-based solutions, by considering the demand and layout parameters of their ASs.

Abbreviations

- AM Additive manufacturing
- ALFS Assembly line feeding subsystem
- ALS Assembly line subsystem
- AS Assembly system
- AS4.0 Assembly System 4.0
- Cobot Collaborative robot
- DSS Decision support system
- E Kanban Electronic Kanban
- FW Fixed-worker
- 14.0 Industry 4.0
- IoT Internet of Things
- KPIs Key performance indicators
- Mocap Motion capture
- NTNU Norwegian University of Science and Technology
- PAR Projected augmented reality
- SD Standard deviation
- SL Straight line
- SLR Systematic literature review
- US U-shaped
- VR Virtual reality
- WIP Work in progress
- WW Walking worker

List of Appended Papers and Declaration of Authorship

Paper	Title	Declaration of authorship
1	Dolgui, A., Sgarbossa, F. and Simonetto, M., 2022. Design and management of assembly systems 4.0: systematic literature review and research agenda. International Journal of Production Research, 60(1), pp.184-210.	Simonetto conceptualized the paper and conducted the systematic literature review. Simonetto wrote the paper with feedback from Dolgui, and Sgarbossa.
2	Simonetto, M. and Sgarbossa, F., 2020, August. Introduction to Material Feeding 4.0: Strategic, Tactical, and Operational Impact. In IFIP International Conference on Advances in Production Management Systems (pp. 158-166). Springer, Cham.	Simonetto conceptualized the paper and conducted the literature review. Simonetto wrote the paper with feedback from Sgarbossa.
3	Simonetto, M. and Sgarbossa, F., 2021, September. Straight and U-Shaped Assembly Lines in Industry 4.0 Era: Factors Influencing Their Implementation. In IFIP International Conference on Advances in Production Management Systems (pp. 414-422). Springer, Cham.	Simonetto conceptualized the paper and conducted the exploratory research. Simonetto wrote the paper with feedback from Sgarbossa.
4	Simonetto, M., Peron, M., Fragapane, G. and Sgarbossa, F., 2020, October. Digital assembly assistance system in Industry 4.0 era: a case study with projected augmented reality. In International Workshop of Advanced Manufacturing and Automation (pp. 644-651). Springer, Singapore.	Simonetto conceptualized the paper. Simonetto and Peron collected the data and Fragapane analyzed the data with input from Simonetto and Peron. Simonetto, Peron, and Fragapane wrote the paper with feedback from Sgarbossa.
5	Simonetto, M., Arena, S. and Peron, M., 2022. A methodological framework to integrate motion capture system and Virtual Reality for assembly system 4.0 workplace design. Safety Science, 146, p.105561.	Simonetto conceptualized the paper and created the methodological framework together with Peron. Simonetto and Peron collected the data and Peron analyzed the data with input from Simonetto. Simonetto and Peron wrote the paper with feedback from Sgarbossa.
6	Simonetto, M., Saporiti, N., (under review). Traditional Kanban, e-Kanban or Digital Twin: a Decision Support System for material management solution. Industrial Management & Data Systems.	Simonetto conceptualized the paper together with Saporiti. Simonetto and Saporiti created the simulations models and analyzed the data. Simonetto and Saporiti wrote the paper.

Summary	iii
Part I: Main report	1
1. Introduction	3
1.1. Background	3
1.2. Research motivation	7
1.3. Research scope	8
1.4. Research objectives and questions	9
1.5. Thesis outline	11
2. Theoretical background	15
2.1. Assembly Systems	15
2.2. Industry 4.0 technologies	18
2.3. Assembly Systems 4.0	20
3. Research design	25
3.1. Research methods	25
3.2. Research quality	32
4. Design and management of Assembly Systems 4.0	35
4.1. State of the art of Assembly Systems 4.0 and future research opportunities	36
4.2. Factors affecting the configurations of Assembly Line Subsystems with Industry 4.0 technologies) 52
4.3. Framework to design the workplace of an assembly line subsystem by using virtual reality and motion capture system	l 62
4.4 Decision support system for the selection of the best material management solutio	n
among traditional Kanban, electronic Kanban and digital twin	71
4.5. General discussion of the results	82
5. Conclusions	85
5.1. Summary	85
5.2. Contributions to theory	88
5.2. Implications for practice	89
5.3. Research limitations	91
5.4. Future research	92
References	95
Part II: Collection of papers	111

Contents

Part I: Main report

1. Introduction

This section discusses the context and motivation for the current research investigation. Therefore, first, as the background, the main topics of the research study—assembly system (AS) and Industry 4.0—are introduced to familiarise readers with the terms that will be used throughout the rest of the study. Second, the main motivation to conduct this research study is explained. Next, the scope of the current study clarifies and positions the research study within the literature to then present the objectives of the study and derived research questions. Finally, to guide readers who may be interested solely in a specific part of the work, a summarised version of the content of each section and the structure of the present research study are presented.

1.1. Background

Assembly is an essential part of the manufacturing process that is performed in what are called ASs. Generally, an AS is composed of two subsystems: the assembly line subsystem (ALS) and the assembly line feeding subsystem (ALFS) (Battini et al., 2009). In each subsystem, different activities are performed to make sure that the goals of the AS are achieved and maintained (e.g., to have high efficiency, that is, by maximising throughput with minimum resources, having high flexibility, having rapid changes in the production volume and type of products to be assembled, minimising the work in progress (WIP), assembling high-quality products, minimising the inventory level, utilising the workforce effectively and minimising disruption to production (Bukchin and Masin, 2004; Hu et al., 2011; Vallandingham et al., 2017). Hence, the activities of an ALS are assembly and control, while the activities of an ALFS are transportation, preparation and material management (Battini et al., 2009; Sali and Sahin, 2016; Schmid and Limère, 2019). Depending on whether these activities are done manually, automatically or both, an AS can be manual, automatic or hybrid (Levitin et al., 2006; Gil-Vilda et al., 2017; Weckenborg et al., 2019; Zhang et al., 2019).

The costs of an AS is typically between 25% and 50% of the total cost of manufacturing, with the percentage of operators involved in their execution ranging from 20% to 60% (Ritchie et al., 1999; Bi et al., 2007; Buckhorst et al., 2022). If the companies aim to remain competitive in the market, they should try to keep the costs of their ASs as low as possible. In fact, regardless of the goals companies want to achieve regarding their AS, achieving high profits

while keeping costs low is considered one of the most important overall goals for most companies. To achieve this, companies would require to learn how to design and manage their ASs (Battini et al., 2011).

The design of an AS, as the word may suggest, concerns all the decisions to be made, at a strategic and a tactical level, with the aim of designing and making ASs operational to reach the desired goals (Battini et al., 2011). Therefore, when companies have to design their ASs, they first need to know the products that they are going to assemble and resources necessary to assemble them. The characteristics of the products and equipment will influence the decision about the configuration and level of the automation of the ASs. The configuration of an AS is related to how the workplaces where the resources will perform the assembly tasks will be arranged in the space allocated for the creation of the AS. The most widely adopted configuration is the straight-line (SL) configuration, followed by the U-shaped (US) line configuration (Bagher et al., 2011; Rabbani et al., 2012; Mukund and Ponnambalam, 2016). The SL configuration is represented by a sequence of workplaces in a line. By simply rearranging the workplaces in the SL into a U-shape, it is possible to obtain the US line configuration. Other examples of configurations that can all be reconfigured or treated as a line in terms of how they operate are two-sided lines, systems with rotary tables, the SL with fixed workers (FW), the US line with walking workers (WW) and so on (Battini et al., 2011; Calzavara et al., 2021); here, the level of automation refers to whether the AS is manual, automatic or hybrid. Next, the workplaces of the ASs need to be designed, along with the assembly tasks assigned to the workplaces, hence balancing the work that will be performed in each of them. Finally, it is necessary to choose the feeding policies necessary to transport the components and then schedule the transportation devices to transport these components to the ASs.

Once the design is complete, the ASs need to be managed to meet the established goals and be cost-efficient to operate, especially in terms of their adaptability to changing product and volume requirements triggered by a turbulent market environment (Kuzgunkaya and ElMaraghy, 2006; Matt, 2013). In fact, the management of an AS is related to all those activities responsible, at a tactical and operational level, for planning the short period of the AS and controlling the AS so that it can operate while keeping throughout its life the performance to reach the goals for which it was designed (Gordon et al., 2002; Gao et al.,

4

2012). The main decisions during the management of an AS are the planning of the sequence of products that need to be assembled day by day, the planning of the routes that transportation devices must take to transport the components to assemble those products, the planning of the quantity of components and the type of information that need to be managed in the ASs, along with how to control the entire AS so that everything works. If something is not working properly, the management of an AS must intervene so that any problems can be resolved as soon as possible. Therefore, the management of an AS has the important role of making the entire AS more flexible and reactive by ensuring that the decisions made during the design of the AS, such as workforce balancing, feeding policies and transportation scheduling, are properly processed, which can be done by changing the sequence of products to be produced or the routes of the transportation devices and by training and monitoring the resources of the ASs so that they can handle the demand for a higher quantity and variety (MacDuffie et al., 1996; Colledani et al., 2016).

Therefore, based on the above definitions, during the design and management of an AS, managers and practitioners must make different decisions (Wänström and Medbo, 2009; Battini et al., 2011; Fragapane et al., 2021; Dolgui et al., 2022), such as deciding on the configuration of the system, the level of automation, workplace design, assembly line balancing, sequencing, the feeding policy selection, scheduling, routing and material/information management and control of the AS.

However, although companies make the best decisions for their ASs, nowadays, it is more complicated than ever for ASs to achieve their goals. In fact, because of the rapid growth in customer demands for products with greater variety and shorter life cycles, AS activities are becoming extremely complicated, resulting in time losses, human errors and other negative effects on system performance (Battaïa et al., 2018; Bläsing et al., 2020). Therefore, to cope with these negative effects, companies have started to implement Industry 4.0 (I4.0) technologies in their ASs, creating the so-called Assembly System 4.0 (AS4.0) (Bortolini et al., 2017, Dolgui et al., 2022). According to the literature, Industry 4.0 technologies include Internet of Things (IoT) technologies, cyber-physical systems (CPSs), data analysis, cloud computing, collaborative robots (cobots), augmented reality (AR), virtual reality (VR), mobile robots, cybersecurity, blockchain and additive manufacturing (AM) (Stock and Seliger, 2016; Dalenogare et al., 2018; Ghobakhloo, 2018). These technologies can be implemented in the

ASs with the objective of reducing the set-up and processing times, as well as reducing the labour and production costs (Brettel et al., 2014; Jeschke et al., 2017). In fact, thanks to I4.0 technologies, ASs could become up to 35% faster and 30% more efficient (Nokia, 2022). Moreover, other similar facts have been reported by MIT, PWC and Deloitte. In fact, an MIT study has shown that I4.0 technologies help cut employee downtime by 85% and increase the average productivity rates by 11% (Globalluxsoft, 2017). A PWC study on the impact of I4.0 technologies on business instead forecasts a 18% increase in average productivity within 5 years of the mass implementation of these technologies in the ASs because of the ability to predict and minimise equipment downtime, thus optimising the operational efficiency (Globalluxsoft, 2017). Finally, Deloitte has reported that, by using I4.0 technologies, ASs can reduce the time needed to plan and execute the asset maintenance by up to 50%, resulting in 20% longer equipment uptime and up to 10% lower asset and total expenses, which can save millions in the long run (Coleman et al., 2017). Furthermore, more benefits can be derived from the increase in production line availability by 5% to 15% and the opportunities for energy-saving through optimisation. Indeed, for example, in a case study of a multinational in the plastics sector, I4.0 technologies were found to reduce the power consumption in one of its plants by around 40%, which saved over USD200,000 a year in energy (Sirimanne, 2022). If we look at those countries that have so far invested the most in 14.0 technologies, that is, the US and China, and in general at the worldwide investment in 14.0 technologies, the estimated value was at USD 71.7 billion in 2019 and is expected to reach USD 156.6 billion by 2024 (Globalluxsoft, 2017). Considering the creation of job opportunities, Lorenz et al. (2015) explained that, in Germany, by 2025, the use of I4.0 technologies will add 350,000 jobs, particularly in information technology and data science. Finally, in its 2020 Industry 4.0 study, the consulting company McKinsey & Co discovered that companies employing I4.0 technologies were better able to withstand the effects of the COVID-19 pandemic on business. In fact, 94% of the more than 400 companies polled stated that I4.0 technologies helped them continue operating during the pandemic. Furthermore, 56% of respondents said that I4.0 technologies were crucial to how they dealt with the pandemic (Agrawal et al., 2020; Uzialko, 2022).

Although 14.0 is implemented in the ASs, it is not certain that the benefits that these technologies seem to promise are then automatically achieved. In fact, for technologies to

work correctly and, hence, ensure performance, they must be properly implemented and integrated into the ASs (Bortolini et al., 2019; Cohen et al., 2019a; Cohen et al., 2019b). Therefore, proper models and methods should be followed when implementing technology in an AS. However, current research has focused more on understanding how I4.0 technologies work or on the creation of new applications with the technologies rather than studying how the technologies impact the decision areas related to the design and management of an AS. Consequently, the absence of models and methods can mean that, once implemented, the technologies do not perform as expected, generating a negative impact on the performance of the ASs.

Therefore, the current research study will investigate how the application of I4.0 technologies impacts the decision areas related to the design and management of ASs. In particular, it will start to examine some of the abovementioned decision areas, providing solutions (e.g., a framework and a decision support system (DSS)) that can support managers and practitioners who want to understand, for example, which I4.0 technologies to implement in their ASs, how to use the I4.0 technologies or when to implement I4.0 technologies during the design of their ASs and for managing their ASs.

1.2. Research motivation

The present research study is motivated by the fact that, although companies are implementing the I4.0 technologies in their ASs, creating what is referred to as AS4.0, many of them then still design and manage their ASs as if they had not in fact implemented any of these technologies (Bortolini et al., 2017; Cohen et al., 2017; Cohen et al., 2019b). Therefore, although I4.0 technologies are being implemented to improve productivity, increase flexibility, have higher levels of automatisation, reduce costs and have better quality in the ASs, if they are not properly implemented and not monitored, it is possible that, instead of a benefit, these technologies could turn into a problem for companies and their ASs (Bortolini et al., 2021; Souifi et al., 2022). Because of the high investment that can be required to implement new technology (in terms of the cost of the infrastructure needed to use the technology, acquiring the skills then to use it and buying and maintaining it), if it then does not perform as it was supposed to, the companies may be discouraged from wanting to use it and will invest in other technologies (Agostini and Nosella, 2019; Palominos et al., 2019; Götz et al., 2020). The noninvestment in new technologies may result in losing the

opportunity to benefit from the advantages that technologies could bring to ASs, with the possible consequence of failing to reach their goals (e.g., to minimise the WIP, assemble highquality products, minimise the inventory level, utilise the workforce effectively and minimise disruption to production).

In general, the researchers who first studied I4.0 technologies were interested in learning how I4.0 technologies perform in specific applications, in the development of new technologies or in the development of new applications and functionalities of the technologies (Cohen et al., 2019a; Cohen et al., 2019b; Veiga et al., 2021). Therefore, further studies are needed now to understand the impact of I4.0 technologies on the design and management of ASs; these studies should create new models, methods and frameworks to understand, for example, which I4.0 technology to use, how to apply an I4.0 technology and when to use an I4.0 technology during the design of an AS and to manage the AS.

Moreover, the current research has aimed to enhance the investment that the Department of Mechanical and Industrial Engineering of NTNU is making by providing some of the I4.0 technologies to the Logistics 4.0 Lab, which is Norway's first logistics laboratory and merges I4.0 technologies with traditional production and logistics systems, enabling researchers, practitioners, engineers, pioneers, students and other enthusiasts to come together and collaborate on the common ground; this can help in developing and improving their knowledge and skills in the use of the technologies. Many I4.0 technologies such as AR, VR, IoT and mobile robots can be found in the Logistics 4.0 Lab. Here, the basic idea was to use one or more of these technologies in collaboration with the production management group to answer the research questions of this study.

1.3. Research scope

The current research study lies within the research area of ASs and examines the impact of I4.0 technologies on the decision areas related to the design and management of ASs. However, a complete discussion of the topics introduced so far is too extensive a task for the present research. Thus, further scoping of the domains is needed.

As shown in the background, an AS can be manual, automatic or hybrid, according to whether the AS activities are performed manually, by robots or both manually and by robots. However, because performing assembly tasks still requires intensive manual activity (Bi et al., 2007), the present research focuses on manual and hybrid ASs of small-to-medium sized products such as hydraulic pumps, engines, actuators, hose reels and so forth. Moreover, because investigating all the decision areas related to the design and management of an AS would exceed the time frame of the current study, only three decision areas have been considered. In fact, the present study focused on how to configure the ALS, the workplace design of an ALS and the control of the ALFS. In particular, for the control of the ALFS, the focuses was on the control of the material management of the ALFS.

Regarding I4.0 technologies, it is known that there are many available technologies that can be studied in an AS. However, the Logistics 4.0 Lab at the Department of Mechanical and Industrial Engineering at the Norwegian University of Science and Technology (NTNU) came to our aid. In fact, the present research study was conducted with the idea of exploiting the technologies that were already available, without having to purchase new ones. Therefore, in this sense, the Logistics 4.0 Lab limited the possible I4.0 technologies to be studied to only those available internally. In addition, among these technologies were selected those that could help investigate the three decision areas configuration of an ALS, workplace design of an ALS and control of the ALS. Therefore, from the I4.0 technologies available in the Logistics 4.0 Lab, AR, VR and digital twin (DT) technologies were selected and analysed.

The scope of the research study is summarised as follows:

The current study focuses on three decision areas (configuration of an ALS, workplace design of an ALS and control of the ALS) related to the design and management of a manual/hybrid AS4.0 that assembles small/medium products. During the study of the three decision areas, we considered the three I4.0 technologies of AR, VR and DT.

1.4. Research objectives and questions

Motivated by the challenges and research problem and following the research scope outlined above, the current study aims to support and create new knowledge for the academy and managers and practitioners who want to design and manage AS4.0.

To do so, the first main goal was to understand the state of the art of AS4.0 and propose possible future research opportunities in AS4.0 that represent, respectively, the first two main outcomes of the current study. Therefore, we performed a systematic literature review (SLR). From this SLR, we understood how I4.0 technologies impact the decision areas of an AS and

derived nine future research opportunities that can generate many research questions related to the design and management of AS4.0.

However, because there was no time to explore all the possible future research opportunities and unanswered questions regarding this topic, the objective is limited to providing a better understanding and knowledge of the following:

- The factors that are relevant for modelling dynamic reconfigurable configurations of an ALS and the impact of I4.0 technologies on these factors
- The use of VR, together with a motion capture system to design the workplace of an ALS
- The use of DT for the control of the material management in an ALFS

Therefore, from the research objectives, the following research questions were defined to guide the research process:

RQ1: How can AS4.0 configurations be affected by Industry 4.0 technologies?

The configuration of an ALS is related to how the workplaces where the resources that will perform the assembly tasks will be arranged in the space allocated for the creation of the AS. Nowadays, due to the change in the market demand companies should model dynamic reconfigurable configurations for their ALSs that can help them assembly different products day by day without hindering the performance of their ALSs. The modelling of the configuration of the ALSs should not be made randomly, but it should be based on some factors that companies need to know if they want to make the best decision. These factors are the ones that not only determine the configurations of the ALSs but can also determine how dynamic reconfigurable can be these configurations. In addition, the implementation of I4.0 technologies in ALSs can have an impact on these factors thus facilitating the design of dynamic reconfigurable configurations of ALSs. However, at the moment, no studies have investigated this topic. Therefore, the first research question aims to discover and define the factors that are relevant for modelling dynamic reconfigurable configurations determine these factors. In addition, to validate our results, we analysed the impact of AR on the discovered factors.

RQ2: How can ergo-efficient AS4.0 workplaces be designed?

The second research question aims to create a framework for guiding the designer of ALSs to create ergo-efficient workplaces using VR together with a motion capture system. The framework consists of five steps that are thoroughly explained, leaving no doubt in the minds of the designers who will use it to design the workplaces in their ALSs. In addition, we integrated in the framework the age of the operators working in the workplace in two different modes: a dynamic mode that considers age in terms of operators' experience in the workplace and a static mode that considers age in terms of two formulas, one calculating the age-based multiplier for muscle strength and the other the age-based multiplier for reduced joint flexibility.

RQ3: How can material management be controlled in AS4.0?

The third research question aims to introduce a DSS that can guide managers and practitioners in deciding whether to adopt a DT solution to control the material management in their ALFS compared with two other solutions that are commonly adopted in this regard (traditional Kanban and e-Kanban). In addition, the DSS can also guide managers and operators to change the current solution they have for controlling materials management in their ALFS.

1.5. Thesis outline

The present thesis is divided into two parts. The first part (I) contains the main report, while the second part (II) includes the collection of published papers from this thesis. Hence the thesis report is developed on the research carried out and described in the supplementary papers. It also provides an overview of the research process and synthesises on the contributions of the independent publications into a coherent argument.

Part I is organised as follows:

Section 1 is the introduction part. It describes the issues encountered in practice and the reason for research in this area. Furthermore, it describes the research problem investigated and defines the research objectives and questions addressed through the current study. The section finishes by explaining the scope and structure of the study.

11

Section 2 presents the theoretical background of the research. It starts with the definition of the AS and of its subsystems, explaining their decision areas at the strategic, tactical and operational levels. Next, the most common I4.0 technologies are introduced and defined. To conclude the section, together with the definition of Assembly System 4.0, we introduce some of the benefits that, according to the literature, I4.0 technologies can bring to AS.

Section 3 introduces the research design. It provides a full description of the research methodologies employed, as well as thoughts on key methodology-related decisions made over the course of the research. Finally, the topic of research quality is examined in terms of the four accepted requirements of research quality.

Section 4 presents and discusses the results and findings of the study. It presents the key outcomes addressed by the research questions and discusses them. The first subsection explains the SLR done to study the state of the art of AS4.0, which gave, as an output, the future research opportunities that inspired the three research questions. In the rest of the subsections, we then answered the three research questions, one subsection per question.

Section 5 outlines the research study and provides the final remarks. Furthermore, the limitations of the research are noted, as well as recommendations for further research.

Figure 1 illustrates the outline of the thesis and compares it with the commonly used introduction, methods, results, discussion and conclusion (IMRaD) structure.



Figure 1. Thesis outline

Part II contains the supplementary papers, which are the papers created to communicate the findings of the current research study. It consists of five published papers (Papers 1 to 5) and one paper that is under the review process (Paper 6):

- Dolgui, A., Sgarbossa, F. and Simonetto, M., 2022. Design and management of assembly systems 4.0: systematic literature review and research agenda. International Journal of Production Research, 60(1), pp.184–210. (*Paper 1*).
- Simonetto, M. and Sgarbossa, F., 2020, August. Introduction to material feeding 4.0: Strategic, tactical, and operational impact. In IFIP International Conference on Advances in Production Management Systems (pp. 158–166). Springer, Cham. (*Paper* 2).
- Simonetto, M. and Sgarbossa, F., 2021, September. Straight and U-shaped assembly lines in Industry 4.0 Era: Factors influencing their implementation. In IFIP International Conference on Advances in Production Management Systems (pp. 414–422). Springer, Cham. (*Paper 3*).
- Simonetto, M., Peron, M., Fragapane, G. and Sgarbossa, F., 2020, October. Digital assembly assistance system in Industry 4.0 era: A case study with projected augmented reality. In International Workshop of Advanced Manufacturing and Automation (pp. 644–651). Springer, Singapore. (*Paper 4*).
- Simonetto, M., Arena, S. and Peron, M., 2022. A methodological framework to integrate motion capture system and virtual reality for assembly system 4.0 workplace design. Safety Science, 146, p.105561. (*Paper 5*).
- Simonetto, M., Saporiti, N., (under review). Traditional Kanban, e-Kanban or Digital Twin: A Decision Support System for material management solution. Computers & Industrial Engineering. (*Paper 6*).

2. Theoretical background

This section introduces the relevant theoretical background that frames and supports the current research study. Here we present, the topics of AS (Subsection 2.1), Industry 4.0 technologies (Subsection 2.2) and Assembly System 4.0 (Subsection 2.3). In Subsection 2.1, we present the traditional ASs in which no 14.0 technologies are implemented. In this subsection, we explain the parts that form an AS with the activities performed in each part. Then, we illustrate the decision areas at three levels—strategic, tactical and operational—that companies have to know while designing and managing traditional ASs. Second, in Subsection 2.2, we present the topic of 14.0 by explaining what it is and which 14.0 technologies managers and practitioners can implement in their companies. Finally, in Subsection 2.3, we present the AS4.0 by discussing what 14.0 technologies can be implemented in it.

2.1. Assembly Systems

An AS is composed of the following two parts (Battini et al., 2009):

- The ALS
- The ALFS

An ALS is a type of production system in which various tasks are executed at one or more workplaces to create the final product (Rekiek et al., 2002; Dolgui and Proth, 2010; Akpinar et al., 2017; Zhong and Ai, 2017). The main components of an ALS are the workplaces in which the operators perform the tasks, and the components can be stored. Based on how workplaces are placed within the space designated for the ALS, ALSs can be designed in different configurations, such as a single line (SL), a two-sided line, a US line, a system with rotary table, a SL with FW, a US line with WW and so on (Becker and Scholl, 2006; Battini et al., 2011; Battaïa and Dolgui, 2013). However, the SL is the most frequently utilised configurations can be reconfigured into or treated as a line in terms of how they operate. Moreover, ALSs can be manual, automatic or hybrid, here based on whether the assembly activity is performed only by human workers, only by robots or by both, in which human workers are supported by robots (Levitin et al., 2006; Gil-Vilda et al., 2017; Weckenborg et

al., 2019; Zhang et al., 2019). As reported in Section 1, we are interested in manual and hybrid ASs.

Two main activities are executed in an ALS: assembly and control. Assembly is the capstone process for product realisation, where the components and subassemblies are combined to make the finished products (Hu et al., 2011). Assembly encompasses all the assembly and subassembly processes and equipment required to (i) bring together, configure, align, orient and adjust components and materials to form the end-product and to (ii) physically attach parts, materials and components, such as screwing, riveting, stapling, nailing, gluing, wrapping, interlocking, tying, fusing, sewing, welding, soldering, bonding, pegging, coupling, laminating, insertion, sealing and similar activities (Bi et al., 2007). Control, instead, is related to monitoring the quality and performance of the ALS (Hu et al., 2011; Battini et al., 2011).

An ALFS, instead, is responsible for the transportation, preparation and management of all the components to the ALS that are needed to perform the assembly process (Battini et al., 2009; Sali and Sahin, 2016). The main components of an ALFS are the warehouses in which the components are stored, all the containers of the components and all the devices and operators that are responsible for managing and transporting the components. The right component must be delivered in the right quantity, in the right container and at the right moment to the right ALS.

To achieve this, the three main activities of an ALFS are: transportation, preparation and material management (Schmid and Limère, 2019). Transportation involves the process of moving all the components or parts from point A, where they are stored, to point B, where they are needed, while preparation relates to the processes of handling and repacking parts into the load carriers used for the corresponding line feeding policy (Battini et al., 2009; Zuin et al., 2018; Adenipekun et al., 2022). Finally, material management includes all the processes related to the storage of components and products (Battini et al., 2010).

To provide an AS that can achieve its goals, the two subsystems must be correctly designed and managed. To do this, numerous distinct decision areas have been established in the literature, as summarised in Table 1.

16

Table 1. Decision areas at different levels of a traditional AS

		Assembly Line Subsystem (ALS)	Assembly Line Feeding Subsystem (ALFS)
Long term	STRATEGIC	Configuration of the systemLevel of automation	Configuration of the systemLevel of automation
Medium term	TACTICAL	 Workplace design Assembly line balancing 	 Feeding policy selection Scheduling
hort term	OPERATIONAL	SequencingControl of the ALS	 Routing and material/information management Control of the ALFS

These decision areas may relate to various levels, such as strategic, tactical or operational. Strategic decisions have a long-term (measured in years) influence on the company's operations, whereas tactical decisions have a medium-term (weekly) impact, and operational decisions are made daily and have a short-term impact.

At the strategic level, decisions are made regarding the configuration of the system and level of automation for both ALSs and ALFSs. This means that companies must select the product family to be assembled and estimate how many resources the family will require, how many ASs will be required (e.g., one flexible system for the entire product family or several more dedicated ones), the shape of their ASs and the type of equipment, that is, whether the various tasks associated with these systems will be executed manually, automatically or both (Bassan et al., 1980; Cormier and Gunn, 1992; Roodbergen and Vis, 2006; Wänström and Medbo, 2009; Battini et al., 2011; Fragapane et al., 2021).

At the tactical level, companies may first carry out workforce dimensioning (Dolgui et al., 2018), determining how many workers are necessary to perform the work in the worst-case scenario (large demand, complex product). Then, this information—together with time-and-motion and ergonomic analyses—can be used for designing workplaces for the ALSs before assigning these tasks to the workplaces using assembly line balancing algorithms (Lindenmeyer, 2001; Boysen et al., 2008; Battaïa and Dolgui, 2013; Zülch and Zülch, 2017). The decisions at this level in the ALFS first relate to how to transport the various components to the ALS (Battini et al., 2009; Caputo and Pelagagge, 2011; Caputo et al., 2018), workforce

dimensioning and, then, to the scheduling problem of who needs to deliver them (Mei et al., 2005; Dang et al., 2014).

At the operational level, the AS must function properly and, hence, must be controlled. When the product, production or demand changes, there may be reconfiguration issues. To decide and regulate this process, several types of data are collected from the AS, such as the sequence of goods produced in the ALS, the quantities of components to store in various warehouses and the routes utilised to transport these components to the ALS (DeCroix and Zipkin, 2005; Hu et al., 2008; Boysen et al., 2011; Gebser et al., 2018).

2.2. Industry 4.0 technologies

Introduced for the first time in 2011 at the Hannover Fair in Germany, Industry 4.0 (I4.0) is the term used to refer to the Fourth Industrial Revolution (Rainer and Alexander, 2014). In fact, after the first three industrial revolutions (mechanisation, electrification and digitalisation), where in a period of nearly 200 years, companies went from mechanical looms driven by steam engines to having programmable logic controllers that enable the use of automation systems, we are now in the 'digitalisation era' thanks to the implementation in the industrial environment of new technologies that can communicate with each other and collect data in real time (Lasi et al., 2014; Rainer and Alexander, 2014; Buer et al., 2018). The connection between various pieces of technology and the possibility to have real-time data promise increased flexibility, higher levels of automatisation, better quality and improved productivity in manufacturing systems such as ASs (Thames and Schaefer, 2016; Zhong et al., 2017). Therefore, Industry 4.0 technologies can impact how products are made and, at the same time, may also have an impact on customers' perceptions of product value because the products will have higher quality, will be produced faster and possibly have a greater variety (de Sousa et al., 2018).

Based on the literature, the technologies responsible for the Fourth Industrial Revolution, known as Industry 4.0 technologies, are the IoT technologies, CPSs, data analysis, cloud computing, collaborative robots (cobots), AR, VR, mobile robots, cybersecurity, blockchain and AM (Stock and Seliger, 2016; Dalenogare et al., 2018; Ghobakhloo, 2018).

The term IoT technologies indicates all the objects, systems and processes that are exchanging data through the Internet (Morlet et al., 2016). In addition to the IoT, the fusion

18

of the physical and the virtual world is a further important component of Industry 4.0 (Lu, 2017); this fusion is made possible by CPS, which are technological systems that integrate cyberspace with physical processes and objects to transform the machines and devices of production and ASs into a network so that real-time data are available (Lee et al., 2015). Once real-time data are available, they need to be processed to create useful information that can help in making decisions, such as the prioritisation of production orders, optimisation of tasks, maintenance requirements and so forth to increase the performance. Data analysis aims at analysing these raw data to extract useful information and convert it into effective knowledge, to improve process understanding and to support decisions (Ge et al., 2017). All the real-time data and information can be stored, managed and shared thanks to the cloud. The National Institute of Standards and Technology (NIST) defined cloud computing as 'a model for enabling ubiquitous, convenient, on-demand network access to a shared pool of configurable computing resources (e.g., networks, servers, storage, applications, and services) that can be rapidly provisioned and released with minimal management effort or service provider interaction' (Mell and Grance, 2009). Cloud computing encompasses both the applications offered as services over the Internet and the hardware and system software in data centres that deliver these services.

Companies are always looking for solutions that can assist their operators in the execution of different activities. These solutions are collaborative robots (cobots), AR, VR and mobile robots. Cobots are powerful devices that actively cooperate with the operators during the execution of specific tasks, providing a powerful source of automation and assistance for specific activities (Fast-Berglund et al., 2016). AR refers to the integration of additional computer-generated information into a real-world environment (Paelke, 2014), while VR allows for the creation of a three-dimensional world in which users can interact with three-dimensional objects in real time by using their natural senses and skills (Riva, 2002). Mobile robots can be defined as 'industrial robots that use a decentralised decision-making process for collision-free navigation to provide a platform for material handling, collaborative activities, and full services within a bounded area' (Fragapane et al., 2021).

Today, a huge number of 'things' can be connected through the Internet or among themselves to create a fully interconnected industrial networked environment, that is, a cyberspace, across companies. Therefore, it is necessary to ensure secure, safe and reliable

19

communication so that any decisions and actions made in the cyberspace by companies are based on dependable and properly authorised information (Mehnen et al., 2017; Ghobakhloo, 2018). To do so, companies can rely on cybersecurity, which aims at protecting the cyberspace (which includes both information and infrastructures) from any cyber-threat or -attack (Lezzi et al., 2018).

Blockchain is defined as 'a decentralized database technology that is tamper-proof and ensures consistent transactions across many users' (Yetis et al., 2022). Blockchain may be used as a ledger to establish trustworthy and autonomous relationships between various components of companies, suppliers and even customers (Ghobakhloo, 2018). In fact, with consensus, immutability, security and smart contracts, blockchain improves the efficiency of the systems in which it is used (Ouyang et al., 2019).

Finally, AM is the 'process of joining materials to make objects from 3D model data, usually layer upon layer' (ASTM 2010). It is considered an essential ingredient in the new paradigm of I4.0 (Dilberoglu, 2017) and is known also as three-dimensional printing (3D printing).

2.3. Assembly Systems 4.0

The combination of AS with I4.0 technologies has led to what is referred to as Assembly System 4.0 (AS4.0) (Bortolini et al., 2017; Cohen et al., 2019b). Therefore, as in an AS, in an AS4.0, the ALS and ALFS will be present, but in this case, their activities, are shown in Subsection 2.1, can be supported or executed by one or more I4.0 technologies (Figure 2). However, not all the I4.0 technologies described in the previous subsection are used in ASs. In fact, the I4.0 technologies that are used in AS4.0 are IoT technologies, CPSs, data analysis, cloud computing, cobots, AR, VR, mobile robots and AM (Bortolini et al., 2017; Cohen et al., 2019a; Cohen et al., 2019b).


Figure 2. The parts of an Assembly System 4.0

In fact, IoT technologies can help companies collect data from all the activities and actors in an AS, for which knowledge of their status and behaviour can be useful to keep the AS under control and provide opportunities to improve its performance (Alavian et al., 2020; Baumann et al., 2020). Hence, in an ALS, sensors can be applied to operators (e.g., motion capture systems to collect over the time data of the position and orientation of the operator's different limbs in a common reference system (Oyekan et al., 2017)), workplaces and in all the equipment used during the execution of the various assembly tasks (Oyekan et al., 2019; Peruzzini et al., 2020). For the ALFS, sensors can be applied again to operators, to containers used to store components and to all the transportation devices, such as forklifts, trolleys, mobile robots and so forth used to move the components within the AS (Chien et al., 2017; Zhao et al., 2018). The data that can be collected can be related, for example, to the status of the assembly activities, the status of the products that are being assembled, the location of the components in the AS, the number of components in the containers in the AS and environmental factors such as the temperature inside the AS. In addition, the connection of different components of the AS can create a CPS of the AS giving the connected actors the opportunity to communicate with each other and have a better understanding of what is happening not only in the specific position of the AS in which they are located, but also throughout the entire AS (Sgarbossa et al., 2020). This can help in reducing the mistakes made during the execution of different activities, improve the quality and reliability of the activities and create ASs that are synchronised with the requirements of the customers and the capacity of the AS. However, to achieve all these benefits, it is necessary that the data collected from the IoT technologies are saved and analysed to be transformed into useful

information before then being delivered to the final users. Data analysis is responsible for analysing the data and transforming them into information, such as the amount of components to be stored in warehouses, when it is time to deliver the components, the number of products assembled in a certain period of time, the time when it is necessary to start assembling a new product, which products can be assembled based on the current components stored in the warehouse and so forth (Storti et al., 2018; Yao et al., 2018; Adams et al., 2021; Zangaro et al., 2021). There are no limits on the information that can be generated; the important thing is to understand what information is relevant for the AS (Cattaneo et al., 2018). To send this information in their ASs, companies can decide to use cloud computing, which is not only a 'place' where data from the AS can be saved, but also providing the opportunity to synchronise all of the actors in the AS with the same information in real time thanks to the possibility of executing cloud computing apps on practically any device, including PCs, laptops and smartphones. Moreover, the cloud computing apps can also give the opportunity to do data analysis directly from the cloud (Alexopoulos et al., 2016; Guo et al., 2020). Therefore, with cloud computing, the performance of the ASs can be watched and managed everywhere around the AS and in real time, with the option of taking action if something is not operating properly or needs to be changed (Nuzzi et al., 2020; Ruppert and Abonyi, 2020; Tao et al., 2020).

By introducing workplaces with collaborative task execution by human workers and cobots, the advantages of both automated and manual ALS can be realised in a hybrid ALS (Weckenborg et al., 2020). Moreover, cobots can be installed on top of mobile robots and can be used in warehouses to help operators pick up and place components in containers (Fager et al., 2020; Fager et al., 2021). Regardless of the application in which the cobots are used, it is important that adequate control systems are designed and implemented to avoid possible collisions with the operators (Malik and Bilberg, 2019). In this regard, AR and VR can give valuable help in preventing collisions. In fact, in real time, AR can show operators who are working together with cobots the areas in which the cobots will be active, enabling them to avoid being in that position and, thus, avoiding potential collisions (San Martín and Kildal, 2019; Gruenefeld et al., 2020; Lee et al., 2021). VR, on the other hand, can be used to recreate the area where the operators will work together with cobots in a virtual environment. Through recreating a high-fidelity copy of the area, even operators who have no previous

experience with cobots will have the opportunity to learn how to work together with them before doing so for real in the AS (Brough et al., 2007; Etzi et al., 2019). Moreover, AR and VR can be used for other applications in the AS (Ong et al., 2008; Seth et al., 2011; Ong and Nee, 2013). For example, AR can be used to guide the operators in the execution of the different assembly and maintenance tasks, reducing the time to execute the tasks and errors made during their execution (Hořejší, 2015; Longo, Nicoletti and Padovano, 2017). By creating virtual environments that recreate the real AS, VR can, for example, be used to train the operators who will work in the AS and to design or make changes to the design of the AS (Brough et al., 2007; Grajewski et al., 2013; Gorecky et al., 2017). VR can also be used to remotely control mobile robots that are moving around the AS performing tasks (Chen and Chen, 2014). However, mobile robots are generally autonomous and can, unlike other transportation devices such as forklifts, offer opportunities to create new guide paths, improving the performance of the ALFS in terms of both time and safety (Elhoseny et al., 2018; Moysis et al., 2020; Panda et al., 2020). Moreover, mobile robots can be used not only to transport simple containers, but also to move entire shelves within the AS (Yoshitake et al., 2019; Gharehgozli and Zaerpour, 2020) or transport different types of objects by linking two or more mobile robots together, depending on the object being transported (Groß and Dorigo, 2009).

Finally, in the ALS, some components may be produced directly by AM in the workplaces, especially in the case of parts needed in small quantities, thus avoiding the need for their transportation and storage in workplaces or warehouses (Cohen et al., 2019b).

While in Table 1 it is possible to see the decisions that need to be made to design and manage an AS, these decisions are related to what we call traditional ASs, where the technologies responsible for the Fourth Industrial Revolution are not implemented. Assembly System 4.0 may need new decisions, in addition to those of traditional ASs; or the decisions of a traditional AS may change based on the technologies applied in AS4.0. However, at the moment, to the best of the author's knowledge, there are no studies that have tried to investigate this topic. Therefore, to fill this gap, one of the results of the present research study was recreating a table like Table 1 but for AS4.0. Thus, the present study has created two tables, one for the ALS and one for the ALFS of an AS4.0 (Subsection 4.1, Table 3 and Table 4).

3. Research design

This section presents and explains the research design of the current study (Figure 3). First, the research methods adopted to perform the study are described (Subsection 3.1). Second, the four requirements to ensure research quality (construct validity, internal validity, external validity and reliability) are presented (Subsection 3.2).

3.1. Research methods

Knowing the research methods used in the elaboration of a scientific document, such as a thesis dissertation, is of primary importance for understanding its validity and quality (Borrego et al., 2009; Queirós et al., 2017). Research methods can be divided into qualitative and quantitative methods. If qualitative methods seek to comprehend a complex reality and the meaning of actions in a specific context, quantitative methods instead seek precise and trustworthy measurements that allow for statistical analysis (Reswick, 1994; Queirós et al., 2017).

In the present study, both types of methods have been investigated. In fact, two qualitative methods (SLR and exploratory research) and two quantitative methods (experimental research and simulation modelling) provided the main outcomes of the study. After allowing us to understand the state of the art of AS4.0 and its future research opportunities, the SLR helped derive the three research questions. Exploratory research and experimental research helped answer the first research question (**RQ1**). Experimental research and simulation modelling helped answer the second (**RQ2**) and third (**RQ3**) research questions, respectively.

3.1.1. Systematic literature review

An SLR can make an important contribution to the advancement of research; it aims to give a historical perspective on the relevant study topic and an in-depth assessment of independent research activities (Mentzer and Kahn, 1995). In particular, an SLR is a review of precisely defined questions that employs a methodical and evidence-based approach for discovering, selecting and assessing secondary data. This method varies from others in its transparency, inclusiveness and explanatory and heuristic character (Tranfield et al., 2003). Moreover, as a scientific inquiry, an SLR should be valid, reliable and repeatable (Xiao and Watson, 2019). This is the reason why the steps applied to select the works analysed in an SLR need to be clearly defined (Seuring and Gold, 2012). Therefore, an SLR, compared with other types of

literature reviews, allows for a more objective overview of the search results and eliminates bias and error issues (Buchanan and Bryman, 2009). The SLR's primary purpose is to facilitate theory development, organise research being conducted and propose areas for further investigation (Webster and Watson, 2002).

The literature on the design and management of AS4.0 is not yet well structured and investigated. Hence, an SLR was a suitable approach to organise and unify knowledge within this field.

Therefore, in the present research study, an SLR not only helped to provide an overview of AS4.0, but also served to identify future research opportunities. Moreover, the future research opportunities inspired the research questions investigated here (Figure 3). Following the guidelines outlined by Tranfield et al. (2003), the SLR consisted of five steps. In the first steps, the relevant keywords for the research were identified and then combined to generate the query used during the research process performed on the Scopus database. During the second step, the refinement, limitations on the period of research, on the document type and on the language of the papers to study were delineated. Thus, the research was limited to only articles and reviews written in English in the period from 2005 to 2020. A combination of the keywords with these first limitations gave us an initial selection of 16,849 papers, which was reduced to 9,424 after we removed duplicates and considered only journals with a Scimago Journal Rank (SJR) greater than or equal to 0.5. The third and fourth steps—creating the inclusion and exclusion criteria and performing a second refinement, respectivelyhelped reduce the number of papers from 9,424 to 140. In these two steps, it was possible to remove all the papers that were related to topics outside the scope of the review and those that appeared relevant from the abstract but which turned out to be outside the scope once read in their entirety. Finally, in the fifth step, the snowball search, 17 papers were added. By the end of this step, 157 papers had been selected for inclusion in the SLR.

The entire process of the SLR, the selection of the papers to arrive at the final 157 papers and the analysis of these papers to arrive at the results was carried out using Excel.

26

3.1.2. Exploratory research

Exploratory research is often qualitative in nature, and it is a methodology approach that analyses previously unstudied research questions (Tegan, 2022). In fact, exploratory research is conducted when problems are in their early stages, when the topic or issue is novel or when data collection is challenging for certain reasons (Babbie, 2007). Therefore, exploratory research can be seen as the preliminary research to clarify the exact nature of the problem to be solved. It is used to guarantee that more research is considered throughout an experiment, as well as setting the research goals, gathering data and focusing on certain issues that would be difficult to notice without exploratory research (Wikipedia, 2021).

According to Tegan (2022), if you have a general notion or specific question that you want to investigate but no underlying knowledge or paradigm with which to do so, you can conduct exploratory research. This was the situation of the first research question (RQ1) of the present research study. In fact, when investigating the literature, there was no precise knowledge of what factors are relevant to understand when a company needs to decide on the configuration of its ALSs; because of this, there was also no knowledge related to the impact that new Industry 4.0 technologies have on these factors.

Therefore, in the current study, the exploratory research not only allowed us to define the seven factors relevant to ALS configuration decision making, but also gave us the opportunity to then understand whether and how new I4.0 technologies impact these factors. Moreover, because it is in the nature of exploratory research to help in proposing new ideas (Swedberg, 2020), this research opened the door to promising new opportunities. In fact, now that the factors and impact of the I4.0 technologies on factors are known, it would be interesting to quantify both the threshold values of the factors that determine the choice of one ALS configuration over another and the numerical values related to how the technologies have changed these thresholds.

The entire process of the exploratory research, which consists of the classification of the papers to define the factors and understand the impact of the I4.0 technologies on those factors, was carried out using Excel.

27

3.1.3. Experimental research

Experimental research is a type of study that rigidly follows a scientific research design. It involves testing or attempting to prove a hypothesis by way of experimentation (Pollfish, 2022). Therefore, during the experiment, researchers gather evidence that may be used to support or deny this hypothesis and that should help them make better decisions; hence, the present study is also known as hypothesis testing or deductive research approach. In general, experimental research has three categories that researchers might implement, which are preexperimental research design, quasi-experimental research design and true experimental research design (Pollfish, 2022). A pre-experimental study design involves observing a group or groups once the elements of cause and effect are established. As a result, pre-experimental research is a necessary first step in justifying the presence of the researcher's involvement. A quasi-experimental design, such as a true experiment, seeks to demonstrate a cause-andeffect link between an independent and dependent variable (Voxco, 2022). However, unlike a true experiment, a quasi-experiment does not rely on random assignment. The subjects are instead assigned to groups based on non-random criteria. A true experimental design is a statistical method for determining a cause-and-effect link between many factors (Voxco, 2022). This is one of the most precise types of research design, providing sufficient evidence to demonstrate the presence of correlations.

Therefore, knowing that experimental research is best suited for exploratory research, in which the goal y is to examine cause-and-effect relationships (Bhattacherjee, 2022), we decided to adopt it in two of our studies. In the first one, which was moved by the fact that the next step of exploratory research can be experimental research, we were interested in understanding the impact of AR on the factors relevant to determining the configuration of an ALS (RQ1). In the second one, we were interested in understanding if the methodological framework that we developed is valid (RQ2). Although, generally, experimental research is considered to be time-consuming and expensive to carry out (Berinsky et al., 2012), we were lucky to have some students who voluntarily participated in carrying out the studies. In particular, a total of 15 students participated in the AR studies and one PhD student participated in the experiment to validate the methodological framework.

Furthermore, because experimental research must be carried out in a controlled environment, we conducted the two experimental studies in the Logistics 4.0 Lab at the

Department of Mechanical and Industrial Engineering at the Norwegian University of Science and Technology (NTNU), where there were no disturbing elements. During the first experiment to develop the AR solution, we used a Kinect camera, the Webcam Zone Trigger Pro motion recognition software, work instruction programme (HTML file) and a projector. For the second experiment, to validate the methodological framework, we used the software Siemens Jack[™] to model the 3D environment, Synertial mocap as the motion capture system and the HTC VIVE[™] as the VR device. Moreover, in both the experiments, when it was necessary to do some calculations, we again used Microsoft's Excel software.

3.1.4. Simulation modelling

'Simulation modelling and analysis is the process of creating and experimenting with a computerised mathematical model of a physical system' (Chung, 2003). Simulation models may be either stochastic or deterministic (Harrison et al., 2007). Stochastic models feature probabilistic components, which means that the behaviour of a model in any given instance is influenced by chance. Monte Carlo techniques are commonly used in stochastic simulations. Deterministic models instead contain no probabilistic elements, providing the same outputs every time, so they only need to be run once for a specific model. According to Pedgen et al. (1995), the simulation of stochastic and deterministic models is performed for the following reasons: gaining insights into a system's operation, developing operating or resource policies to improve system performance, testing new concepts and/or systems before implementation and gaining information without disturbing the actual system. Furthermore, in general, simulation models consist of the following components: system entities, input variables, performance measures and functional relationships (Maria, 1997).

In the present study, simulation modelling was applied to answer the third research question (RQ3), where we were interested in understanding when a DT-based solution can be adopted to control the material management in an ALFS compared with a traditional Kanban or e-Kanban solution. Simulation modelling was chosen as the methodology because DT is still an emerging technology, especially in the case of materials management, and the physical development process of a DT-based solution requires an a sizeable investment (Gabor et al., 2016; Singh et al., 2021). Therefore, first, a general problem, together with the assumptions and all the parameters of the problem, was defined. Then, to solve the problem, three models were developed: the DT-based model, the traditional Kanban model and the e-Kanban model.

Each model attempts to faithfully replicate what is usually carried out to control the materials management in an ALFS when that specific solution has been adopted. Finally, the three models were simulated to arrive at the end at the creation of the DSS. The DSS gives the opportunity to support managers and practitioners in understanding which of the three alternatives is the best material management solution to adopt in their ALFS.

Regardless of the simulation model, all simulations were carried out using the statistical software RStudio and functions of the library 'simmer' (Ucar et al., 2019). Moreover, if there was the necessity to perform some calculations that were not particularly complicated, we used Excel software.



Figure 3. The research design of this research study

3.2. Research quality

In the present research study, qualitative and quantitative methods were applied. According to Karlsson (2016), Halldorsson and Aastrup (2003) and Voss et al. (2002), to determine whether qualitative and quantitative research can be judged to be of good quality, it must satisfy four requirements: construct validity, internal validity, external validity and reliability. Therefore, the following subsections explain how these four requirements were considered during the research process to produce good quality research that is reliable and ethical and can be used by other researchers and practitioners.

3.2.1. Construct validity

Construct validity assesses whether the right operational definitions for the ideas under consideration have been established (Voss et al., 2002). To appropriately account for construct validity, Yin (2017) suggested two essential aspects: (i) offer precise definitions of what is to be studied and (ii) demonstrate that the operational definitions do truly represent what is supposed to be examined.

In present study, the introduction section has provided a clear description of the scope. Definitions and explanations have been presented in both the research study and all appended papers. Furthermore, we attempted to maintain a clear chain of evidence, which, according to Yin (2013), indicates that readers should be able to trace the derivation of the results from the collected data. Therefore, to achieve this, as far as feasible, the reasons for each step, the information from each step and the decisions made at each step were documented and are reported in this study.

For the experimental research and simulation models, verification and validation techniques were used to ensure that the research reflected what was intended.

3.2.2. Internal validity

Internal validity entails revealing the right causal links while not ignoring the other factors that may explain these correlations (Karlsson, 2016). In other words, if it is determined that X occurred as a result of Y, ignoring the fact that X occurred also as a result of Z, the internal validity is poor. Internal validity is more appropriate as an evaluation criterion, particularly in exploratory and causal research, but not always in descriptive studies (Croom, 2009; Yin, 2017).

If the question of internal validity does not arise for the descriptive and exploratory research, one of the key techniques used to verify the internal validity of the experimental research is theoretical replication. In fact, in the experimental studies, which compared different application solutions in ASs, the expected results were formulated based on the literature before data collection. Following this, the empirical findings were then compared with this prediction.

For the simulation study, the system's behaviour and identification of causality were the main curiosities that drove the entire research. In this case, causality was established by adjusting each independent variable individually and then evaluating the causal effects on the dependent variables. Thus, simulation is suitable for investigating causal relationships (Croom, 2009; Bertrand and Fransoo, 2016).

3.2.3. External validity

External validity considers whether the findings from the study's data and environment are generalisable to larger groups and contexts (Cook et al., 1979).

In particular, to assess whether a study has external validity, one should ask whether the results apply to production environments and companies whose circumstances differ from those of the studied cases. This question allows others to make judgements regarding whether the research findings are transferable to other situations. As a result, to give the reader this opportunity, the goal of the present research study is to offer detailed information on all of the situations and cases studied.

Although the generalisability of a research study is not always easy to determine because each case has its own particularities (Yin, 2013), researchers should be adept at removing these peculiarities and creating a general case that can refer to as many situations as possible. After the general case, they can then focus on their specific case.

3.2.4. Reliability

The extent to which research may be duplicated and provide the same results is referred to as its reliability (Voss et al., 2002). The goal is to reduce bias so that another researcher may replicate the study and acquire the same results and conclusions.

Various strategies have been used to ensure reliability. First, presenting the research design and outlining the methodology utilised in the current research study, as well as each attached publication, makes it easier for researchers to repeat all the studies.

Furthermore, the fact of having several researchers participating in the study process and examining the data will eliminate any bias from a single researcher.

4. Design and management of Assembly Systems 4.0

This section presents and discusses the results and findings.

Following the research design (Figure 3), we began the study by performing an SLR. With the SLR, we obtained two main outcomes: the state of the art of AS4.0 (divided in two parts) and proposal of possible future research opportunities in AS4.0. Moreover, from three of the nine proposed future research opportunities derived the three research questions (**RQ1**, **RQ2**, and **RQ3**) we could obtain the rest of the main outcomes. The results are presented in Subsection 4.1 and are based on Papers 1 and 2.

To answer **RQ1** (How can AS4.0 configurations be affected by Industry 4.0 technologies?), we performed exploratory research and experimental research. Because of the exploratory research, we derived two main outcomes, including the factors relevant to determining the configuration of an ALS and impact of 14.0 technologies on these factors. Because of the experimental research instead investigating the impact of AR on the identified factors, we derived the last main outcome related to RQ1. The results and findings are presented in Subsection 4.2 and are based on Papers 3 and 4.

Furthermore, to answer **RQ2** (How can ergo-efficient AS4.0 workplaces be designed?), we performed experimental research. Because of the experimental research, we could verify the validity of the five-step methodological framework identified to design ergo-efficient workplaces of an ALS using VR and a motion capture system and to identify how the age of human workers can be included in the design of a workplace of an ALS representing the first and second main outcomes related to RQ2. The results and findings are presented in Subsection 4.3 and are based on Paper 5.

Finally, to answer **RQ3** (How can material management be controlled in AS4.0?), we performed simulation modelling. Because of the simulation modelling, we had first to develop a DT-based model to control the material management in ALFS that represents the first main outcome related to RQ3, and then, we developed a DSS to identify what is the best solution to control the material management in an ALFS between traditional Kanban, electronic Kanban and DT that represents the second main outcome related to RQ3. The results and findings are presented in Subsection 4.4 and are based on Paper 6.

4.1. State of the art of Assembly Systems 4.0 and future research opportunities

In the theoretical background section (Section 2), after presenting all the I4.0 technologies that can be adopted by companies (IoT technologies, CPSs, data analysis, cloud computing, collaborative robots (cobots), AR, VR, mobile robots, cybersecurity, blockchain and AM), we pointed out the ones that are more suitable for creating AS4.0 according to the authors who first introduced the topic of AS4.0 (Bortolini et al., 2017; Cohen et al., 2019a; Cohen et al., 2019b). However, these researchers have mainly focused on the introduction of such technologies or on part of them, their proof-of-concept development, analysis of their characteristics and their performance rather than studying how the I4.0 technologies are implemented in the different activities (assembly, control, transportation, preparation, material management) performed in an AS, studying the decisions at the strategic, tactical and operational levels that companies must make when implementing I4.0 technologies in their ASs. Therefore, an SLR was conducted to fill this gap, from which two main outcomes were derived. The first main outcome of the SLR refers to the state of the art of AS4.0 and is composed of two parts the I4.0 technologies in AS activities and the decisions that need to be made in AS4.0 at the strategic, tactical and operational levels. The second main outcome of the SLR instead refers to the future research opportunities that can open many research streams related to the design and management of AS4.0.

14.0 technologies in AS activities. The first part of the first main outcome derived from the SLR is presented in Table 2. The table illustrates the 14.0 technologies based on the different AS activities in which they have been implemented. From this result, one 14.0 technology, the AM, was found to be a possible disruptive technology in ASs and is currently not applied in any of the ASs activities. Indeed, based on our SLR, the technologies of the AS4.0 are collaborative robot, mobile robots, AR, VR, IoT technologies and cloud computing and data analysis. For the ALS, the technologies are divided into the two activities of assembly and control, while for the ALFS, they are divided into the three activities of transportation, preparation and material management. It should be noted that the family of IoT technologies include sensors, CPSs, DTs, motion capture system and wearables. Table 2. I4.0 technologies for an AS, here based on ALS and ALFS activities.

	A	LS	ALFS				
Technology	Assembly	Control	Transportation	Preparation	Material management		
Collaborative robots	х			х			
Mobile robots			х	х	x		
Augmented Reality	х	х		х			
Virtual Reality	х	х					
IoT technologies	х	х	x	х	x		
Cloud computing and Data Analysis	х	х	х	х	х		

Table 2. 14.0 technologies for an AS, here based on ALS and ALFS activities

Table 2 shows that there are two sets of technologies (IoT/cloud computing and data analysis) that are employed in all AS4.0 activities. Because of the nature of I4.0, this result is to be expected. In fact, IoT technologies let businesses collect all accessible data from AS4.0. These data must then be stored and shared, and cloud computing is quite useful in this aspect. Cloud computing can provide both the resources required to save and transform these data into information using data analysis techniques and then deliver them to the users who require them.

The technologies VR and mobile robots are only used in the activities of the ALS and ALFS, respectively. Although this finding was expected for the usage of mobile robots, it was not for VR. In fact, even though VR may be utilised in simulations, such as the study of human employees operating in the same environment as mobile robots, we could not locate any work addressing its usage in an ALFS.

The last information possible to obtain from the table is about AR and cobots. These technologies are used in both ALSs and ALFS. In particular, cobots are employed in ALF preparation activities, where they are used to do picking and kitting, while AR is used to instruct and coach human assembly workers and monitor assembly quality.

The decisions of an AS4.0 at the strategic, tactical and operational levels. Once we understood in which activities the technologies are applied in AS4.0, the study moved to the second part of the first main outcome derived from the SLR, which consists of understanding what decisions companies need to make when they decide to implement one or more of the I4.0 technologies to their AS. These decisions are divided based on the different levels (strategic, tactical and operational) at which they need to be made and on the subsystem of an AS in which they are applied (ALS, Table 3, and ALFS, Table 4). Moreover, the decisions for an AS4.0 are also related to the decisions areas of a traditional AS (see Table 1 above).

			Industry 4.0 Technologies in Assembly Line Subsystem (ALS)							
	Decision areas		Collaborative robots	Augmented Reality	Augmented Reality Virtual Reality		Cloud computing and Data Analysis			
term	Strategic	Configuration of the system	Type of cobot	Decision of which AR instructions create	Decision of which VR instructions create	 Type of IoT technologies 	 Data to storage and analysis 			
Long		Level of automation	 Type of cobot Skills to use the cobot 	 Type of AR devices Skills to use the AR devices 	 Type of VR devices Skills to use the VR devices 	 Type of IoT technologies Skills to use the IoT technologies 	 Type of Cloud Computing system and/or Data Analytics techniques Skills to use these technologies 			
Medium term	Tactical	Workplace design	Workplace design	 How to create and share the instructions 	Workplace design	Workplace design	 Information to create and share How to create the information 			
		Assembly line balancing	 Balancing of the assembly line with the cobots 	 Number of AR devices 	 Number of VR devices 	 Number of IoT tecnologies 	Number of information to create			
Short term	Operational	Sequencing	Sequencing of the operations	 Real-time information sharing → assembly instructions 	 Real-time information sharing → assembly environment and instructions 	 Real-time control of the quality of the assembled products, machines, workers, and environment 	 Real-time sharing and displaying of the information 			
		Control of the ALS	Control of the system	Control of the system	Control of the system	Control of the system	Control of the system			

Table 3. Decision areas divided into different levels for each I4.0 technology in an ALS

		Industry 4.0 Technologies in Assembly Line Feeding Subsyst							
Medium term Long term	Decision areas		Mobile robots	Augmented Reality	IoT technologies	Cloud computing and Data Analysis			
		Configuration of the system	 Warehouse design Type of mobile robots Design the guide path 	 Decision of which AR instructions create Type of AR devices 	Type of containers	 Data to storage and analysis 			
	Strategic	Level of automation	 Type of mobile robots Skills to use the mobile robots 	 Type of AR devices Skills to use the AR devices 	 Type of IoT technologies Skills to use the IoT technologies 	 Type of Cloud Computing system and/or Data Analytics techniques Skills to use these technologies 			
	Tactical	Feeding policy selection	Feeding policies	 How to create and share the instructions 	 Positioning of the IoT technologies 	 Information to create and share How to create the information 			
		Scheduling	 Scheduling of the mobile robots Number of mobile robots 	Number of AR devices	 Number of containers in the warehouses and in the ALS 	How to share the information			
Short term	Operational	Routing and material/information management	Routes of the mobile robots	 Real-time information sharing → sequencing of kitting and picking information 	Real-time control of the capacity of the container	 Real-time sharing and displaying of the information 			
		Control of the ALFS	Control of the system	Control of the system	Control of the system	Control of the system			

Table 4. Decision areas divided into different levels for each I4.0 technology in the ALFS

The decisions can be summarised as:

Strategic. When we talk about the configuration of the system in an AS4.0, we are referring to what is necessary to know before deciding which technology would be best to implement in the AS. For example, the integration of a cobot into an AS can change the spaces that are necessary to allow the robot to work alongside humans; if a company wants to create instructions to train their human workers in a virtual environment without the need for a real one, the firm may prefer to use VR. In contrast, if the company wants to guide their human workers during the execution of their tasks within the environment where they work, AR may be the best solution (Song et al., 2016; Elhoseny et al., 2018; Yao et al., 2018; Mateus et al., 2019; Stadnicka and Antonelli, 2019; Danielsson et al., 2020; Miller et al., 2020). Moreover, the decisions made regarding the types of containers and the types of warehouses in which these will be used may influence the kind of sensors that can be applied to them and the transportation devices that can transport them. In view of all the data that can be collected, companies also now need to decide on the level of automation of their AS (Kolbeinsson et al. 2019; Fragapane et al., 2020; Peron et al., 2020). This is because the different forms of data that can be collected require different sensors and techniques to analyse them. Moreover, regarding the type of information generated by the data analysis, the selection of the device that will work the best is critical. This information can be as simple as just numbers or text, but it can also be more complex, such as pictures or videos. This choice depends not only on the type of information to deliver, but also on who will receive it and which tasks must be performed. For example, in the case of large products, wearable devices will be more preferable than fixed ones more useful in the case of small products assembled in standard workstations. Furthermore, the main purpose of using such technology will also affect its implementation and, hence, the level of automation of the ASs. The selected technology will influence the skills that will be required to use it. The company then needs to know whether it already has someone with these skills or whether it will have to hire or train someone. In the case where the company can rely on internal resources with these skills, technology selection will be facilitated, as well as its implementation. This will also increase the general technological knowledge of the whole company in the case where these resources share their expertise. Furthermore, even if a company is not planning to implement any new technology in its AS4.0, a strategic decision may be to hire skilled human workers, with previous experience in I4.0 projects (Dolgui et al., 2022). These workers are valuable resources that could assist and guide the company toward the possible implementation of I4.0 technologies.

Tactical. When the technologies have been chosen, the company needs to design the workplace in which they will be used and determine how they will be used (Havard et al., 2019; Faccio et al., 2020). Some forms of technology, for example, VR and motion capture systems, can help in designing the workplaces where other technologies will be used. Indeed, VR provides the possibility of creating virtual environments, where human workers toward motion capture systems can interact with the virtual elements and where the technologies, such as cobots and mobile robots, can be modelled. This facilitates the study of different workplace designs. As a result of the virtual environment, different alternatives between the human workers and technologies can be studied before their implementation in real workplaces. This helps create optimal workplaces where technologies and human workers can interact with optimal results (Choobineh et al., 2012; Yoshitake et al., 2019; Gharehgozli and Zaerpour, 2020). The feeding policies determined by the company can also influence the design of the workplace because the transportation of a container, like a pallet, may require a different space when it is delivered to the workplace when compared with the transportation of a shelf. A time analysis and decision aid may be necessary to identify the numbers of cobots, workers, AR devices or mobile robots required to execute the different tasks in the AS. In conjunction with this time analysis, it may be necessary to evaluate the ergonomic impact of the technologies while they are being used as a way to understand where and when it is best to use them. The time analysis and ergonomic impact can be evaluated because of the sensors applied in the technologies and by the human workers, such as IoT sensors or wearable devices. Another possibility is the combination of cameras that monitor the movements of the technologies and human workers as a result of computer vision algorithms. At this level, a company is aware of which data it must collect and now needs to know where to install the sensors to collect these data. The sensors need to be installed in protected and proper places to reliably and safely collect the required data. Therefore, the significant amount of data that companies can collect from the sensors need to be analysed and validated (Papakostas et al., 2016; Plantar et al., 2017; Wu et al., 2018). This should be done because next step is to decide what information the company is interested in creating from the collected data and how this information will be created. The final decision also relates to the issue of the final recipient of such information (Papakostas et al., 2016; Plantar et al., 2017; Wu et al., 2018). It is important to create specific information for each user to avoid possible misunderstandings, which can result in the execution of incorrect activities or decisions.

Operational. At this level, the technologies have been implemented and are working. The employees now need to know how to do their jobs, and the company needs to control them to ensure that they are doing them to the best of their ability (Faccio et al., 2019; Alavian et al., 2020; Baumann et al., 2020). The daily activities that are assigned to cobots need to be sequenced based on the products that they are required to produce, and the routes of the mobile robots must be generated according to the paths that they need to take. The AR and VR equipment must also receive the information necessary for the human workers to carry out their activities. This information needs to be sequenced based on what activities the human workers will carry out when they are using the technology, and it needs to be easily comprehensible and personalised. Indeed, there can be a negative impact in the case of incorrect information, but also if the information is not understandable or clearly visible to the user (Tarallo et al., 2018; Oyekan et al., 2019; Chen et al., 2020a; Chen et al., 2020b; Negri et al., 2020). Therefore, a good control loop of accuracy and usability is important for continuously improving their utilisation. Moreover, the correct information at the

appropriate time and place not only needs to be sent to the AR and VR devices, but also to all the actors in the AS4.0 that require it. This needs to be carried out efficiently while avoiding a possible loss of information during the communication process. Cloud computing can provide a way of sharing all this information, without the need for physical connections, across all of the devices in the AS that have the capacity to give them in output (Liu et al., 2017; Nuzzi et al., 2020). All of this information can be generated using data analysis techniques and algorithms, and these need to be checked periodically to ensure that the parameters that they are using are still valid; alternatively, they may need to be updated, for example, because of variations in customer demand. These techniques and algorithms also need to be updated if new techniques or algorithms that are more efficient are developed or if they fail to work properly. It is not only the software aspects of the technologies that need to be checked: the hardware parts of technologies, such as mobile robots, cobots, AR and VR devices, also need to be periodically maintained and updated to avoid problems during the execution of their tasks (Zhuang et al., 2018; Nikolakis et al., 2019a; Nikolakis et al., 2019b; Malik and Brem, 2020). For example, the malfunctioning of the batteries that power the mobile robots affects their performance in terms of speed and autonomy, thereby having a negative impact on the performance of the AS4.0. Checking the hardware can help to avoid not only losses in the performance of the AS4.0, but also in its security. Therefore, it is clear that appropriate maintenance measures are important for strictly controlling the reliability and availability of the hardware of the technologies.

Future research opportunities. Table 3 and Table 4 provide the results leading to the second main outcome of the SLR study and can hopefully inspire future researchers in the field of AS4.0 to address new research challenges that go beyond the study of the mere use of these technologies. As seen from Figure 4, for each of the levels of strategic, tactical and operational, three future research opportunities have been identified. These nine future research opportunities are now described.

42



Figure 4. Future research opportunities for Assembly System 4.0

4.1.1. Future research opportunities at the strategic level Research opportunities SL1: Models of dynamic reconfigurable AS4.0

From the SLR, we observed that mobile robots and cobots were the two most researched technologies. One aspect in our analysis that was not deeply investigated was related to the high level of reconfigurability that can arise because of these two technologies, especially from the use of mobile robots. In a reconfigurable system all the equipment, human workers or material handling systems should be rapidly added, removed, modified or interchange in response to changing needs and opportunities. Because of their flexibility, mobile robots can facilitate companies in creating AS4.0 configurations (primarily networks rather than line configurations) that change dynamically based on the current demand. In fact, using mobile robots as a workplace in which human workers perform the assembly tasks can give the opportunity to increase or decrease the number of workplaces that are needed based on the demand, and the same mobile robots can also be used to transport all the components that the human workers need for each task (Battaïa et al., 2018). Moreover, a mobile robot can be equipped with a cobot to create a collaborative mobile robot that can pick and transport components and then execute assembly tasks based on the same components. Although these approaches can be easily adopted to create different AS4.0 configurations, it is essential to understand what is the best configuration in terms of performance and cost, here based on the situation under study. However, to do so, first, it is necessary to know what are the factors impacting the decision of the configuration of an AS to choose and how I4.0 technologies impact these factors. Once this is understood, future studies should focus on mathematical and simulation modelling of the different configurations created by the different I4.0 technologies. These models will allow researchers to improve the knowledge related to the factors that can affect the reconfigurability of the system. Sensitivity analyses will also be useful in studying how these parameters can change the reconfigurability and performance of the different configurations. In the end, decision support systems, like decision trees and evolutionary algorithms, can be created to support decision makers in determining what is the best configuration to adopt based on their needs.

Research opportunities SL2: Models to support the selection of a suitable level of automation

14.0 technologies can be used to execute a task or collect and share data, for example, to assist operators or support managers. Hence, the choice of a piece of technology is not only related to the help that it can give in terms of the physical and cognitive aspects of the execution of a task, but also to the data that can be collected and shared with it. A huge amount of data can be collected and sent to the AS4.0, and new models are needed to support decision makers regarding how to automatise the AS4.0. These models can provide a methodological toolbox that can guarantee a structured implementation of the technologies to ensure a suitable level of automation for AS4.0. For example, the decisions that need to be made on the most suitable amount of information to give to the operator through AR or assistive technologies (which relates to the level of collaboration between operators and collaborative robots) or on the configuration of the systems can be made by managers based on the available data. To create these models, further research is required to deepen our knowledge of the case where multiple technologies work together to complete different tasks. Simulations and experiments are needed to validate these models. The results from these methods, together with the possibility to do sensitive analysis, give the opportunity to evaluate different cases with advanced data analytics tools that can create decisional support systems. Companies can benefit from these results because they will help them simplify the decision process of the level of automation to adopt in their AS4.0.

Research opportunities SL3: Models for dynamic assignment of technologies on AS4.0 configurations

We have seen that, at this level, companies have to choose the configuration and level of automation of their AS4.0. Here, this can mean that the choice of a configuration compromises the possibility of using a specific technology and instead that the technology is most appropriate for another kind of configuration. Moreover, the performance of the AS4.0 is correlated to the combination of configurations and technologies, with synergic effects arising in some cases. Therefore, at this level, to determine the goals of the AS4.0 that the companies want, it is necessary to have a guide that supports decision makers in the correct assignment of the different technologies to the configurations under analysis. The decision of the assignment of the technologies on the different configurations will facilitate other decisions, such as, for example, the number of devices to buy and skills that are necessary in the different configurations. To understand if a technology is good or not for a specific configuration, future research should consider its behaviour and performance in different configurations. Case studies, experiments and simulations represent opportunities to collect the necessary data to create new decisional support systems that can guide companies in better deciding which configurations to implement for the different technologies. It will also be important to study the behaviour and performance of more technologies together and the interactions between the technologies and human workers to see if their implementation together generates issues in the configuration or not.

4.1.2. Future research opportunity at the tactical level

Research opportunities TL1: Methods for ergo-efficient workplace design

When a human worker is involved in the execution of activities in the AS4.0, they should be able to do these without worrying about the workplace in which these activities take place. To enable this, the workplace needs to be optimised, and VR can be used to help create virtual environments that replicate the real AS, hence allowing a user to design and study a range of configurations of workplaces without needing a physical model. These virtual solutions need to be further investigated, and new parameters, such as the ages of the human workers or integration of additional technologies into virtual environments, should be evaluated. Different human characteristics may require different workplace design solutions

to facilitate high levels of ergonomic comfort for manual processes and to increase efficiency. The data collected from human workers while they are executing their activities can be used to adapt the workplace based on the tasks that need to be executed within the workplace and characteristics of the human worker executing them. Motion capture (mocap) systems can be used to collect these data. A mocap system gives the opportunity to create a virtual copy of the object under study by utilising sensors or cameras to recreate its movements. Experiments are first needed to determine the relevant parameters for optimising a workstation, and case studies will then be required to verify whether these parameters are correct, hence designing a workplace that is not just optimal in terms of performance but is also efficient in terms of human well-being. In relation to the acceptance of technology, further studies could provide insights into the factors affecting success. Data analysis and artificial intelligence techniques can be useful during this step.

Research opportunities TL2: Models of new feeding policies

The use of mobile robots, with or without a cobot, can open up the possibility of creating new feeding policies because if these are used to move shelves, each level of the shelves can be designed to store different components in different ways. For example, a kit can be stored at one level and only one type of component can be stored at another level, so the shelf may be delivered to one or more workplaces. Instead, if mobile robots are used in conjunction with a cobot, these collaborative mobile robots can be used to pick the individual components from warehouses or supermarkets and transport them to the ALSs that need them. To study these new opportunities, advanced modelling and simulations and, then, real case studies will be needed to validate the results of the models. Moreover, AR can be used to support the human workers involved in feeding tasks by giving them instructions, such as which component to pick, where to put the components and how many components to pick. When AR is used together with mobile robots, the two technologies need to be integrated. This means that the AR instructions should not only able to change based on the demand for products. but also based on the characteristics of the mobile robots used to execute the feeding policies. In addition, these AR instructions need to change based on the components that the human workers have and must be visible and in the correct position for the particular type of mobile robot used. Further investigation and testing through experiments that replicate real scenarios can better highlight the strengths, weaknesses and potential advantages that the use of AR with mobile robots can generate in terms of developing new feeding policies.

Research opportunities TL3: Models for real-time multi-objective balancing of ALS and scheduling of ALFS

The balancing of the new ALS needs to take into account not only the assignment of the different tasks within the different workplaces, but also the technologies used in the ALS. However, there is yet to be research on how these technologies can affect line balancing. For example, the choice to adopt AR to support operators in executing the tasks is not yet well investigated, particularly regarding the impact on the balancing of the ALS. In any case, the use of AR is very interesting because it can help to reduce line balancing effort for the ALS by creating flexible, dynamic and self-balancing ALSs in which the number of workplaces may change based on the current demand from customers. Hence, if demand is low, a single operator may be able to do all the tasks in one workplace, here thanks to the use of AR, whereas if demand grows, the number of workplaces can increase and the AR instructions will need to be divided between the workstations. One avenue for future research would be to investigate the number of AR devices needed and the ways in which the instructions that they display to human workers needs to change. The various types of mobile robots, which may or may not have a cobot mounted on them, allow for new tasks to be carried out in the ALS and ALFS. For example, we saw that a mobile robot with a cobot can pick up a component from the warehouse, transport it to the ALFS and then execute a task based on this component in the ALS. The new dynamic network configurations that have become possible require new assembly line balancing methods that consider the evolutionary aspects of such configurations, with multiperiod and multiobjective models that can be applied in real time to adapt the workload of the workstations without affecting the work of the operators. In this case, economic, performance and ergonomic objectives should be considered together.

A mobile robot can be used solely as a workplace or transportation device. Different mobile robots can carry out different tasks, giving rise not only to new opportunities, but also to a more complex problem in relation to scheduling these mobile robots. A company needs to optimise the use of its mobile robots based on the tasks that they can do, so new scheduling models need to be created that can consider all the possible tasks that these mobile robots can carry out. These models first need to be studied through simulations and, then, with experiments or case studies to verify the results. A connection between the different actors in an AS4.0, here thanks to all the sensors and, hence, the collected data, can allow for synchronised assembly line balancing and scheduling to simultaneously optimise both the assembly and activities needed to deliver the components. Future investigations into methodologies for dynamic and synchronised assembly line balancing and scheduling of the AS4.0 are needed to allow companies to use their technologies in an optimal way.

4.1.3. Future research opportunity at the operational level

Research opportunities OL1: Smart and real-time methods for sequencing of ALS and routing of ALFS

ALSs need to know the sequence of products that they have to assemble on a daily basis, and mobile robots need to know the routes to follow so that they can arrive at their destinations. Algorithms developed in the future to solve these problems will need to take into consideration the new information generated by the AS4.0 in a predictive and prescriptive way, which will involve not only reacting to the changes required, but also anticipating them and adapting in a proactive way. For example, a new sequencing algorithm may consider the fatigue of human workers when deciding which products to assemble or which workers should execute the tasks to make them. If the human workers are tired, it may be possible to sequence products that are easy to assemble or to let the workers take a break to recover. Sequencing can be done at the same time as routing when the products to be assembled are determined with the goal of minimising the routes travelled by the mobile robots. Sequencing should also take into account the quantity of components stored in the warehouses and supermarkets, that is, to optimise inventory management by reducing stocked quantities by as much as possible. Sequencing and routing can be synchronised to allow the products in an ALS to be assembled using only the components closest to that ALS. To achieve this, the routing of the mobile robots must consider their positions in real time; only the mobile robots closest to the components will be used to transport them to the ALS. This synchronised sequencing and routing can reduce the travelling time of the mobile robots and can also increase the productivity of the ALS because the components are delivered more quickly.

Research opportunities OL2: Methods for efficient control of AS4.0

Today, for the first time, it has become possible to control the quality of products, assembly activity and all the activities of an AS4.0. This has been facilitated by the huge amounts of data that can be collected from an AS4.0 if its activities are monitored, including data from each step of the assembly tasks, from the products being assembled, from the mobile robots while they are working and so on. After collection, these data need to be understood and analysed so that companies can use them. Data analysis techniques can be applied to these data to give useful information as the output. New machine learning and artificial intelligence algorithms that transform data into information can be developed by studying the different activities of the AS4.0. Simulations will be useful in understanding the performance of the methods, and experiments with real scenarios should then be carried out. It is important that the right data are collected to ensure the success of these methods and to give valid results, and more research is needed to identify the factors that make a data analysis technique reliable. Surveys of practitioners can help in developing an understanding of the factors that can increase reliability. More reliable are the results of a data analysis technique and more precise can be the control of the AS4.0. At the same time higher can be the benefits that can be generate from these analyses. In the material management activity, for example, the adoption of data analysis techniques can open the opportunity to optimise the stock of materials in the warehouses and workplaces and, at the same time, help optimise the flows of these materials in the AS4.0 in a dynamic and real-time way, such as dynamic Kanban systems, dynamic replenishment and integrated replenishment policies. For a dynamic and synchronised stock and flow of the material, new objective functions can be defined to improve the performance of material management in an AS4.0. Benefitting from all these data and information, it is possible to model all, or parts of, the AS4.0 to create its DT. The DT of the AS4.0 gives the opportunity to control in real time what is happening in the AS4.0 and simulate what can happen in the future using predictive analysis models. Indeed, it allows us to see, for example, how mobile robots work in a single day and then simulate their performance for an entire week. This simulation can be useful for understanding whether the mobile robots will decrease their performance or not after a period of continuous work. Although DTs have significant potential, it is important to understand what it is relevant to monitor. Therefore, future studies can be orientated towards understanding which digital copies are the most important to create based on the characteristics of the AS4.0. Case studies, experiments and interviews with the practitioners can help create frameworks to follow when creating the digital copies while simultaneously verifying their reliability.

Research opportunities OL3: Models and methods for maintenance of technologies

As discussed in the previous section, both the software and hardware aspects of these technologies need to be controlled and updated. All of the sensors in the AS4.0, the mobile robots and the AR devices must operate without problems over their lifetimes. For example, the sensors must collect all of the appropriate data without missing any, and the mobile robots must run at a certain speed to execute their tasks. The development of new models and methods for the maintenance of these technologies, both predictive and proactive, should be a subject for future study. These models and methods can help avoid a loss of performance of the AS4.0 because of hardware problems with the technologies. It is important that these methods recognise when the performance of a particular technology is reduced, even if the problems are not evident, and when it must be replaced by a new model. Technology must also be replaced when it becomes obsolete. New data analysis techniques should be developed by creating predictive analytics models to predict when both of these situations are likely to occur. The information created by these techniques can then be sent to the technologies that are recognised as obsolete or that are not working properly to make them stop working. Once they stop, they can be evaluated to understand whether to fix or dispose of them.

4.1.4. The investigated future research opportunities

However, investigating all the identified future research opportunities would exceed the research time frame. Therefore, the results presented in the current research, as already depicted in Figure 3, and highlighted here in Figure 5, have been derived from only one future research opportunity per level: strategic (SL1), tactical (TL1) and operational (OL2). The selection of the three future research opportunities is not casual. In fact, as already written in the research motivation subsection (Subsection 1.3), it stemmed from the idea of investigating three of the decision areas related to the design and management of an AS and

enhancing the Logistic 4.0 Lab in the Department of Mechanical and Industrial Engineering at NTNU using some of its I4.0 technologies.



Figure 5. The future research opportunities investigated in this study

One research question for each of the three future research opportunities of Figure 5 has been identified. These are not to be considered as the only possible research questions that can fulfil the three future research opportunities selected in the current study but as three among all the possible research questions that can be generated to satisfy them. This is not to diminish the importance of the present work but rather to emphasise how many more research possibilities can be created through the current study.

The three research questions investigated in the rest of the section, which are linked with the future research opportunities to which they refer (see Figure 3 and Figure 5) and with the reference subsection where they are answered, are as follows:

- SL1 RQ1: How can AS4.0 configurations be affected by \rightarrow Subsection 4.2 Industry 4.0 technologies?
- TL1 RQ2: How can ergo-efficient AS4.0 workplaces be → Subsection 4.3 designed?
- OL2 RQ3: How can material management be controlled in \rightarrow Subsection 4.4 AS4.0?

Each subsection (4.2, 4.3 and 4.4) contains the following:

- The reasons why the research questions were created, which are the research gaps that were intended to be filled based on the future research opportunities investigated (Figure 5)
- The main outcomes that help answer these questions

Although there is a connection between the subsections (4.2, 4.3 and 4.4), each subsection is developed to be understood without the need to read the previous one. In fact, each

subsection is developed with the intention of allowing any reader to fully understand the research conducted, whether they are interested in knowing the answers at all the research questions (thus following all the 'branches' of the research design of Figure 3 after understanding the status of AS4.0) or if they want to know only the answer at one specific research question (thus following only one 'branch' of the research design of Figure 3 after understanding the status of AS4.0).

4.2. Factors affecting the configurations of Assembly Line Subsystems with Industry 4.0 technologies

This subsection aims to answer the first research question, **RQ1**, which has been derived from SL1, Models of dynamic reconfigurable AS4.0 configuration:

How can AS4.0 configurations be affected by Industry 4.0 technologies?

In recent years, companies have faced great changes in customers demands (Battaïa et al., 2018). In fact, customers are no longer passive clients who buy only the products that companies offer them; they are now more often active clients who ask for personalised, customised products that are closer to their needs and desires (Pollard et al., 2008; Kucukkoc and Zhang, 2017). If companies decide to accept these requests from their customers, they must be ready to manage the production of more complex product models (Otto and Li, 2020), and configuration of their ALSs should be dynamic reconfigurable (Battaïa et al., 2020). In fact, dynamic reconfigurable configuration of ALSs can be model with a certain level of flexibility, which can involve the reconfiguration of human workers, technologies and equipment, to be able to face the rapid change in market conditions (Hashemi-Petroodi et al., 2020a; Hashemi-Petroodi et al., 2020b; Yelles-Chaouche et al., 2020). Companies such as Sew Eurodrive, Porshe an Audi are already trying to model dynamic reconfigurable configurations of their ALSs, and to do this, one of the first decisions to be made concerns the configuration of the ALS to be adopted (Ilika, 2017; Porsche, 2019; Sew Eurodrive, 2022). However, this strategic decision is not easy to make, so it is important for companies to know which factors characterise the different configurations as these are also responsible for determining how dynamic reconfigurable a configuration can be. Although in the literature it is possible to find some qualitative guidelines about when to model one ALS configuration with respect to another (Battini et al., 2011) and a work that gives an idea of factors that can be considered is the one of Aase et al. 2004, the factors that need to be taken into consideration to model dynamic reconfigurable configuration of ALSs have still not been well identified and defined. Therefore, through exploratory research, we first identified and defined the factors that need to be taken into consideration when one wants to model dynamic reconfigurable configurations of ALSs and identify how Industry 4.0 technologies impact these factors.

The factors that are relevant for modelling dynamic reconfigurable configurations of an ALS. Thanks to the exploratory research of the literature, in which we were interested in detecting how companies can model dynamic reconfigurable configurations of their ALSs, we derived the first main outcome of this subsection, which is related to the identification and definition of the relevant factors that companies need to know when they have to model the configuration of their ALSs. The knowledge of these factors is of considerable importance when companies want to model dynamic reconfigurable configurations since based on their values it is possible to know if is possible or not to model dynamic reconfigurable configurations of ALSs. The factors are the following:

1. *Cycle time:* indicates the time elapsed between completions of two consequent units (Battini et al., 2011).

2. *Number of tasks:* the number of different activities that each operator must do in each workstation; it is a measure used to indicate the problem size for combinatorial problems such as the assembly line balancing (ALB) problems (Mastor, 1970).

3. *Network density:* calculating network density entails dividing the number of actual precedence relationships by the theoretical maximum number of relationships that could exist for a problem of that size. Extreme network density values of 0.0 and 1.0 correspond to flexible and rigid assembly sequences, respectively (Aase et al., 2004). For a more complete description of network density factors, see Johnson (1981).

4. Products variety: is the number of different products assembled in an assembly line. The performance of the assembly lines may be different depending on the number of different products variety to be assembled (Hu et al., 2011; MacDuffie et al., 1996). The higher this number is, the more likely it is that more experienced operators will be requested to assemble them (Johansson et al., 2016).

5. Volumes variety: is the difference between the quantity to produce a product one day and quantity to produce the same product the following day (Li and Gao, 2014).

For example, one day of product A can produce 100 pieces and the next day only 50. This can mean that a different number of operators are requested to assemble the same product in the two different days based on the different quantities that need to be assembled each day (Şahin and Kellegöz, 2017).

6. Quality: is the quality of the final assemble products that come out from the assembly line. To represent the quality in an assembly line, it is possible to count the number of products that come out from the line with or without defects (Monden, 1993). The more products that come out with defects, the worse the performance of the line will be.

7. *Complexity of tasks:* is the complexity of each task of the assembly process. This is the experience that each operator has in the assembly process. It is possible that an operator with more experience can complete a task in less time than an operator with less experience or that they can complete more tasks than another operator with less experience (Mossa et al., 2016). The operator's experience can be represented in a quantitative way, for example, by the number of years that an operator is working, time working in the same company or the number of hours that they have worked doing the same activities (Lin et al., 2007).

The impact of I4.0 technologies on the factors. Because the impact of I4.0 technologies on the aforementioned factors is unknown and this impact on the factors could facilitate modelling of dynamic reconfigurable configurations, modelling an ALS configuration may become even more complex when I4.0 technologies are implemented into ALSs (Shtub and Dar-El, 1989; Boysen et al., 2008; Cohen et al., 2017). However, now that the factors that companies need to be aware of to model dynamic reconfigurable configurations of their ALSs are known, it is possible to investigate how the I4.0 technologies impact them. From the SLR, it was found that the I4.0 technologies used in ALS are collaborative robots, AR, VR, IoT technologies and cloud computing and data analysis. However, when it is time to model the configuration of the ALS, VR does not have an impact. In fact, the exploratory research we conducted has led to a result for each I4.0 technology as to whether they decrease or increase the value of the different factors, but no result was found for VR (Table 5). Table 5 represents the second main outcome of this subsection, which is also the response to **RQ1**.

Table 5. 1	The	impact	of	Industry	4.0	on	the	factors
------------	-----	--------	----	----------	-----	----	-----	---------

	Industry 4.0 technology							
Factor	Collaborative robot	Augmented Reality	IoT technologies	Cloud computing and Data analysis				
Cycle time	Decreases	Decreases/ Increases	Decreases	Decreases				
Number of tasks	/	/	/	/				
Network density	/	/	/	/				
Products variety	Increases	Increases	Increases	Increases				
Volumes variety	Increases	/	Increases	Increases				
Quality	Increases	Increases	Increases	Increases				
Complexity of tasks	Decreases	Decreases	Decreases	Decreases				

From the table, for example, it is possible to see that collaborative robots, improving the performance of the ALS (Bloss, 2016; Fast-Berglund et al., 2016), giving the opportunity to decrease the 'Cycle time' or the IoT technologies and cloud computing and data analysis can increase the 'Quality' of the final products analysing, here because of data analysis techniques, such as machine learning algorithms, the data collected from sensors that control the assembly process (Georgakopoulos et al., 2016; Schmitt et al., 2020). Moreover, it is possible to see that it has been identified that no technology has an impact '/' on the factors 'Number of tasks' and 'Network density'. This is because these two factors are more related with the design characteristics of the product that need to be assembled than to the assembly process to assemble it. Although these abovementioned results are interesting, one finding that stands out from Table 5 is the 'Decreases/Increases' related to the factor 'Cycle time' for the technology 'AR'. This finding indicates that the true impact of this technology on 'Cycle time' is not yet well understood. In fact, current studies using AR to assist human workers in performing assembly activities have yielded conflicting results. This might be connected to the human workers' experience with the technology or on how the AR instructions are developed and delivered to the human workers.

Based on the results presented in Table 5, it is possible to derive how the modelling of the configuration of an AS can change if a company decides to implement one or more of the I4.0 technologies. In fact, it is possible that a decrease or increase in one or more of the considered factors may give companies the opportunity to model a configuration that they would not

have been able to model if they had not implemented I4.0 technologies. In particular, we focused on two of the most adopted configurations: the SL and US line (Bagher et al., 2011; Rabbani et al., 2012; Mukund and Ponnambalam, 2016). A SL configuration is represented by a sequence of workplaces in a line, as shown in Figure 6a. Simply rearranging the SL into a U-shaped line, it is possible to obtain the US line, as shown in Figure 6b. Usually, in SL, operators perform their tasks staying in one designated workplace, thus being FW configurations. In US, instead, operators can move across the workplaces to assemble the finished products and, thus, are WW configurations (Calzavara et al., 2021). Therefore, the results shown in Figure 7 that are related to SL and US can be generalised for FW configurations and WW configurations, respectively.



Figure 6. Straight-line configuration (a) and U-shaped line configuration (b)

Because technologies create the same impact for the various factors, in Figure 7, instead of making a table for each technology, we have considered only the two cases where companies have to model the AS configuration, between SL and US, without technologies and with technologies. In this last case, if more than one piece of technology is adopted, the positive or negative impact of each technology is considered to arrive at the final decision of the configuration chosen. In addition, we can see that the choice of configuration to be adopted is also based on the number of workplaces that will form the configurations and number of operators who will work in them.


Without technologies



Figure 7. How the choice of the AS configuration changes based on the impact of I4.0 technologies

However, it is worth mentioning that, at the moment, these results are based only on theory and need to be validated through quantitative analyses. In fact, all the threshold values that determine the transition from one configuration to another—in both cases without and with technologies—are not yet known. Therefore, more quantitative studies are needed that can also investigate this; once the threshold values of the factors are known, the ability of I4.0 technologies in creating dynamic reconfigurable ALS configurations will be improved.

The impact of AR on the identified factors. One of the I4.0 technologies that, in our opinion, seems to be very promising for creating dynamic ALS configurations is AR. In fact, this technology gives the opportunity for operators to assemble different products variety during the same day, even if they have never assembled them without making mistakes, or reducing mistakes, thus increasing the quality of the ALSs. However, as shown in Table 5, there is a result that seems to contradict our opinion: the one on factor cycle time. In fact, it is not clear if the adoption of AR can make companies choose an ALS configuration that would be chosen only if the cycle time is higher compared with another. Therefore, to conclude this subsection, following this result, experimental research with an example of a prototype of projected augmented reality (PAR) has been conducted, and the third main outcome of this subsection has been obtained. The PAR was used as assistance system to provide assembly instructions to human workers during assembly activities. The experimental research has been designed to have three different results: 'task completion time', 'quality', and 'mental workload'. With these results, we were able to study the impact of AR on the factors cycle time, products variety, quality and complexity of tasks. Based on Table 5, these are the factors for which AR have an impact.

The prototype of PAR, consisting of a Kinect camera, motion recognition software, work instruction programme and a projector, was developed and installed in the Logistics 4.0 Laboratory at the Department of Mechanical and Industrial Engineering at the Norwegian University of Science and Technology. To calculate the 'test completion time', 'quality' and 'mental workload' to understand the impact of AR on the factors, a test population consisting of 15 participants carried out a series of assembly tests of two components (Front wing and Side pod, Figure 8) of a LEGO product when being assisted by the instructions provided by the PAR. The 'test completion time' was measured by using a stopwatch. The 'quality' of the assembly activities was measured by counting the number of assembly and picking errors made by the participants. Finally, the 'mental workload' was measured using a simplified NASA-TLX questionnaire.



Figure 8. The Front wing (on the left) and the Side pod (on the right)

For the 'task completion time' and 'quality', the results showed the following when using the PAR:

- On average, the Front wing was assembled in 34.92 seconds, with an SD = 7.14 seconds, and the 'Side pod' in 58.36 seconds, with an SD of 9.45 seconds.
- On average, only one error was made when assembling both the components, that is, the 'Front wing' and 'Side pod'.

However, these results alone do not say much. Therefore, they have been compared with the results obtained from the other two assistance systems. Because in most manufacturing contexts paper-based manuals and workplace-mounted monitors assist human workers, the two factors 'task completion time' and 'quality' have been calculated also for the case in which paper-based and computer-assisted instructions were used to assemble the two components shown in Figure 8. The results obtained to assemble the two components using all three assistance systems are shown in Figure 9. As Figure 9 and the comments of the participants of the experiments show, it was possible to compare the three systems and contextualise the results of the PAR.



Figure 9. The 'task completion time' on the left and 'quality' on the right

Looking at the righthand side of Figure 9, the computer-assisted instructions achieved the best results in the 'task completion time' compared with the other alternatives (see Figure 9 left). In fact, the 'Front wing' was assembled in 29.98 seconds, with an SD = 5.35 seconds, and

the 'Side pod' in 46.19 seconds, with an SD of 6.79 seconds. When compared with the paperbased manual instructions, the PAR achieved almost the same 'task completion time' for the Front wing and achieved a better 'task completion time' for the 'Side pod'. Therefore, the PAR was not best option in terms of 'time completion time'. This could lead to the conclusion that AR has a negative impact on the factor 'cycle time'. In terms of 'quality', the results show that the PAR system was the best option to assist human workers compared with the other two assistance systems, reducing the possibility of errors during the assembly activity, especially for the more complex assembly task of the 'Side pod' (eight different parts to be assembled in eight work steps). In fact, regarding the 'Front wing', it was difficult for the participants to make errors, regardless of the assistance system, given the simplicity of the assembly activity (three different parts to be assembled in four work steps). This could lead to the conclusion that AR has a positive impact on the factors 'quality' and 'complexity of tasks'.

After experience with the three different assistance systems (paper-based manual instructions, computer-assisted instructions, and the PAR), the 15 participants were asked what they thought of them. Therefore, all the participants completed a simplified NASA-TLX questionnaire. The questionnaire asked to the participants to give feedback on perceived enjoyment, frustration, perceived ease of use, effort and mental demand for each assistance system; the results are shown in Figure 10. The PAR system obtained the overall best results and scored especially high in perceived enjoinment (86.7%) and ease of use (86%)Figure 10.



Figure 10. Results of the simplified NASA-TLX

The questionnaire helped us understand more about the impact of AR on the examined factors. The participants said that, although no one had previous experiences whit AR

solutions, they did not experience any difficulties related to the use of the PAR while assembling the two components. In fact, the participants found that the PAR was the easiest assistive system to use compared with those proposed and could help in assembling products that someone may have never assembled. Therefore, we can exclude that the results concerning 'task completion time' were derived from the participants' experience with the technology. However, the participants pointed out that how the PAR was developed and delivered were the main reasons because the results related to the 'task completion times' were worse than expected. In fact, in each work step, the participants were performing the following actions: pick the part, confirm the picking, assemble the part and confirm the assembly. Thus, there were two confirmation steps that were time-consuming and inevitably had a negative impact on the 'task completion times'. Therefore, the drawback preventing the PAR developed in this experimental research from being the fastest solution and ensuring better performance for companies in terms of time was that each task must be validated after it has been completed because there was no system able to detect the execution of the assembly process step by step and comprehend when one work was completed and the next was supposed to begin.

Therefore, the results from the experimental research that represent the third main outcome of the subsection were found to be the following:

- For what concern the factor 'cycle time', the current study has confirmed the uncertainty found in Table 5 ('Decreases/Increases'). The reason behind this result was mainly related to how the AR was developed and then delivered to the final users (readers can check Paper 4 for more information). Therefore, if a company wants to use AR in its ALS to reduce time spend on a procedure, it must create a solution that does not present time-consuming steps, such as the ones explained in the abovementioned solution, and that would be user friendly, meaning that even human workers who never experienced AR can use it without problems.
- For the factor 'products variety', the results have shown a positive effect found, as shown in Table 5 ('Increases'). In fact, although the participants never assembled the two components assembled during the experiments, they said that, thanks to the PAR, they were able to do it without any problem. Therefore, if companies decide to use

AR as an assistive system in their ASs, they can also increase the variety of products assembled in their AS.

- Regarding the factor 'quality', there was a positive effect, as shown in Table 5 ('Increases'). In fact, the PAR was able to reduce the assembly and picking errors made by the participants, thus reducing the number of defective assembled products and increasing the 'quality' of the ALS. Moreover, from the experimental research, there was an increased benefit when constructing more complex components, increasing the benefit of using the AR when assembling them (see Figure 9 right).
- Regarding the factor 'complexity of tasks', there was a positive effect found, as shown in Table 5 ('Decreases'). This result is directly connected with the increase in quality of the final products. In fact, the participants felt that, thanks to the AR, the assembly process was easier than when they performed it with the other two assistive systems because, with the AR, everything they had to do was clearly marked and guided, minimising the possibility of making mistakes.

4.3. Framework to design the workplace of an assembly line subsystem by using virtual reality and motion capture system

This subsection aims to answer **RQ2**, which was derived from TL1, Methods for ergo-efficient workplace design:

How can ergo-efficient AS4.0 workplaces be designed?

At the tactical level, once companies have decided the configuration of their ALSs, it is time to design the workplaces in which the operators will perform the assembly tasks (Table 1). Designing a workplace is not an easy process, especially considering the time- and resourceconsuming AS workplace design procedures that are conventionally used (e.g., physical mockups and computer-aided systems). However, with the advent of Industry 4.0, firms can facilitate this process (Burggräf et al., 2019). In fact, new solutions have started to emerge to accelerate the AS workplace design procedures that use I4.0 technologies. Specifically, the combined use of VR and mocap systems has been considered very promising technologies. VR allows for the creation of a three-dimensional world in which users can interact with threedimensional objects in real time using their natural senses and skills (Riva, 2002). A mocap system allows for the creation of a digital copy of the operator in real time because of 'a virtual representation of the skeleton and its movements' (Bortolini et al., 2020), facilitating and improving the ergonomic assessments because AS workplace designers have access to the exact movement over time of the position and orientation of the operator's different limbs in a common reference system (Oyekan et al., 2017). The use of VR and a mocap system during the design phase of an AS can eliminate the main constraints of conventional AS workplace design procedures: namely subjective and time-consuming assembly time measurements and ergonomic assessments (Battini et al., 2011; Battini et al., 2014). The combined use of these two technologies enables fast and reliable assembly time measurements and ergonomic assessments. Although many researchers have indicated its potential (Peruzzini et al., 2017; Vosniakos et al., 2017; Michalos et al., 2018; Caputo et al., 2017; Battini et al., 2018), none have suggested a clear methodology to follow when designing AS workplaces using VR and mocap system. Therefore, we first identified the steps to design an ergo-efficient workplace of an ALS using a VR and mocap system; then, we verified its validity through experimental research at the Logistic 4.0 Lab at NTNU. Before and during the steps identification process, we were in contact with the developer of the software tools Siemens Jack[™] to get feedback on how best to use their solution and understand the reasons why companies contact them. From the discussions we had, one of the reasons why companies were contacting them was because they were curious about how to design their ALSs using their software tool with VR and mocap system. Although the companies were asking questions about the software and technologies, in the end, most were not interested in buying anything because they did not know how to properly use the technologies. Therefore, it is from the discussions with the software developer that we were inspired to create the methodological framework to design an ergo-efficient workplace of an ALS using VR and a mocap system.

The steps to design ergo-efficient AS4.0 workplaces. The five-step methodological framework can guide the design process of ergo-efficient AS workplaces and represents the first main outcome of this subsection; this framework also answers **RQ2** and is illustrated in Figure 11.



Figure 11. Methodological framework; the dashed arrows represent the steps that may change during the iterative process Below, the core description of each step is reported. For a detailed description of the steps, the reader is referred to Paper 5.

Step 1 – Input collection

The first step is to gather the input parameters. As previously stated, the methodological framework developed herein is intended to be used in the detailed design phase, so the (i) AS layout configuration, (ii) cycle time (paced/un-paced), (iii) workstation type (i.e., open/closed, parallel/serial, two-sided, ...) and (iv) automation level (i.e., percentage of automation, type of equipment, ...) are all known. Furthermore, those product features that impact ergonomics

and safety (e.g., component weights) are discovered during this design phase. Moreover, the operators who will operate on the AS workplace under construction are well known, as are their anthropometric data and age.

Step 2 – Workplace design

Once the inputs have been gathered, the AS workplace design can begin. Unlike the traditional AS workplace design approach, which requires physical mock-ups, the workplace is developed virtually, by utilising a 3D modelled environment. Many distinct options may be quickly produced and evaluated without the need to construct a physical mock-up for each one. As previously noted, this has significant time and cost savings benefits.

During this phase, it is fundamental that the AS workplace designers are assisted by experienced operators. Indeed, as is known from the literature, operators experience a gradual decline in both physical and cognitive abilities as they age (Shepard, 2000; Bouma, 2013; Bures and Simon, 2015), and experienced operators can provide useful advice to AS workplace designers regarding taking these aspects into account. Furthermore, throughout this phase, it is critical that AS workplace designers and experienced operators take a holistic approach, viewing productivity and operator well-being as two complementing components rather than as two distinct aspects. The AS workplace layout, in fact, influences both productivity and operator well-being throughout the assembly process.

It is worth mentioning that, although numerous software tools are available to create a 3D environment, only those that can communicate with the mocap system and VR (e.g., Siemens Jack[™]) may be utilised. The interface with the mocap system is vital because it allows for the exact replication of the operator's real-world movements in the 3D modelled environment.

Step 3 – Data collection

Data gathering can begin after the virtual workspace has been completed. In this step, an operator must be outfitted with a mocap system and VR. The operator will be immersed in the 3D-designed world and able to move and interact with it. In this approach, the operator can simulate the assembly process without the necessity for a real mock-up.

Step 3 involves creating a DT of the operator carrying out the virtual assembly process. In this way, it is ensured that the anthropometric data of the operator's DT corresponds to those

from the operator who will perform the actual virtual assembly process. Moreover, in this step, it is recommended that the operator chosen to carry out the virtual assembly process is representative in terms of the anthropometric data of the operators who will work on the AS workplace under development.

Step 4 – Data analysis

Once the data have been obtained, they must be analysed with respect to productivity and ergonomics. Productivity can be measured by using time-related parameters (e.g., task execution times), whereas ergonomics can be measured by using ergonomic indices (RULA, REBA, NIOSH, OWAS, etc.). During the data analysis step, in order to optimise the outcomes, the AS workplace designers are required to divide the whole assembly process into the different assembly tasks that constitute it. Thus, each assembly task has to be associated with its own data (mocap-based recording and ergonomics-relevant data). As a result, each assembly activity must be coupled with its own data (mocap-based recording and ergonomics-relevant data). In this way, AS workplace designers may assess the productivity KPIs (Key performance indicators) and ergonomics index ratings for each assembly activity.

During the ergonomic assessments, it is crucial that the age of the operator is explicitly considered. In particular, it is advised to use a preventive approach, in which the age of the oldest operator who will work on the AS workplace under development is considered because the operator utilised for virtual assembly does not have to be the oldest operator who would work on the AS workspace under development; it might be a younger coworker. The reason for this is derived from the second outcome of this subsection, which will be explained later.

Step 5 – Ergo-productivity satisfaction

The productivity KPIs and ergonomic scores obtained from the previous step serve as an input in this step: ergo-productivity satisfaction. This is a decision step in which the AS workplace designers must determine whether the productivity KPIs and ergonomic scores are adequate (in terms of the company's requirements, legal restrictions and so on). If they are, this is the final AS workplace design; if not, the user must return to Step 2. It should be emphasised that, if the user must return to Step 2, until the operator performing the virtual assembly procedure changes, the operator's DT, which comes in at Step 3, is not need to be created (or updated) every time an iteration occurs. To conclude with the first main outcome of this subsection, the results of a practical example of how to use the methodological framework have been reported. A simple but representative case study was carried out at the Logistic 4.0 Lab at NTNU. Specifically, we applied the methodological framework for redesigning an AS workplace where a medium size pump, which is shown Figure 12, comprising 25 components assembled through 17 tasks was produced (readers can refer to Paper 4 to see more about this case study).



Figure 12. Pump assembled in the case study

The case study was carried out by a second-year PhD student (27 years old) who designed, based on his experience, the initial workplace; from now on, we will refer to this workspace as the 'as-is configuration', which is where the pump was supposed to be assembled. Once the workplace was ready, the PhD student carried out the assembly process, during which the data necessary to measure the task execution times and REBA index were collected according to Step 3 and then analysed according to Step 4 of the methodological framework. Specifically, we considered the task execution times and REBA index as the productivity KPI and ergonomics index, respectively. Moreover, for the purpose of showing the importance of considering the age of the operators in the ergonomic assessments, we considered a fictitious case where the age of a second case study operator was 60 years old. This second case is called fictitious because, unlike the first operator, there was no real person who performed the assembly tasks at the workplace, but we considered only the age to calculate the REBA index. In fact, the ages of the operators (27 years and 60 years old) were explicitly considered in the ergonomics assessment by means of the age-based multipliers $f_{strength}$ and $f_{flexibility}$, which are going to be explained later as part of the second main outcome of this subsection. From the study of the 'as-is configuration', here considering both the ages (27 and 60), it has been possible to calculate the total task execution time and total average REBA score, as shown in Table 6. Moreover, because of the study of these two KPIs, for each task executed, it has been possible to identify the tasks that reduced the productivity (i.e., tasks with low value-added ratios) and operators' well-being (i.e., tasks with high ergonomic risks).

	Task execution time (s)	Average REBA score			
		Age = 27	Age = 60		
Total	758.6	4.12	4.97		

Table 6. Total task executed time and total average REBA score of the 'as-is configuration' of the AS workplace

After collecting this information, we then proceeded with the redesign of the AS workplace, here according to the methodological framework developed herein. The workplace has been redesigned based on the advice of an expert, a full professor; then, the same two KPIs were calculated by assembling the same product but this time in the redesigned workplace, as shown in Table 7.

Table 7. Total task executed time and total average REBA score of the 'redesign configuration' of the AS workplace

	Task execution time (s)	Average REBA score			
		Age = 27	Age = 60		
Total	641.6	3.74	4.24		

Therefore, comparing the results form Table 6 and Table 7, it is possible to see that, because of the use of VR together with a mocap system and following the framework we developed, the total task executed time (758.6 s vs. 641.6 s) and total average REBA score (4.12 vs. 3.74 for the 27-year-old operator and 4.97 vs. 4.24 for the 60-year-old operator) in the redesigned configuration were lower than in the as-is configuration. These results have confirmed the validity of the developed methodological framework in designing ergo-efficient AS workplaces using VR together with a motion capture system.

The inclusion of the age of human workers in the design of an ALS workplace. In Step 4 of the methodological framework, it is recommended that during the ergonomic assessment the age of the operator be explicitly considered. However, how to consider this is still overlooked in the literature, with only Wolf and Ramsauer (2018) proposing a solution. Therefore, what is

presented next is a tentative method to fill this gap, here also representing the second main outcome of this subsection.

The age of the operator in the design process of ergo-efficient ALS workplaces can be considered in two different modes:

The first mode is more dynamic and involves the most experienced 'senior' operators who work in the ALS. Experienced operators, because of the experience they gained over the years, can provide advice that can simplify and speed up the design procedure (Di Pasquale et al., 2020); during Step 2, they can suggest to the AS workplace designers whether (i) the operators need to be supported by new equipment (e.g., lifter, automatic screwdrivers, collaborative robots, etc.), (ii) where it is better to place the components and (iii) the environmental conditions (e.g., lighting) that need to be adapted or not based on their age.

We can define this mode as dynamic because different experienced operators can give different suggestions on (i), (ii) and (iii), even if they have the same years of experience in the ALS. However, the fact that the suggestions are not all the same can be a positive. In fact, during the iterative design process, if a suggestion during Step 2 (workplace design) did not give a positive result, it can be changed to another during the next interaction.

• The second mode is more static and required 'only' two formulas. These two formulas are derived from the work of Wolf and Ramsauer (2018).

The first formula, as reported in Equation (1), was developed by fitting data from the literature on variations in muscle strength with age to evaluate an age-based multiplier ($f_{strength}$):

$$f_{strength}(\%) = 0.00058 \cdot age^3 - 0.08478 \cdot age^2 + 3.24439 \cdot age + 62.92006 \text{ (1)}$$

The second formula, as reported in Equation (2), was developed by fitting the joints flexibility reduction data reported in Table 1 of Wolf and Ramsauer's work (2018); it evaluate the age-based multiplier for joints flexibility reduction ($f_{flexibility}$), which is needed when the ergonomic indexes contain assessments on joint flexibility (e.g., allowable joint angle limits in the REBA index):

 $f_{flexibility}(\%) = -0.00019 \cdot age^3 + 0.034286 \cdot age^2 - 2.538095 \cdot age + 145 \text{ (2)}$

By considering these age-based multipliers in the ergonomic assessments in Step 4, it is ensured that the age of the operator is explicitly considered.

Because in the formulas age is represented only as a number, it does not matter if the operator who physically performed Step 3 (data collection) is the oldest operator working in the ALS. In fact, the decision of the age of who will perform Step 3 will depend on a trade-off between the workforce characteristics (in terms of technological skills) and the assembly process under consideration (i.e., whether the assembly process is affected timewise by the age of the operator). However, it is recommended that, in moment of the ergonomic assessment (Step 4), and, thus, the moment to use (1) and (2), the age of oldest operator that work in the ALS should be considered in the formulas. The reader is referred to Appendix A of Paper 5 to see an example of how the formulas here have been applied. We can define this mode as static because it considers age as a number, so the results obtained using the same age at different interactions of the design process will always be the same.

It is important that, during the design process of the ALS workplaces, both modes, dynamic and static, are considered.

Because of the methodological framework proposed in Figure 11, companies have a simple but effective guide to follow when they are looking to design ergo-efficient workplaces in their ALSs using VR and a mocap system. The methodological framework allows for the maximisation of both production and operator well-being through a holistic approach. In fact, it does not focus only on productivity: it also makes it possible to consider the current labour market because it allows for the inclusion of ageing employees and their related advantages, along with their downsides, in the AS workplace design approach.

4.4 Decision support system for the selection of the best material management solution among traditional Kanban, electronic Kanban and digital twin

This subsection aims to answer **RQ3**, which was derived from OL2, Methods for efficient control of AS4.0:

How can material management be controlled in AS4.0?

At the operational level, companies must control their ASs (Table 1). In particular, in the current study, we were interested in the case in which companies must decide how to control the material management in their ALFS by considering the scenario where components are stored at the workplaces in bins (which is one of the most frequently adopted (Battini et al., 2009)). Based on the literature, different material management solutions have been developed over the years that take into consideration this scenario. The first material management solution was represented by the traditional Kanban (Huang and Kusiak, 1996), in which the replenishment of components at the workplace is controlled by physical cards applied to the bins: once the bins are empty, the Kanban card is removed and sent manually to the supermarket warehouse to request new components (Singh et al., 1990; Junior and Godinho Filho, 2010). The traditional Kanban solution, then, evolved into the electronic Kanban (e-Kanban) solution, where the need for replenishment of the component at the AWs was still triggered manually but in a digital way (e.g., by pressing a button or by scanning a barcode or QR code, etc.) (Kouri et al., 2008). Although there are pros in adopting Kanban and e-Kanban solutions, these are accompanied by a common limitation that is represented by the fact that the inspection of empty bins and requests for the replenishment of components need to be manually executed from the operators, resulting in inefficiencies (e.g., loss of Kanban cards, delayed triggered of replenishment, etc.). To overcome these inefficiencies, researchers and practitioners have been evaluating the potentialities provided by certain I4.0 technologies (e.g., sensors, computer vision, DT, etc.) for the development of new material management solutions (Bortolini et al., 2017; Cohen et al., 2019b; Dolgui et al., 2022). Among these technologies, DT appears to be the most promising and disruptive. A DT is 'an integrated simulation technology, which aims at developing a model of the environment that has to be fed with real-time data, to provide high fidelity of the overall system' (Saporiti et al., 2020); a DT-based material management solution allows knowing in real time the status of the components in the bins, which can automatically trigger the replenishment of components,

hence simulating what can happen in the near future in terms of the status of components in the bins and forecasting when is the best moment to carry out replenishment (Qi and Tao, 2018).

However, despite the importance of the material management activity, managers and practitioners must choose the proper material management solution to adopt in their ASs. This was confirmed by the managers from Bossard (2022), a company that provides components and digital solutions for the management of components. During discussions with them, it emerged that most of their customers decided on the solution to control the material management in their ALFS by following their own experience rather than proper guidelines or tools that can tell them what is the best solution to choose based on their situation. Therefore, we have developed a DSS that can guide managers and practitioners in choosing the most convenient material management solution among traditional Kanban, e-Kanban and DT-based material management solutions to adopt in their ALFSs.

To develop the DSS, a three-step procedure must be followed:

- Step 1: development of three simulation models, one per each material management solution considered (traditional Kanban, e-Kanban and DT-based solutions).
- Step 2: parametrical analysis, from which it is possible to have an output of man hours and average inventory level required for different scenarios (where a scenario corresponds to a certain combination of parameters adopted in the models). Specifically, we varied the parameters' mean demand of final products *Q* and standard deviation of the demand *σ*, the distances between the supermarket warehouse and the assembly area *d* and the number of assembly lines *L* and assembly workplaces *W*, which are the parameters that are commonly considered when companies choose the material management solution to adopt in their ASs (Houti et al., 2017; Urru et al. 2018; Sapry et al. 2020). From the parametrical analysis, we obtained 243 different scenarios per each material management solutions considered.
- Step 3: development of the DSS that, given the unitary cost of the stocks of components, suggests the most convenient material management solution, here based on the investment and operational costs for each scenario considered.

72

For a detailed explanation of the DSS development, the reader is referred to Paper 6. Below are only reported the results required to reach the main outcome of this subsection.

The DT-based model. Because the only example of a DT-based solution for material management that can be found in the literature was the one proposed in the conference proceedings by Kim et al. (2021), the first main outcome of this subsection concerns the development of a DT-based model.

In the DT-based model, the DT is responsible for the determination of the best moment (best time interval) when a feeder must go to replenish the components in the workplaces. In fact, the DT monitors in real time the availability of components in the bins through the use of sensors and scales. In this way, once a bin falls under the desired replenishment level, a signal is immediately sent to the DT. When this situation occurs, the DT activates a system, which performs a simulation based on the current state of the workplaces. This analysis can simulate the assembly processes and forecast the consumption of components in a very precise and accurate way. The DT stops the simulation whenever a component reaches a status of prestarvation, that is, when it is very close to the starvation point. The starvation point is reached when, in the bin, there are enough components to assemble a maximum of two products. Then, the DT records the stopping time and collects all the signals of replenishment that occurred in the simulated period. Finally, the DT communicates to the feeders the time in which they will have to perform the replenishment, as well as what bins will need to be refilled. The process of the developed DT-based model is depicted in Figure 13.



Figure 13. Digital twin-based model process representation

In this way, the DT-based model determines the best replenishment time (best time interval), which will be variable and depend on the current state of the system, as well as on the demand profile. In this way, the operators in the workplaces, compared with the cases of traditional Kanban and e-Kanban, do not perform any non-value-added activity linked to the replenishment of the workplaces as it is completely managed by the DT (i.e., the inspection of empty bins and requests for the replenishment of components is not manually executed by the operators).

The decision support system (DSS). Once the DT-based model has been determined and simulated, together with the results of the simulations of the other two material management solutions (traditional Kanban and e-Kanban), it was possible to develop the DSS. The DSS represents the second main outcome of this subsection, which also answers **RQ3**.

The aim of the DSS is to support managers and practitioners in identifying the most convenient material management solution, here based on investment and operational costs, between traditional Kanban, e-Kanban and DT-based solutions, depending on their situation in terms of the mean demand of final products Q and standard deviation of the demand σ , the distances between the supermarket warehouse and assembly area d, and number of

assembly lines *L* and assembly workplaces *W*. The DSS was built to be a tool that is easy to read and interpret. In fact, if we consider in all the figures below (Figures 14, 15 and 16) the same case where $\sigma = 10\%$, d = 20, L = 2 and W = 4, we have companies that should choose a different material management solution based on the unitary cost of stocks © in their ALFSs, as shown in Table 8.

	Mean demand of final products			
Material management solution	Q = 10 pcs/h	Q = 20 pcs/h	Q = 40 pcs/h	
Traditional Kanban	0.01 €/pc < c ≥ 0.70 €/pc	0.01 €/pc < c ≥ 0.30 €/pc	0.01 €/pc < c ≥ 0.10 €/pc	
E-Kanban	0.70 €/pc < c ≥ 8.10 €/pc	0.30 €/pc < c ≥ 3.10 €/pc	0.11 €/pc < c ≥ 1.90 €/pc	
DT-based solution	c > 8.10 €/pc	c > 3.10 €/pc	c > 1.90 €/pc	

Table 8. Decision of the material management solution in different scenarios based on the unitary cost of stocks c

Moreover, the DSS has been analysed to draw some considerations that can be made when analysing the impact of the single parameters of the scenarios.

First, considering the demand parameter, ceteris paribus, a rise in the demand level negatively affects the convenience of the traditional Kanban solution, as supported by Houti et al. (2017). Indeed, the mean value of the extent of the convenience range of the traditional Kanban solution drops from $1.08 \notin pc$, when considering the lowest level of demand, to 0.27 $\notin pc$, when considering the highest level of demand (-75%). The same behaviour can be noticed also for the convenience of the e-Kanban solution (-60.4%). On the other hand, the convenience of a DT-based material management solution is generally higher when considering a remarkable level of demand (+36.4%).

Second, considering the standard deviation of the demand parameter, the convenience of the traditional Kanban solution follows the same behaviour of the first parameter, but with a more moderate impact when considering the change in value, which can be noticed when comparing the highest standard deviation with the lowest (-4.6%). For the convenience of the e-Kanban solution, it presents the highest level in correspondence to the highest variability, while the lowest level can be associated with the middle value of the standard deviation parameter.

The DT-based solution presents the highest level of convenience when considering a middle value of standard deviation, while it shows the lowest level in correspondence to the highest value of variability. However, as for traditional Kanban solution, the impact of the

standard deviation of the demand on the convenience of the e-Kanban and DT-based solutions is quite limited.

Third, considering the parameter linked to the distance between the supermarket warehouse and assembly area, both for the traditional Kanban and e-Kanban solutions, the highest level of the convenience extent is reached at the middle level of distance. In the case of a DT-based solution, this value is reached at the lowest value of distance. However, in this situation, the impact of this parameter on the convenience of the material management solutions is quite limited.

Fourth, the last two parameters, that is, the number of AWs and number of lines, affect the convenience of the material management solutions in a very similar way. Indeed, regarding the traditional Kanban solution, the convenience is negatively influenced by the rise in the number of AWs and lines at -78.6% and -80%, respectively. A similar behaviour can be noticed when considering the convenience extent of the e-Kanban solution, with a negative influence associated with a rise in the number of AWs and lines equal to -61.6% and -63.25%, respectively. In the case of DT-based solution, the behaviour is completely the opposite. Indeed, an increment in the value of AWs and lines can be associated with a rise in the convenience of this solution by 35% and 37%, respectively.

Finally, to enhance the results of the study and understand the statistical relevance of all the parameters exploited to develop the DSS, an analysis of variance (ANOVA) has been carried out, as shown in Table 9.

Table 9. ANOVA p-value results

	Material management solution			
Parameter	Traditional Kanban	e-Kanban	DT	
Q	5.72e-12	3.36e-15	1.79e-15	
σ	0.958	0.685	0.747	
d	0.479	0.937	0.896	
W	6.98e-16	3.85e-13	4.11e-14	
L	2.46e-16	8.86e-15	1.04e-15	

The table shows that the parameters standard deviation of the demand (σ) and the distances between the supermarket warehouse and assembly area (d) do not present a statistically significant impact on the convenience range extent of the DSS at any confidence level. Therefore, because these two parameters were statistically not relevant, they have been excluded from the DSS and its final version, which only considers three parameters that were found to be statistically significant—Q, L and W—as reported in Figure 17.

To conclude this subsection, the DSS can help managers and practitioners thinking about changing the material management solution adopted in their ALFS from a traditional or e-Kanban solution to a DT-based one or in creating a new DT-based solution to control the material management in their ALFS. The DSS has shown that traditional Kanban and e-Kanban solutions are more convenient in environments where the number of workplaces and assembly lines, demand level and average cost of stocks are low and medium, respectively, whereas the DT-based solution is more convenient in environments characterised by a high number of AWs and assembly lines, with a high demand level and with a high average cost of stocks. Moreover, companies should not pay much attention to the standard deviation of the demand (σ) and of the distance between the supermarket and assembly area (d) when planning which material management solution to adopt in their ASs. In fact, although these two parameters are commonly considered when companies have to choose the material management solution to adopt, based on the ANOVA analysis, they resulted as not being not statistically significant when deciding which material management solution.















Figure 17. Final DSS after the ANOVA

4.5. General discussion of the results

So how can companies move from traditional AS to AS4.0? How can researchers continue the research stream in the design and management of AS4.0? From the results of the present research study, a few reflections and recommendations can be offered.

Implementing 14.0 technologies in ASs can be problematic for companies (Karadayi-Usta, 2019; Mahmood et al., 2021). However, we believe that companies can significantly benefit from the results that have been presented in the current study (Tables 3 and 4) to ease the implementation of Industry 4.0 in their AS. Tables 3 and 4 can be used as a guide to move from a traditional AS to an AS4.0, here by following the decisions that they will have to make for each technology during all the steps of their implementation (from the strategic to operational level). It is possible that more decisions can be discovered and added to the two tables, especially with the increased knowledge about the implementation of I4.0 in AS4.0.

At the strategical level, the decision of the configuration of the ALS is one of the first decisions that companies must make (Battaïa and Dolgui, 2013; Battini et al., 2011). However, there was still no clear idea of how the configurations should be chosen. Therefore, our investigation identified the relevant factors that companies need to know when choosing the configuration of their ALSs. The identified factors are the ones that also determine if a configuration can be dynamic reconfigurable. Moreover, the present study, in identifying the impact of 14.0 technologies on these factors, has shown how the implementation of 14.0 technologies can open opportunities for companies to choose new ALS configurations that should even be more dynamic reconfigurable. However, at the moment, these results are qualitative in nature, so it is not possible to say precisely when one configuration can be. Therefore, it will be challenging to find the threshold values of the factors that determine when one should choose a configuration instead of another in the case without 14.0 technologies and then to see how the identified threshold values will change when one or more 14.0 technologies are implemented in the ALS.

At the tactical level, companies must design their ALSs. I4.0 technologies can support companies to do design their ALSs (Peruzzini et al., 2017; Vosniakos et al., 2017). In the present study, a methodological framework to design ergo-efficient ALS workplaces using VR

and a mocap system has been developed for the first time. This framework can help companies in designing their ALS workplaces by following five simple steps. The strength of the developed methodological framework is that it can be followed using all types of VR devices and all types of mocap systems available on the market, making it accessible to all companies. Moreover, the study has identified how to include the age of human workers in the design of the ALSs' workplaces. The results from the methodological framework were not focusing only on the productivity of the ALS, but also aiming to make it possible to consider the current labour market, including ageing employees and their related advantages and downsides in the ALS workplaces design process. Although the experimental research that was performed to validate the developed framework gave positive results, because of the simple case investigated, there was a need for more complex cases that would not only help to confirm the validity of the framework, but also continuously improve it.

At the operational level, companies need to know how they will control the material management in their ALFS (Battini et al., 2009). I4.0 technologies—especially DT—can open new opportunities to control the material management (Bortolini et al., 2017; Dolgui et al., 2022). However, it deciding to use a DT-based solution is not always going to be the best solution. In fact, if companies consider the costs (investment and operational costs) of their material management solutions, it is possible that other solutions like traditional Kanban or e-Kanban can be more convenient. Therefore, the DSS developed here has aimed to give to companies a tool that allows them to choose the best material management between traditional Kanban, e-Kanban and DT-based solutions. The DSS is not only a tool that is easy to read, but it is also easy to develop. However, at the moment, the results related to the DSS are based only on simulations. Results derived from real case studies can increase the quality and validity of the DSS and can better define the ranges of when one material management solution is more convenient than another.

Finally, researchers interested in this topic will be able to find—primarily through the nine future research opportunities identified—ways to increase the knowledge about the design and management of AS4.0. Even with the relevance of the presented results, the current research study is only a starting point for what can be a prosperous future of research. The continuation of the research on the design and management of AS4.0 can pave the road to the easier implementation of I4.0 technologies in AS.

83

5. Conclusions

This section summarises the research study and provides contributions to theory and implications for practice. Furthermore, the research limitations are addressed, as well as recommendations for future research.

5.1. Summary

The objective of the current research was to support and create new knowledge for academy and managers and practitioners who want to design and manage AS4.0. In addition, an aim was to propose future research opportunities related to the design and management of AS4.0 to inspire new researchers to continue investigating the topic. To do so, the study focused on three decision areas related to the design and management of an AS (configuration of an ALS, workplace design of an ALS and control of the ALFS) and on three of the I4.0 technologies (AR, VR and DT). Moreover, to develop a richer and more thorough knowledge of the research area, a mixed methods approach was used, integrating components of qualitative and quantitative research:

- The SLR provided in-depth knowledge about the state of the art of AS4.0 while also contributing possible future research opportunities that inspired the research questions investigated in the present study.
- Exploratory research enabled the identification and definition of the relevant factors that companies need to know when they have to choose the configuration of their ALSs.
- Experimental research for investigating the impact of AR in the previous identified factors and for helping the validation of the developed methodological framework to design ergo-efficient workplaces of an ALS.
- Simulation modelling allowed the analysis of different material management solutions (traditional Kanban, e-Kanban and DT-based solutions) to identify which is the best for controlling the material management in ALFSs.

Therefore, the SLR made it possible to understand what decisions companies need to make when they decide to implement one or more of the I4.0 technologies into their AS, showing nine future research opportunities related to AS4.0. The future opportunities not only inspired the three research questions answered in the present research study, but they can also inspire many other researchers to continue exploring the topic by creating new knowledge.

Moreover, using the other three research methods (exploratory research, experimental research and simulation modelling) in the studies carried out for this research enabled us to answer the three research questions, thus providing the following contributions:

RQ1: How can AS4.0 configurations be affected by Industry 4.0 technologies?

The findings of Paper 3 revealed the factors that companies need to know when they have to model the configurations of their ALSs and the impact of I4.0 technologies on these factors. The seven identified factors are cycle time, number of tasks, network density, products variety, volumes variety, quality and complexity of tasks. The knowledge of these factors is of considerable importance when companies want to model dynamic reconfigurable configurations of their ALSs since their values determine if a configuration is dynamic reconfigurable or not. Regarding the impact of I4.0 technologies on these factors, except for the factors of number of tasks and network density, where no impact was identified, the technologies decreased or increased these factors, hence being beneficial for the ALSs. Moreover, the most interesting result was related to the factor cycle time for AR. Here, it is not clear if AR is able to decrease the cycle time or if it increases it. Intrigued by this result, Paper 4 tried to see if it was possible to have a clearer idea of the impact of AR on the factor cycle time and other factors. Although the findings of Paper 4 have confirmed the uncertainty of this theory, we found that this this uncertainty was more caused by how the AR technology was developed than by the experience that the operators of the ALSs had using the technology. Furthermore, the findings of Paper 4 have shown that AR can help operators assemble products they have never assembled before and that the more complicated the product is to assemble, the greater the quality that can be achieved using AR because AR allows for fewer errors in the assembly process.

RQ2: How can ergo-efficient AS4.0 workplaces be designed?

The findings of Paper 5 revealed the methodological framework that designers should follow if they want to design ergo-efficient workplaces of an ALS using VR together with a motion capture system. The methodological framework consists of five steps: input collection, workplace design, data collection, data analysis and ergo-productivity satisfaction. When arriving at Step 5 (ergo-productivity satisfaction), designers need to see if the productivity KPIs and ergonomic scores obtained from Step 4 (data analysis) satisfy the company's requirements. If they do, the design process ends, and the company puts into place the final ALS workplace design; if not, the designer must return to Step 2 (workplace design). Therefore, the design process proposed by the methodological framework is an iterative process that stops only when the productivity KPIs and ergonomic scores are adequate (in terms of the company's requirements, legal restrictions and so on). Moreover, in the methodological framework, the age of the operators has been considered in two different modes: a dynamic mode that involves the most experienced 'senior' operators working in the ALS and a static mode that involves two formulas: one that calculates the age-based multiplier for muscle strength and the other for the age-based multiplier for reduced joint flexibility.

RQ3: How can material management be controlled in AS4.0?

The findings in Paper 6 have revealed a DSS that can help managers and practitioners thinking about changing the material management solution adopted in their ALFS from a traditional or e-Kanban solution to a DT-based one or who want to create a new DT-based solution to control the material management in their ALFS. The DSS suggests that traditional Kanban and e-Kanban solutions are more convenient in environments where the number of workplaces and assembly lines, the demand level and the average cost of stocks are low and medium, respectively, whereas the DT-based solution is more convenient in environments characterised by a high number of AWs and assembly lines, with a high demand level and high average cost of stocks. Finally, based on the ANOVA analysis, two of the parameters considered (the standard deviation of the demand and the distance between the supermarket warehouse and the assembly area) to develop the DSS were not statistically significant when deciding which material management solution to choose between the traditional Kanban and e-Kanban and DT-based solution.

87

5.2. Contributions to theory

The current research study has several contributions to theory, which are summarised in the seven main outcomes reported in Table 10.

Main outcomes		Paper				
	1	2	3	4	5	6
State of the art of AS4.0	х	х				
Future research opportunities for AS4.0	х	х				
Factors for the modelling of dynamic reconfigurable configurations of an						
Assembly Line Subsystem and the impact of Industry 4.0 technologies on these			х	х		
factors						
Identification and definition of the factors relevant for modelling dynamic						
reconfigurable configurations of an ALS and identification of how Industry 4.0			х			
technologies impact those factors						
Investigation of the impact of augmented reality on the identified factors				х		
A framework to design the workplace of an Assembly Line Subsystem by using					v	
Virtual Reality and a motion capture system					^	
Identification of the steps to design ergo-efficient workplace of an ALS by				х	v	
using virtual reality (VR) and motion capture (mocap) system					^	
Identification of how the age of human workers can be included in the design					v	
of a workplace of an ALS.					^	
A Decision Support System to select the best material management solution						v
among traditional Kanban, electronic Kanban, and Digital Twin						^
Development of a digital twin-based model for the control of the material						v
management in an ALFS.						^
Identification of the best solution among traditional Kanban, electronic Kanban						
and DT to control the material management in an ALFS.						х

Table 10. Contributions to theory: main outcomes

The first and second contribution of comes from the findings of the SLR that support the organisation and unification of knowledge related to the design and management of AS4.0. The decisions that companies have to make when implementing I4.0 technologies in their AS have been identified, together with the need for more research, as summarised in nine future research opportunities, which can open numerous research areas related to the design and management of AS4.0. Three of the future research opportunities inspired the three research questions answered in this research study.

The third contribution is related to the identification and definition of the factors relevant to know when companies want to model dynamic reconfigurable configurations of their ASs. Furthermore, analysing the literature helped understand the impact of I4.0 technologies in the identified factors. The implementation of I4.0 technologies in ASs can play a crucial role

when it is time to choose the configuration of the AS and it can help in the creation of dynamic reconfigurable configurations. Based on our results, I4.0 technologies may change the decision regarding which AS configuration to choose compared with the case where no technologies are implemented in the AS. Finally, the fourth contribution, which was derived by experimental research, supported the results discovered from the literature related to AR, here in relation to the impact of this technology on the identified factors. The third and fourth contributions have addressed RQ1.

The fifth contribution helps explain how it is possible to design ergo-efficient ALS workplaces using VR together with a mocap system. The five-step methodological framework can be applied using all VR devices and mocap systems available on the market. Furthermore, the sixth contribution helps identify how to consider the age of operators during the design process of an AS workplace. The fifth and sixth contributions address RQ2.

The seventh contribution is related to the development of a DT-based model for controlling the material management activity in ALFS. Finally, the eighth contribution supports the decision regarding the solution for controlling the material management in an ALFS between the traditional Kanban and e-Kanban and DT-based solutions, hence providing the DSS. The seventh and eighth contributions address RQ3.

5.2. Implications for practice

For managers and practitioners who are responsible of the design and management of ASs, the findings of the present research have several relevant implications. ASs need to achieve and maintain their goals (e.g., have high efficiency, maximise throughput with minimum resources, have high flexibility, rapid change in production volume and type of products to be assembled, minimise the WIP, assemble high-quality products, minimise the inventory level, effectively utilise the workforce and minimise disruption to production) while achieving high profits and keeping costs low. To do this, great potential lies in implementing new 14.0 technologies in the ASs. Therefore, the current research first presented how companies can implement 14.0 technologies in their ASs. The study identified the decisions that companies must make at each level (strategic, tactical and operational) and for each technology possible to implement in ASs. This is an important step to help companies that must make a difficult decision like the one to buy one or more technologies. Second, because companies must use

the technologies they purchase, they need to know what the impact of these technologies may be in their ASs. In particular, the current research has focused on the impact that I4.0 technologies can have in the decision of the configuration of the ASs. Therefore, the present research first highlighted the seven factors that companies need to be aware of when they want to model a dynamic reconfigurable configurations of their AS and then gave insights into the impact of I4.0 technologies on these factors. Third, the current research has described the five steps that form the methodological framework that companies can follow to design ergoefficient AS workplaces using VR. The methodological framework can guide companies deciding to use VR together with a mocap system to design the workplaces of their ASs and that, at the same time, want to take into consideration the age of the operators who will work in those workplaces. Finally, the current study has offered a tool that management in their ASs. The developed DSS suggests the most convenient solution, here based on investment and operational costs, between traditional Kanban, e-Kanban, and DT-based solutions for controlling the material management in ALFSs.

To sum up, first, we have supported companies in their implementation of I4.0 technologies in their ASs, showing them the decisions that they must make level by level (strategic, tactical and operational) for each technology (Table 3 andTable 4) if they want to reach the goals of the ASs. Second, to give them even more information during the design phase, we explained the impact that each I4.0 technology has on the factors relevant to the decision of which AS configuration to choose (Table 5). Third, we encouraged companies to use VR when designing the workplaces in their ASs; here, they can use our five-step methodological framework that considers the age of operators to design ergo-efficient AS workplaces using VR together with a mocap system (Figure 11). Finally, we have supported companies in the decision of the solution for controlling the material management activity in their ASs. We provided a DSS that they can use to choose the most convenient material management solution, here based on investment and operational costs, among traditional Kanban and e-Kanban and DT-based solutions by considering the demand and layout parameters of their ASs (Figure 7).

5.3. Research limitations

No research is conducted without limitations. This subsection highlights limitations that may be of significant concern.

First, because of time constraints, not all the decision areas related to the design and management of an AS have been investigated. In fact, a longer period would be necessary to study all the decision areas, but the current research had only a three-year period. Therefore, the present study focused only on three (one decision per level: strategic, tactical and operational) of the 12 decision areas related to the design and management of an AS. Moreover, for the same reason, not all the nine future research opportunities identified thanks to the SLR have been explored, here focusing only on the three connected to the three decision areas selected.

Furthermore, not all the I4.0 technologies have been considered. In fact, the I4.0 technologies were chosen from those already available in the Logistic 4.0 Lab of the Department of Mechanical and Industrial Engineering at NTNU to make the most of the resources already available. Moreover, because of time constraints, the present study has also not considered using other I4.0 technologies to address the same problems investigated during the research period.

Moreover, to conclude with the main limitations, no case studies from companies were undertaken. The quantitative results of the research study were derived from experimental research and simulations that, although giving the opportunity to arrive at solid conclusions, were created in a controlled environment without the presence of possible uncertainties that may occur in a real work environment such as an AS.

Finally, in addition to the main limitations, it is necessary to highlight that each study in the appended publications have their own specific limitations. Therefore, if interested, the reader is referred to read the conclusion section of Papers 1 through 6 to know more about the limitations of each work.

5.4. Future research

In this section, some possible future research areas that can be explored related to the design and management of AS4.0 are reported. If interested, the reader should refer to Paper 1 to get a full picture of the future research opportunities related to this topic.

Considering mobile robots and cobots, future studies should focus on the mathematical and simulation modelling of different new configurations that can be created by these two technologies. These models will allow researchers to determine the impact of these technologies on the factors that companies need to know when they must choose the AS configuration. Sensitivity analyses will also be useful to study how these factors can change the configuration and performance of the different configurations. In the end, decision support systems, such as decision trees and evolutionary algorithms, can be created to support decision makers in determining what is the best configuration to adopt based on their needs.

Mobile robots, which can be used with or without a cobot, can open up the possibility of creating new feeding policies. In fact, if cobots are used to move shelves or in conjunction with other cobots, they can be used in applications that were not possible before. Moreover, a mobile robot can only be used as a workplace or transportation device, so new scheduling models need to be created that can consider all the possible tasks that mobile robots can carry out. Because companies need to optimise the use of their mobile robots according to the applications that they have to do or tasks able to do, advanced modelling and simulations and then real case studies are needed to study these new opportunities.

AR can be used not only to support operators, but also to show what is happening in the surrounding environment, especially if there are mobile robots or cobots working within it. Therefore, further investigation and testing through experiments that replicate real-world scenarios can better highlight the strengths, weaknesses and potential benefits that the use of AR can generate both in facilitating the execution of various assembly tasks and in increasing the safety of AS4.0.

In general, a piece of technology can be chosen not only in relation to the help that it can give in terms of the physical and cognitive aspects of the execution of a task, but also based on the data that can be collected and shared with it. Therefore, new models are needed to support

92
decision makers to decide on the right technology to implement in their AS4.0, here based on what they have to do with the technology and level of automation that they want to reach in their AS4.0.

Finally, it should be noted that each study in the appended publications that give us the opportunities to answer our three research questions reports with possible future research opportunities to improve the works we have created, thus creating additional knowledge. Therefore, if interested, the reader is referred to read the conclusion sections of Paper 3 through 6.

References

- Aase, G. R., Olson, J. R. and Schniederjans, M. J., 2004. 'U-shaped assembly line layouts and their impact on labor productivity: An experimental study', European Journal of Operational Research, 156(3), pp. 698–711. doi: 10.1016/S0377-2217(03)00148-6.
- Adams, D. and Krulicky, T., 2021. Artificial Intelligence-driven Big Data Analytics, Real-Time Sensor Networks, and Product Decision-Making Information Systems in Sustainable Manufacturing Internet of Things. Economics, Management and Financial Markets, 16(3), pp.81-93.
- 3. Adenipekun, E.O., Limère, V. and Schmid, N.A., 2022. The impact of transportation optimisation on assembly line feeding. Omega, 107, p.102544.
- 4. Agostini, L. and Nosella, A., 2019. The adoption of Industry 4.0 technologies in SMEs: results of an international study. Management Decision.
- Agrawal, M., Eloot, K., Mancini, M. and Patel, A., 2020. Industry 4.0: Reimagining manufacturing operations after COVID-19. https://www.mckinsey.com/businessfunctions/operations/our-insights/industry-40-reimagining-manufacturingoperations-after-covid-19. Last accessed 07/07/2022.
- Akpinar, S., Elmi, A. and Bektaş, T., 2017. Combinatorial Benders cuts for assembly line balancing problems with setups. European Journal of Operational Research, 259(2), pp.527-537.
- Alavian, P., Eun, Y., Meerkov, S.M. and Zhang, L., 2020. Smart production systems: automating decision-making in manufacturing environment. International Journal of Production Research, 58(3), pp.828-845.
- Alexopoulos, K., Makris, S., Xanthakis, V., Sipsas, K. and Chryssolouris, G., 2016. A concept for context-aware computing in manufacturing: the white goods case. International Journal of Computer Integrated Manufacturing, 29(8), pp.839-849.
- 9. Babbie, E., 2007. The practice of social research. 11th. Belmont, CA: Thomson Wadsworth, 24(511), p.66.
- Bagher, M., Zandieh, M. and Farsijani, H., 2011. Balancing of stochastic U-type assembly lines: an imperialist competitive algorithm. The International Journal of Advanced Manufacturing Technology, 54(1-4), pp.271-285.
- 11. Bassan, Y., Roll, Y. and Rosenblatt, M.J., 1980. Internal layout design of a warehouse. AIIE Transactions, 12(4), pp.317-322.
- Battaïa, O. and Dolgui, A., 2013. A taxonomy of line balancing problems and their solution approaches. International Journal of Production Economics, 142(2), pp.259-277.
- 13. Battaïa, O., Otto, A., Sgarbossa, F. and Pesch, E., 2018. Future trends in management and operation of assembly systems: from customized assembly systems to cyber-physical systems. Omega, 78, pp.1-4.

- 14. Battini, D., Faccio, M., Persona, A. and Sgarbossa, F., 2011. New methodological framework to improve productivity and ergonomics in assembly system design. International Journal of industrial ergonomics, 41(1), pp.30-42.
- Battini, D., Faccio, M., Persona, A. and Sgarbossa, F., 2009. Design of the optimal feeding policy in an assembly system. International Journal of Production Economics, 121(1), pp.233-254.
- Battini, D., Faccio, M., Persona, A. and Sgarbossa, F., 2010. "Supermarket warehouses": stocking policies optimization in an assembly-to-order environment. The International Journal of Advanced Manufacturing Technology, 50(5), pp.775-788.
- 17. Battini, D., Faccio, M., Persona, A. and Sgarbossa, F., 2011. New methodological framework to improve productivity and ergonomics in assembly system design. International Journal of industrial ergonomics, 41(1), pp.30-42.
- Battini, D., Persona, A. and Sgarbossa, F., 2014. Innovative real-time system to integrate ergonomic evaluations into warehouse design and management. Computers & Industrial Engineering, 77, pp.1-10.
- Battini, D., Calzavara, M., Persona, A., Sgarbossa, F., Visentin, V. and Zennaro, I., 2018. Integrating mocap system and immersive reality for efficient human-centred workstation design. IFAC-PapersOnLine, 51(11), pp.188-193.
- Baumann, D., Mager, F., Wetzker, U., Thiele, L., Zimmerling, M. and Trimpe, S., 2020. Wireless Control for Smart Manufacturing: Recent Approaches and Open Challenges. Proceedings of the IEEE.
- 21. Becker, C. and Scholl, A., 2006. A survey on problems and methods in generalized assembly line balancing. European journal of operational research, 168(3), pp.694-715.
- Berinsky, A.J., Huber, G.A. and Lenz, G.S., 2012. Evaluating online labor markets for experimental research: Amazon. com's Mechanical Turk. Political analysis, 20(3), pp.351-368.
- 23. Bertrand, J. W. M., & Fransoo, J. C. (2016). Modelling and simulation. Research Methods for Operations Management, 306-346.
- Bhattacherjee, A., 2022. Experimental Research [Online]. Available: https://courses.lumenlearning.com/suny-hccc-research-methods/chapter/chapter-10-experimental-research/. Last accessed 20/05/2022.
- 25. Bi, Z.M., Wang, L. and Lang, S.Y., 2007. Current status of reconfigurable assembly systems. International Journal of Manufacturing Research, 2(3), pp.303-328.
- Bläsing, D., Hinrichsen, S. and Bornewasser, M., 2020, April. Reduction of cognitive load in complex assembly systems. In International Conference on Human Interaction and Emerging Technologies (pp. 495-500). Springer, Cham.
- 27. Bloss, R., 2016. Collaborative robots are rapidly providing major improvements in productivity, safety, programing ease, portability and cost while addressing many new applications. Industrial Robot: An International Journal.

- Borrego, M., Douglas, E.P. and Amelink, C.T., 2009. Quantitative, qualitative, and mixed research methods in engineering education. Journal of Engineering education, 98(1), pp.53-66.
- Bortolini, M., Faccio, M., Galizia, F.G., Gamberi, M. and Pilati, F., 2021. Adaptive automation assembly systems in the industry 4.0 era: a reference framework and full– scale prototype. Applied Sciences, 11(3), p.1256.
- Bortolini, M., Faccio, M., Gamberi, M. and Pilati, F., 2020. Motion Analysis System (MAS) for production and ergonomics assessment in the manufacturing processes. Computers & Industrial Engineering, 139, p.105485.
- Bortolini, M., Ferrari, E., Gamberi, M., Pilati, F. and Faccio, M., 2017. Assembly system design in the Industry 4.0 era: a general framework. IFAC-PapersOnLine, 50(1), pp.5700-5705.
- 32. Bossard.<u>https://www.bossard.com/global-en/smart-factory-</u> logistics/systems/smartbin-intelligent-logistics-system/. Accessed 15 Aug 2022.
- 33. Bouma, H., 2013. Accommodating older people at work. Gerontechnology, 11(4), pp.489-492.
- Brettel, M., Friederichsen, N., Keller, M., Rosenberg, M., 2014. How virtualization, decentralization and network building change the manufacturing Landscape . Int. J. Mech. Ind. Sci. Eng. 8, 37–44.
- Brough, J.E., Schwartz, M., Gupta, S.K., Anand, D.K., Kavetsky, R. and Pettersen, R., 2007. Towards the development of a virtual environment-based training system for mechanical assembly operations. Virtual reality, 11(4), pp.189-206.
- 36. Buchanan, D., & Bryman, A. (2009). The Sage handbook of organizational research methods: Sage Publications Ltd.
- Buckhorst, A.F., Grahn, L. and Schmitt, R.H., 2022. Decentralized Holonic Control System Model for Line-less Mobile Assembly Systems. Robotics and Computer-Integrated Manufacturing, 75, p.102301.
- Buer, S.V., Fragapane, G.I. and Strandhagen, J.O., 2018. The data-driven process improvement cycle: Using digitalization for continuous improvement. IFAC-PapersOnLine, 51(11), pp.1035-1040.
- 39. Bukchin, J. and Masin, M., 2004. Multi-objective design of team oriented assembly systems. European Journal of Operational Research, 156(2), pp.326-352.
- 40. Bures, M. and Simon, M., 2015. Adaptation of production systems according to the conditions of ageing population. MM Science Journal, 2015, pp.604-609.
- 41. Boysen, N., Fliedner, M. and Scholl, A., 2008. Assembly line balancing: Which model to use when?. International journal of production economics, 111(2), pp.509-528.
- Boysen, N., Kiel, M. and Scholl, A., 2011. Sequencing mixed-model assembly lines to minimise the number of work overload situations. International Journal of Production Research, 49(16), pp.4735-4760.
- 43. Calzavara, M., Faccio, M., Persona, A. and Zennaro, I., 2021. Walking worker vs fixed worker assembly considering the impact of components exposure on assembly time

and energy expenditure. The International Journal of Advanced Manufacturing Technology, 112(9), pp.2971-2988.

- 44. Caputo, F., Greco, A., Egidio, D.A., Notaro, I. and Spada, S., 2017, July. A preventive ergonomic approach based on virtual and immersive reality. In International Conference on Applied Human Factors and Ergonomics (pp. 3-15). Springer, Cham.
- 45. Caputo, A.C. and Pelagagge, P.M., 2011. A methodology for selecting assembly systems feeding policy. Industrial Management & Data Systems.
- 46. Caputo, A.C., Pelagagge, P.M. and Salini, P., 2018. Selection of assembly lines feeding policies based on parts features and scenario conditions. International Journal of Production Research, 56(3), pp.1208-1232.
- 47. Cattaneo, L., Fumagalli, L., Macchi, M. and Negri, E., 2018. Clarifying data analytics concepts for industrial engineering. IFAC-PapersOnLine, 51(11), pp.820-825.
- 48. Chen, B., Liu, Y., Zhang, C. and Wang, Z., 2020a. Time series data for equipment reliability analysis with deep learning. IEEE Access, 8, pp.105484-105493.
- 49. Chen, C., Tian, Z., Li, D., Pang, L., Wang, T. and Hong, J., 2020b. Projection-based augmented reality system for assembly guidance and monitoring. Assembly Automation.
- Chen, G.S. and Chen, J.P., 2014. Applying virtual reality to remote control of mobile robot. In Proceedings of the 2nd international conference on intelligent technologies and engineering systems (ICITES2013) (pp. 383-390). Springer, Cham.
- Chien, C.H., Wang, W.Y. and Hsu, C.C., 2017. Multi-objective evolutionary approach to prevent premature convergence in Monte Carlo localization. Applied Soft Computing, 50, pp.260-279.
- 52. Choobineh, F.F., Asef-Vaziri, A. and Huang, X., 2012. Fleet sizing of automated guided vehicles: a linear programming approach based on closed queuing networks. International Journal of Production Research, 50(12), pp.3222-3235.
- 53. Chung, C.A., 2003. Simulation modeling handbook: a practical approach. CRC press.
- 54. Cohen, Y., Faccio, M., Galizia, F.G., Mora, C. and Pilati, F., 2017. Assembly system configuration through Industry 4.0 principles: the expected change in the actual paradigms. IFAC-PapersOnLine, 50(1), pp.14958-14963.
- 55. Cohen, Y., Faccio, M., Pilati, F. and Yao, X., 2019a. Design and management of digital manufacturing and assembly systems in the Industry 4.0 era. The International Journal of Advanced Manufacturing Technology, 105(9), pp.3565-3577.
- Cohen, Y., Naseraldin, H., Chaudhuri, A. and Pilati, F., 2019b. Assembly systems in Industry 4.0 era: a road map to understand Assembly 4.0. The International Journal of Advanced Manufacturing Technology, 105(9), pp.4037-4054.
- 57. Coleman, C., Damodaran, S., Chandramouli, M. and Deuel, E., 2017. Making maintenance smarter. Predictive maintenance and the digital supply network. https://www2.deloitte.com/us/en/insights/focus/industry-4-0/using-predictiveechnologies-for-asset-maintenance.html. Last accessed 07/07/2022.

- Colledani, M., Gyulai, D., Monostori, L., Urgo, M., Unglert, J. and Van Houten, F., 2016. Design and management of reconfigurable assembly lines in the automotive industry. CIRP Annals, 65(1), pp.441-446.
- 59. Cook, T. D., Campbell, D. T., & Day, A. (1979). Quasi-experimentation: Design & analysis issues for field settings (Vol. 351): Houghton Mifflin Boston.
- 60. Cormier, G. and Gunn, E.A., 1992. A review of warehouse models. European journal of operational research, 58(1), pp.3-13.
- 61. Croom, S. (2009). Introduction to research methodology in operations management. Researching operations management, 43-83.
- 62. Dalenogare, L.S., Benitez, G.B., Ayala, N.F. and Frank, A.G., 2018. The expected contribution of Industry 4.0 technologies for industrial performance. International Journal of production economics, 204, pp.383-394.
- 63. Dang, Q.V., Nielsen, I., Steger-Jensen, K. and Madsen, O., 2014. Scheduling a single mobile robot for part-feeding tasks of production lines. Journal of Intelligent Manufacturing, 25(6), pp.1271-1287.
- Danielsson, O., Holm, M. and Syberfeldt, A., 2020. Augmented reality smart glasses in industrial assembly: Current status and future challenges. Journal of Industrial Information Integration, 20, p.100175.
- 65. DeCroix, G.A. and Zipkin, P.H., 2005. Inventory management for an assembly system with product or component returns. Management Science, 51(8), pp. 1250-1265.
- 66. Dilberoglu, U. M., Gharehpapagh, B., Yaman, U., & Dolen, M. (2017). The role of additive manufacturing in the era of industry 4.0. Procedia Manufacturing, 11, 545-554.
- 67. Di Pasquale, V., Miranda, S. and Neumann, W.P., 2020. Ageing and human-system errors in manufacturing: a scoping review. International Journal of Production Research, 58(15), pp.4716-4740.
- 68. de Sousa Jabbour, A.B.L., Jabbour, C.J.C., Foropon, C. and Godinho Filho, M., 2018. When titans meet–Can industry 4.0 revolutionise the environmentally-sustainable manufacturing wave? The role of critical success factors. Technological Forecasting and Social Change, 132, pp.18-25.
- 69. Dolgui, A., Kovalev, S., Kovalyov, M., Malyutin, S., Soukhal, A., 2018. Optimal workforce assignment to operations of a paced assembly line. European Journal of Operational Research, 264 (1), pp. 200-211.
- 70. Dolgui, A., Proth, J.-M., 2010. Supply chain engineering: useful methods and techniques, London: Springer.
- Dolgui, A., Sgarbossa, F. and Simonetto, M., 2022. Design and management of assembly systems 4.0: systematic literature review and research agenda. International Journal of Production Research, 60(1), pp.184-210.
- Elhoseny, M., Tharwat, A. and Hassanien, A.E., 2018. Bezier curve based path planning in a dynamic field using modified genetic algorithm. Journal of Computational Science, 25, pp. 339-350.

- Etzi, R., Huang, S., Scurati, G.W., Lyu, S., Ferrise, F., Gallace, A., Gaggioli, A., Chirico, A., Carulli, M. and Bordegoni, M., 2019, August. Using virtual reality to test human-robot interaction during a collaborative task. In International Design Engineering Technical Conferences and Computers and Information in Engineering Conference (Vol. 59179, p. V001T02A080). American Society of Mechanical Engineers.
- 74. Faccio, M., Ferrari, E., Gamberi, M. and Pilati, F., 2019. Human Factor Analyser for work measurement of manual manufacturing and assembly processes. The International Journal of Advanced Manufacturing Technology, 103(1-4), pp. 861-877.
- Faccio, M., Minto, R., Rosati, G. and Bottin, M., 2020. The influence of the product characteristics on human-robot collaboration: a model for the performance of collaborative robotic assembly. The International Journal of Advanced Manufacturing Technology, 106(5), pp.2317-2331.
- Fager, P., Calzavara, M. and Sgarbossa, F., 2020. Modelling time efficiency of cobotsupported kit preparation. The International Journal of Advanced Manufacturing Technology, 106(5), pp.2227-2241.
- Fager, P., Sgarbossa, F. and Calzavara, M., 2021. Cost modelling of onboard cobotsupported item sorting in a picking system. International Journal of Production Research, 59(11), pp.3269-3284.
- Fast-Berglund, Å., Palmkvist, F., Nyqvist, P., Ekered, S. and Åkerman, M., 2016.
 Evaluating cobots for final assembly. Procedia CIRP, 44, pp.175-180.
- 79. Fragapane, G., de Koster, R., Sgarbossa, F. and Strandhagen, J.O., 2021. Planning and control of autonomous mobile robots for intralogistics: Literature review and research agenda. European Journal of Operational Research.
- 80. Fragapane, G., Ivanov, D., Peron, M., Sgarbossa, F. and Strandhagen, J.O., 2020. Increasing flexibility and productivity in industry 4.0 production networks with autonomous mobile robots and smart intralogistics. Annals of operations research, pp.1-19.
- Gabor, T., Belzner, L., Kiermeier, M., Beck, M.T. and Neitz, A., 2016, July. A simulationbased architecture for smart cyber-physical systems. In 2016 IEEE international conference on autonomic computing (ICAC) (pp. 374-379). IEEE.
- Gao, L., Xu, S.H. and Ball, M.O., 2012. Managing an available-to-promise assembly system with dynamic short-term pseudo-order forecast. Management Science, 58(4), pp.770-790.
- 83. Ge, Z., Song, Z., Ding, S.X. and Huang, B., 2017. Data mining and analytics in the process industry: The role of machine learning. leee Access, 5, pp.20590-20616.
- Gebser, M., Obermeier, P., Schaub, T., Ratsch-Heitmann, M. and Runge, M., 2018. Routing driverless transport vehicles in car assembly with answer set programming. Theory and Practice of Logic Programming, 18(3-4), pp.520-534.
- Georgakopoulos, D., Jayaraman, P.P., Fazia, M., Villari, M. and Ranjan, R., 2016. Internet of Things and edge cloud computing roadmap for manufacturing. IEEE Cloud Computing, 3(4), pp.66-73.

- Gharehgozli, A. and Zaerpour, N., 2020. Robot scheduling for pod retrieval in a robotic mobile fulfillment system. Transportation Research Part E: Logistics and Transportation Review, 142, p.102087.
- 87. Ghobakhloo, M., 2018. The future of manufacturing industry: a strategic roadmap toward Industry 4.0. Journal of manufacturing technology management.
- Gil-Vilda, F., Sune, A., Yagüe-Fabra, J.A., Crespo, C. and Serrano, H., 2017. Integration of a collaborative robot in a U-shaped production line: a real case study. Procedia Manufacturing, 13, pp.109-115.
- 89. Globalluxsoft, 2022. Impact of Industry 4.0 in global manufacturing. <u>https://globalluxsoft.com/impact-industry-4-in-global-manufacturing</u>. bLast accessed 07/07/2022.
- Gordon, V., Proth, J.M. and Chu, C., 2002. A survey of the state-of-the-art of common due date assignment and scheduling research. European Journal of Operational Research, 139(1), pp.1-25.
- Gorecky, D., Khamis, M. and Mura, K., 2017. Introduction and establishment of virtual training in the factory of the future. International Journal of Computer Integrated Manufacturing, 30(1), pp.182-190.
- 92. Götz, M. and Jankowska, B., 2020. Adoption of Industry 4.0 technologies and company competitiveness: case studies from a post-transition economy. Форсайт, 14(4 (eng)), pp.61-78.
- Grajewski, D., Górski, F., Zawadzki, P. and Hamrol, A., 2013. Application of virtual reality techniques in design of ergonomic manufacturing workplaces. Procedia Computer Science, 25, pp.289-301.
- 94. Gruenefeld, U., Prädel, L., Illing, J., Stratmann, T., Drolshagen, S. and Pfingsthorn, M., 2020, September. Mind the ARm: Realtime visualization of robot motion intent in head-mounted augmented reality. In Proceedings of the Conference on Mensch und Computer (pp. 259-266).
- 95. Guo, D., Zhong, R.Y., Lin, P., Lyu, Z., Rong, Y. and Huang, G.Q., 2020. Digital twinenabled Graduation Intelligent Manufacturing System for fixed-position assembly islands. Robotics and Computer-Integrated Manufacturing, 63, p.101917.
- 96. Halldorsson, A., & Aastrup, J. (2003). Quality criteria for qualitative inquiries in logistics. European Journal of Operational Research, 144(2), 321-332.
- Harrison, J.R., Lin, Z., Carroll, G.R. and Carley, K.M., 2007. Simulation modeling in organizational and management research. Academy of management review, 32(4), pp.1229-1245.
- Hashemi-Petroodi, S.E., Dolgui, A., Kovalev, S., Kovalyov, M., Thevenin, S., 2020a. Workforce reconfiguration strategies in manufacturing systems: a state of the art. International Journal of Production Research, doi: 10.1080 00207543.2020.1823028.
- 99. Hashemi-Petroodi, S.E., Thevenin, S., Kovalev, S. and Dolgui, A., 2020b. Operations management issues in design and control of hybrid human-robot collaborative manufacturing systems: a survey. Annual Reviews in Control, 49, pp. 264–276.

- 100. Havard, V., Jeanne, B., Lacomblez, M. and Baudry, D., 2019. Digital twin and virtual reality: a co-simulation environment for design and assessment of industrial workstations. Production & Manufacturing Research, 7(1), pp.472-489.
- 101. Hořejší, P., 2015. Augmented reality system for virtual training of parts assembly. Procedia Engineering, 100, pp.699-706.
- Hu, S.J., Ko, J., Weyand, L., ElMaraghy, H.A., Lien, T.K., Koren, Y., Bley, H., Chryssolouris, G., Nasr, N. and Shpitalni, M., 2011. Assembly system design and operations for product variety. CIRP annals, 60(2), pp.715-733.
- 103. Hu, S.J., Zhu, X., Wang, H. and Koren, Y., 2008. Product variety and manufacturing complexity in assembly systems and supply chains. CIRP annals, 57(1), pp.45-48.
- 104. Huang, C.-C. and Kusiak, A., 1996. Overview of Kanban systems. International Journal of Computer Integrated Manufacturing, 9(3), pp. 169–189. doi:10.1080/095119296131643.
- 105. Houti, M., El Abbadi, L. and Abouabdellah, A., 2017, April. E-Kanban the new generation of traditional Kanban system, and the impact of its implementation in the enterprise. In Proceedings of the International Conference on Industrial Engineering and Operations Management Rabat (pp. 11-13).
- 106. Ilika, D., 2017. Here's how Audi plans to scrap the assembly line. AutoGuide.com, 17th July. Downloaded from http://www.autoguide.com/auto-news/2017/07/here-is-how- audi-plans-to-scrap-the-assembly-line.html.
- 107. Jeschke, S., Brecher, C., Meisen, T., Özdemir, D., Eschert, T., 2017. Industrial Internet of Things and Cyber Manufacturing Systems. Springer. https://doi.org/10.1007/978-3-319-42559-7_1.
- 108. Johansson, P.E., Mattsson, S., Moestam, L. and Fast-Berglund, Å., 2016. Multi-variant truck production-product variety and its impact on production quality in manual assembly. Procedia CIRP, 54, pp.245-250.
- 109. Johnson, R., 1981. Assembly line balancing algorithms. International Journal of Production Research 19, 277–287.
- 110. Junior, M.L. and Godinho Filho, M., 2010. Variations of the kanban system: Literature review and classification. International Journal of Production Economics, 125(1), pp.13-21.
- 111. Karadayi-Usta, S., 2019. An interpretive structural analysis for industry 4.0 adoption challenges. IEEE Transactions on Engineering Management, 67(3), pp.973-978.
- 112. Karlsson, C. (2016). Research in operations management. Research Methods for Operations Management, 23-61.
- 113. Kim, G.Y., Flores-García, E., Wiktorsson, M. and Noh, S.D., 2021, September. Exploring Economic, Environmental, and Social Sustainability Impact of Digital Twin-Based Services for Smart Production Logistics. In IFIP International Conference on Advances in Production Management Systems (pp. 20-27). Springer, Cham.

- 114. Kolbeinsson, A., Lagerstedt, E. and Lindblom, J., 2019. Foundation for a classification of collaboration levels for human-robot cooperation in manufacturing. Production & Manufacturing Research, 7(1), pp.448-471.
- 115. Kouri, I.A., Salmimaa, T.J. and Vilpola, I.H., 2008. The principles and planning process of an electronic kanban system. In Novel algorithms and techniques in telecommunications, automation and industrial electronics (pp. 99-104). Springer, Dordrecht.
- 116. Kucukkoc, I. and Zhang, D.Z., 2017. Balancing of mixed-model parallel U-shaped assembly lines considering model sequences. International Journal of Production Research, 55(20), pp.5958-5975.
- 117. Kuzgunkaya, O. and ElMaraghy, H.A., 2006. Assessing the structural complexity of manufacturing systems configurations. International Journal of Flexible Manufacturing Systems, 18(2), pp.145-171.
- 118. Lasi, H., Fettke, P., Kemper, H.G., Feld, T. and Hoffmann, M., 2014. Industry 4.0. Business & information systems engineering, 6(4), pp.239-242.
- 119. Lee, J., Bagheri, B. and Kao, H.A., 2015. A cyber-physical systems architecture for industry 4.0-based manufacturing systems. Manufacturing letters, 3, pp.18-23.
- 120. Lee, Y., Do, W., Yoon, H., Heo, J., Lee, W. and Lee, D., 2021. Visual-inertial hand motion tracking with robustness against occlusion, interference, and contact. Science Robotics, 6(58), p.eabe1315.
- 121. Levitin, G., Rubinovitz, J. and Shnits, B., 2006. A genetic algorithm for robotic assembly line balancing. European Journal of Operational Research, 168(3), pp.811-825.
- 122. Lezzi, M., Lazoi, M. and Corallo, A., 2018. Cybersecurity for Industry 4.0 in the current literature: A reference framework. Computers in Industry, 103, pp.97-110.
- 123. Li, J. and Gao, J., 2014. Balancing manual mixed-model assembly lines using overtime work in a demand variation environment. International Journal of Production Research, 52(12), pp.3552-3567.
- 124. Lin, J.H., McGorry, R.W., Chang, C.C. and Dempsey, P.G., 2007. Effects of user experience, working posture and joint hardness on powered nutrunner torque reactions. Ergonomics, 50(6), pp.859-876.
- 125. Lindenmeyer, C., 2001. How to Design and Conduct a Computer-Integrated Time Study with Active Element Performance Rating (CITS APR). Copyright, C-FOUR July 8, 2001. Page 1 of 16.
- 126. Liu, M., Ma, J., Lin, L., Ge, M., Wang, Q. and Liu, C., 2017. Intelligent assembly system for mechanical products and key technology based on internet of things. Journal of Intelligent Manufacturing, 28(2), pp.271-299.
- 127. Lorenz, M., Rüßmann, M., Strack, R., Lueth, K.L. and Bolle, M., 2015. Man and machine in industry 4.0: How will technology transform the industrial workforce through 2025. The Boston Consulting Group, 2.
- 128. Lu, Y., 2017. Industry 4.0: A survey on technologies, applications and open research issues. Journal of Industrial Information Integration, 6, pp.1-10.

- 129. MacDuffie, J.P., Sethuraman, K. and Fisher, M.L., 1996. Product variety and manufacturing performance: evidence from the international automotive assembly plant study. Management Science, 42(3), pp.350-369.
- 130. Mahmood, A., Arshad Ali, A., Nazam, M. and Nazim, M., 2021. Developing an interplay among the psychological barriers for the adoption of industry 4.0 phenomenon. PloS one, 16(8), p.e0255115.
- 131. Malik, A.A. and Brem, A., Digital twins for collaborative robots: A case study in humanrobot interaction. Robotics and Computer-Integrated Manufacturing, 68, p.102092.
- 132. Maria, A., 1997, December. Introduction to modeling and simulation. In Proceedings of the 29th conference on Winter simulation (pp. 7-13).
- 133. Mastor, A.A., 1970. An experimental investigation and comparative evaluation of production line balancing techniques. Management Science, 16(11), pp.728-746.
- 134. Mateus, J.C., Claeys, D., Limère, V., Cottyn, J. and Aghezzaf, E.H., 2019. A structured methodology for the design of a human-robot collaborative assembly workplace. The International Journal of Advanced Manufacturing Technology, 102(5-8), pp.2663-2681.
- 135. Matt, D.T., 2013. Design of a scalable assembly system for product variety: a case study. Assembly Automation.
- 136. Mei, Y., Lu, Y.H., Hu, Y.C. and Lee, C.G., 2005, July. A case study of mobile robot's energy consumption and conservation techniques. In ICAR'05. Proceedings., 12th International Conference on Advanced Robotics, 2005. (pp. 492-497). IEEE.
- 137. Mehnen, J., He, H., Tedeschi, S. and Tapoglou, N., 2017. Practical security aspects of the internet of things. In Cybersecurity for Industry 4.0 (pp. 225-242). Springer, Cham.
- 138. Mell P., Grance T. Perspectives on cloud computing and standards. National Institute of Standards and Technology (NIST). Information Technology Laboratory; 2009.
- 139. Mentzer, J. T., & Kahn, K. B. (1995). A framework of logistics research. Journal of business logistics, 16(1), 231.
- 140. Michalos, G., Karvouniari, A., Dimitropoulos, N., Togias, T. and Makris, S., 2018. Workplace analysis and design using virtual reality techniques. CIRP Annals, 67(1), pp.141-144.
- 141. Miller, J., Hoover, M. and Winer, E., 2020. Mitigation of the Microsoft HoloLens' hardware limitations for a controlled product assembly process. The International Journal of Advanced Manufacturing Technology, 109(5), pp.1741-1754.
- 142. Monden, Y., 1993, Toyota Production System (Norcross, GA: Industrial Engineering and Management Press, Institute of Industrial Engineers).
- 143. Morlet, A., Blériot, J., Opsomer, R., Linder, M., Henggeler, A., Bluhm, A. and Carrera, A., 2016. Intelligent assets: Unlocking the circular economy potential. Ellen MacArthur Foundation, pp.1-25.
- 144. Mossa, G., Boenzi, F., Digiesi, S., Mummolo, G. and Romano, V.A., 2016. Productivity and ergonomic risk in human based production systems: A job-rotation scheduling model. International Journal of Production Economics, 171, pp.471-477.

- 145. Moysis, L., Petavratzis, E., Volos, C., Nistazakis, H. and Stouboulos, I., 2020. A chaotic path planning generator based on logistic map and modulo tactics. Robotics and Autonomous Systems, 124, p.103377.
- 146. Mukund Nilakantan, J. and Ponnambalam, S.G., 2016. Robotic U-shaped assembly line balancing using particle swarm optimization. Engineering Optimization, 48(2), pp.231-252.
- 147. Negri, E., Berardi, S., Fumagalli, L. and Macchi, M., 2020. MES-integrated digital twin frameworks. Journal of Manufacturing Systems, 56, pp.58-71.
- 148. Nielsen, I., Dang, Q.V., Bocewicz, G. and Banaszak, Z., 2017. A methodology for implementation of mobile robot in adaptive manufacturing environments. Journal of Intelligent Manufacturing, 28(5), pp.1171-1188.
- 149. Nikolakis, N., Alexopoulos, K., Xanthakis, E. and Chryssolouris, G., 2019a. The digital twin implementation for linking the virtual representation of human-based production tasks to their physical counterpart in the factory-floor. International Journal of Computer Integrated Manufacturing, 32(1), pp.1-12.
- 150. Nikolakis, N., Alexopoulos, K., Xanthakis, E. and Chryssolouris, G., 2019a. The digital twin implementation for linking the virtual representation of human-based production tasks to their physical counterpart in the factory-floor. International Journal of Computer Integrated Manufacturing, 32(1), pp.1-12.
- 151. Nokia, 2022. Industry 4.0:"think value, not tech". An inside look at the drivers for Industry 4.0 with its 'father', Henrik von Scheel. https://www.nokia.com/networks/insights/industry-4-0/three-key-drivers-forsuccess/?did=D00000001887&gclid=Cj0KCQjw5ZSWBhCVARIsALERCvyTsgTUDCdl3U 0142ID1QRY0HgqHgybh3iXy4JUFMs13k4WdmJ1-oAaAmeuEALw_wcB. Last accessed 07/07/2022
- 152. Nuzzi, C., Pasinetti, S., Pagani, R., Ghidini, S., Beschi, M., Coffetti, G. and Sansoni, G., MEGURU: a gesture-based robot program builder for Meta-Collaborative workstations. Robotics and Computer-Integrated Manufacturing, 68, p.102085.
- 153.Ong, S.K. and Nee, A.Y.C., 2013. Virtual and augmented reality applications in manufacturing. Springer Science & Business Media.
- 154.Ong, S.K., Yuan, M.L. and Nee, A.Y., 2008. Augmented reality applications in manufacturing: a survey. International journal of production research, 46(10), pp.2707-2742.
- 155. Otto, A. and Li, X., 2020. Product sequencing in multiple-piece-flow assembly lines. Omega, 91, p.102055.
- 156. Ouyang, L., Yuan, Y. and Wang, F.Y., 2019, November. A blockchain-based framework for collaborative production in distributed and social manufacturing. In 2019 IEEE International Conference on Service Operations and Logistics, and Informatics (SOLI) (pp. 76-81). IEEE.

- 157. Oyekan, J., A. Fischer, W. Hutabarat, C. Turner, and A. Tiwari. 2019. "Utilising Low Cost RGB-D Cameras to Track the Real Time Progress of a Manual Assembly Sequence." Assembly Automation 40 (6): 925–939.
- 158. Oyekan, J., Prabhu, V., Tiwari, A., Baskaran, V., Burgess, M. and Mcnally, R., 2017. Remote real-time collaboration through synchronous exchange of digitised human– workpiece interactions. Future Generation Computer Systems, 67, pp.83-93.
- 159. Paelke, V., 2014, September. Augmented reality in the smart factory: Supporting workers in an industry 4.0. environment. In Proceedings of the 2014 IEEE emerging technology and factory automation (ETFA) (pp. 1-4). IEEE.
- 160. Palominos, P., Quezada, L., Donoso, J. and Gonzalez, M., 2019. A Model of Economic Evaluation for the Acquisition of Flexible Manufacturing Technologies. Procedia Manufacturing, 39, pp.565-573.
- 161. Panda, M.R., Das, P. and Pradhan, S., 2017. Hybridization of IWO and IPSO for mobile robots navigation in a dynamic environment. Journal of King Saud University-Computer and Information Sciences.
- 162. Papakostas, N., Pintzos, G., Giannoulis, C. and Chryssolouris, G., 2016. An agent-based collaborative platform for the design of assembly lines. International Journal of Computer Integrated Manufacturing, 29(4), pp.374-385.
- 163. Pegden, C.D., Sadowski, R.P. and Shannon, R.E., 1995. Introduction to simulation using SIMAN. McGraw-Hill, Inc.
- 164. Peron, M., Sgarbossa, F. and Strandhagen, J.O., 2020. Decision support model for implementing assistive technologies in assembly activities: a case study. International Journal of Production Research, pp.1-27.
- 165. Peruzzini, M., Carassai, S. and Pellicciari, M., 2017. The benefits of human-centred design in industrial practices: re-design of workstations in pipe industry. Procedia Manufacturing, 11, pp.1247-1254.
- 166. Peruzzini, M., Grandi, F. and Pellicciari, M., 2020. Exploring the potential of Operator4.0 interface and monitoring. Computers & Industrial Engineering, 139, p.105600.
- 167. Piller, F.T., 2005. Mass customization: reflections on the state of the concept. International journal of flexible manufacturing systems, 16(4), pp.313-334.
- 168. Plantard, P., Shum, H.P., Le Pierres, A.S. and Multon, F., 2017. Validation of an ergonomic assessment method using Kinect data in real workplace conditions. Applied ergonomics, 65, pp.562-569.
- 169. Pollard, D., Chuo, S. and Lee, B., 2008. Strategies for mass customization. Journal of Business & Economics Research (JBER), 6(7).
- 170. Pollfish, 2022. What is Experimental Research & How is it Significant for Your Business [Online]. Available: https://resources.pollfish.com/survey-guides/what-isexperimental-research-how-is-it-significant-for-your-business/. Last accessed 20/05/2022.
- 171. Porshe, 2019. A new chapter in the company's history. Newsroom.https://newsroom.porsche.com/en/2019/company/porsche-taycanfactory-zuffenhausen-opening-18644.html
- 172. Queirós, A., Faria, D. and Almeida, F., 2017. Strengths and limitations of qualitative and quantitative research methods. European journal of education studies.

- 173. Rabbani, M., Moghaddam, M. and Manavizadeh, N., 2012. Balancing of mixed-model two-sided assembly lines with multiple U-shaped layout. The International Journal of Advanced Manufacturing Technology, 59(9-12), pp.1191-1210.
- 174. Rainer, D. and Alexander, H., 2014. Industrie 4.0: hit or hype. Industrial Electronics Magazine, 8(2), pp.56-58.
- 175. Rekiek, B., Dolgui, A., Delchambre, A. and Bratcu, A., 2002. State of art of optimization methods for assembly line design. Annual Reviews in control, 26(2), pp. 163-174.
- 176. Reswick, J.B., 1994. What constitutes valid research? Qualitative vs. quantitative research. Technology and Disability, 3(4), pp.255-257.
- 177. Ritchie, J.M., Dewar, R.G. and Simmons, J.E., 1999. The generation and practical use of plans for manual assembly using immersive virtual reality. Proceedings of the Institution of Mechanical Engineers, Part B: Journal of Engineering Manufacture, 213(5), pp.461-474.
- 178. Roodbergen, K.J. and Vis, I.F., 2006. A model for warehouse layout. IIE transactions, 38(10), pp.799-811.
- 179. Ruppert, T. and Abonyi, J., 2020. Integration of real-time locating systems into digital twins. Journal of Industrial Information Integration, 20, p.100174.
- 180.Şahin, M. and Kellegöz, T., 2017. Increasing production rate in U-type assembly lines with sequence-dependent set-up times. Engineering Optimization, 49(8), pp.1401-1419.
- 181. Sali, M. and Sahin, E., 2016. Line feeding optimization for Just in Time assembly lines: An application to the automotive industry. International Journal of Production Economics, 174, pp.54-67.
- 182.San Martín, A. and Kildal, J., 2019, May. Audio-visual AR to improve awareness of hazard zones around robots. In Extended Abstracts of the 2019 CHI Conference on Human Factors in Computing Systems (pp. 1-6).
- 183. Sapry, H.R.M., Sabli, S.F.H.M. and Ahmad, A.R., 2020. Exploring E-Kanban Application in the Inventory Management Process. Journal of Critical Reviews, 7(8), pp.6-10.
- 184. Schmitt, J., Bönig, J., Borggräfe, T., Beitinger, G. and Deuse, J., 2020. Predictive modelbased quality inspection using Machine Learning and Edge Cloud Computing. Advanced Engineering Informatics, 45, p.101101.
- Seuring, S. and Gold, S., 2012. Conducting content-analysis based literature reviews in supply chain management. Supply Chain Management: An International Journal, Vol. 17, No. 5, pp. 544-555, DOI: 10.1108/13598541211258609.
- 186.Seth, A., Vance, J.M. and Oliver, J.H., 2011. Virtual reality for assembly methods prototyping: a review. Virtual reality, 15(1), pp.5-20.
- 187.Sew Eurodrive, 2022. Showcase factory: Our smart factory in Graben-Neudorf. https://www.sew-eurodrive.de/automation/factory-automation/industry-40/showcase-factory/showcase-factory.html. Last accessed 15/08/2022.
- 188. Sgarbossa, F., Grosse, E.H., Neumann, W.P., Battini, D. and Glock, C.H., 2020. Human factors in production and logistics systems of the future. Annual Reviews in Control, 49, pp.295-305.

- 189. Shephard, R.J., 2000. Aging and productivity: some physiological issues. International Journal of Industrial Ergonomics, 25(5), pp.535-545.
- 190. Shtub, A. and Dar-El, E.M., 1989. A methodology for the selection of assembly systems. THE INTERNATIONAL JOURNAL OF PRODUCTION RESEARCH, 27(1), pp.175-186.
- 191. Singh, M., Fuenmayor, E., Hinchy, E.P., Qiao, Y., Murray, N. and Devine, D., 2021. Digital twin: origin to future. Applied System Innovation, 4(2), p.36.
- 192. Singh, N., Shek, K.H. and Meloche, D., 1990. The development of a kanban system: a case study. International Journal of Operations & Production Management, Vol. 10 No. 7, pp. 28-36. https://doi.org/10.1108/01443579010140498.
- 193. Sirimanne, S., 2022. What is 'Industry 4.0' and what will it mean for developing countries?. https://www.weforum.org/agenda/2022/04/what-is-industry-4-0-and-could-developing-countries-get-left-behind/. Last accessed 07/12/2022.
- 194. Song, B., Wang, Z. and Sheng, L., 2016. A new genetic algorithm approach to smooth path planning for mobile robots. Assembly Automation.
- 195. Souifi, A., Boulanger, Z.C., Zolghadri, M., Barkallah, M. and Haddar, M., 2022. Uncertainty of key performance indicators for Industry 4.0: A methodology based on the theory of belief functions. Computers in Industry, 140, p.103666.
- 196. Stadnicka, D. and Antonelli, D., 2019. Human-robot collaborative work cell implementation through lean thinking. International Journal of Computer Integrated Manufacturing, 32(6), pp.580-595.
- 197. Stock, T. and Seliger, G., 2016. Opportunities of sustainable manufacturing in industry 4.0. procedia CIRP, 40, pp.536-541.
- 198. Storti, E., Cattaneo, L., Polenghi, A. and Fumagalli, L., 2018. Customized knowledge discovery in databases methodology for the control of assembly systems. Machines, 6(4), p.45.
- 199.Swedberg, R., 2020. Exploratory research. The production of knowledge: Enhancing progress in social science, pp.17-41.
- 200. Tao, W., Leu, M.C. and Yin, Z., 2020. Multi-modal recognition of worker activity for human-centered intelligent manufacturing. Engineering Applications of Artificial Intelligence, 95, p.103868.
- 201. Tarallo, A., Mozzillo, R., Di Gironimo, G. and De Amicis, R., 2018. A cyber-physical system for production monitoring of manual manufacturing processes. International Journal on Interactive Design and Manufacturing (IJIDeM), 12(4), pp.1235-1241.
- 202. Tegan, G., 2022. Exploratory Research, definition guide and examples. https://www.scribbr.com/methodology/exploratory-research.
- 203. Thames, L. and Schaefer, D., 2016. Software-defined cloud manufacturing for industry 4.0. Procedia cirp, 52, pp.12-17.
- 204. Tranfield, D., Denyer, D., & Smart, P. (2003). Towards a methodology for developing evidence-informed management knowledge by means of systematic review. British journal of management, 14(3), 207-222.
- 205. Ucar, I., Smeets, B., Azcorra, A., 2019. "simmer: Discrete-Event Simulation for R." Journal of Statistical Software, 90(2), 1–30. doi: 10.18637/jss.v090.i02.

- 206. Urru, A., Bonini, M. and Echelmeyer, W., 2018, June. Planning of a milk-run systems in high constrained industrial scenarios. In 2018 IEEE 22nd International Conference on Intelligent Engineering Systems (INES) (pp. 000231-000238). IEEE.
- 207. Uzialko, A., 2022. Industry 4.0: How Technology Is Revolutionizing the Manufacturing Industry. <u>https://www.businessnewsdaily.com/10156-industry-manufacturing-iot.html. Last accessed 07/07/2022</u>.
- 208. Vallandingham, L.R., Strandhagen, J.W., Alfnes, E. and Strandhagen, J.O., 2017, September. Strategic Framework for Manual Assembly System Design. In International Workshop of Advanced Manufacturing and Automation (pp. 229-237). Springer, Singapore.
- 209. Veiga, J.T., Pessoa, M.A., Junqueira, F., Miyagi, P.E. and Dos Santos Filho, D.J., 2021, August. Intelligent Manufacturing Systems: Self-organization in the I4. 0 context. In 2021 14th IEEE International Conference on Industry Applications (INDUSCON) (pp. 153-160). IEEE.
- 210. Verbrugge, B., 2019. Best Practice, Model, Framework, Method, Guidance, Standard: Towards A Consistent Use Of Terminology – Revised. https://www.vanharen.net/blog/best-practice-model-framework-method-guidancestandard-towards-consistent-use-terminology/. Last accessed 23/08/2022.
- 211. Vosniakos, G.C., Deville, J. and Matsas, E., 2017. On immersive virtual environments for assessing human-driven assembly of large mechanical parts. Procedia Manufacturing, 11, pp.1263-1270.
- 212. Voss, C., Tsikriktsis, N., & Frohlich, M. (2002). Case research in operations management. International Journal of Operations & Production Management, 22(2), 195-291.
- 213. Weckenborg, C., Kieckhäfer, K., Müller, C., Grunewald, M. and Spengler, T.S., 2020.
 Balancing of assembly lines with collaborative robots. Business Research, 13(1), pp.93-132.
- 214. Webster, J., & Watson, R. T. (2002). Analyzing the past to prepare for the future: Writing a literature review. MIS quarterly, xiii-xxiii.
- 215. Voxco, 2022. A Complete Guide to Quasi-Experimental Design: Explanation, Methods & FAQs [Online]. Accesible: https://www.voxco.com/blog/quasi-experimental-design-explanation-methods-and-faqs/. Last accessed 20/05/2022.
- 216. Wänström, C. and Medbo, L., 2009. The impact of materials feeding design on assembly process performance. Journal of Manufacturing Technology Management.
- 217. Wikipedia, 2021. Exploratory research [Online]. Available: https://en.wikipedia.org/wiki/Exploratory_research. Last modified 28 October 2021.
- 218. Wolf, M. and Ramsauer, C., 2018, August. Towards an Age-Differentiated Assessment of Physical Work Strain. In Congress of the International Ergonomics Association (pp. 189-205). Springer, Cham.
- 219. Wu, S.Q., Shen, B., Tang, Y.Z., Wang, J.H. and Zheng, D.T., 2018. Ergonomic layout optimization of a smart assembly workbench. Assembly Automation.

- 220.Xiao Y, Watson M. Guidance on conducting a systematic literature review. Journal of Planning Education and Research. 2019 Mar;39(1):93-112.
- 221. Yao, F., Keller, A., Ahmad, M., Ahmad, B., Harrison, R. and Colombo, A.W., 2018, July. Optimizing the scheduling of autonomous guided vehicle in a manufacturing process. In 2018 IEEE 16th International Conference on Industrial Informatics (INDIN) (pp. 264-269). IEEE.
- 222. Yao, B., Zhou, Z., Wang, L., Xu, W., Yan, J. and Liu, Q., 2018. A function block based cyber-physical production system for physical human–robot interaction. Journal of manufacturing systems, 48, pp.12-23.
- 223. Yelles-Chaouche, A.R., Gurevsky, E., Brahimi, N. and Dolgui, A., 2020. Reconfigurable manufacturing systems from an optimisation perspective: a focused review of literature. International Journal of Production Research, pp.1-19.
- 224. Yetis, H., Karakose, M. and Baygin, N., 2022. Blockchain-based mass customization framework using optimized production management for industry 4.0 applications. Engineering Science and Technology, an International Journal, 36, p.101151.
- 225. Yin, R. K. (2017). Case study research and applications: Design and methods: Sage publications.
- 226. Yin, R. K. (2013). Case study research: Design and methods. Sage publications.
- 227. Yoshitake, H., Kamoshida, R. and Nagashima, Y., 2019. New automated guided vehicle system using real-time holonic scheduling for warehouse picking. IEEE Robotics and Automation Letters, 4(2), pp.1045-1052.
- 228. Zangaro, F., Minner, S. and Battini, D., 2021. A supervised machine learning approach for the optimisation of the assembly line feeding mode selection. International Journal of Production Research, 59(16), pp.4881-4902.
- 229. Zhang, Z., Tang, Q., Li, Z. and Zhang, L., 2019. Modelling and optimisation of energyefficient U-shaped robotic assembly line balancing problems. International Journal of Production Research, 57(17), pp.5520-5537.
- 230.Zhao, Z., M. Zhang, C. Yang, J. Fang, and G. Q. Huang. 2018. "Distributed and Collaborative Proactive Tandem Location Tracking of Vehicle Products for Warehouse Operations." Computers and Industrial Engineering 125: 637–648.
- 231.Zhong, Y.G. and Ai, B., 2017. A modified ant colony optimization algorithm for multiobjective assembly line balancing. Soft Computing, 21(22), pp.6881-6894.
- 232. Zhong, R.Y., Xu, X., Klotz, E. and Newman, S.T., 2017. Intelligent manufacturing in the context of industry 4.0: a review. Engineering, 3(5), pp.616-630
- 233. Zuin, S., Sgarbossa, F., Calzavara, M. and Persona, A., 2018. State of the art on design and management of material handling systems.
- 234. Zülch, M. and Zülch, G., 2017. Production logistics and ergonomic evaluation of U-shaped assembly systems. International Journal of Production Economics, 190, pp.37-44.

Part II: Collection of papers

- Paper 1: Dolgui, A., Sgarbossa, F. and Simonetto, M., 2022. Design and management of assembly systems 4.0: systematic literature review and research agenda. International Journal of Production Research, 60(1), pp.184-210.
- Paper 2: Simonetto, M. and Sgarbossa, F., 2020, August. Introduction to Material Feeding 4.0:
 Strategic, Tactical, and Operational Impact. In IFIP International Conference on Advances in Production Management Systems (pp. 158-166). Springer, Cham.
- Paper 3: Simonetto, M. and Sgarbossa, F., 2021, September. Straight and U-Shaped Assembly Lines in Industry 4.0 Era: Factors Influencing Their Implementation. In IFIP International Conference on Advances in Production Management Systems (pp. 414-422). Springer, Cham.
- Paper 4: Simonetto, M., Peron, M., Fragapane, G. and Sgarbossa, F., 2020, October. Digital assembly assistance system in Industry 4.0 era: a case study with projected augmented reality. In International Workshop of Advanced Manufacturing and Automation (pp. 644-651). Springer, Singapore.
- Paper 5: Simonetto, M., Arena, S. and Peron, M., 2022. A methodological framework to integrate motion capture system and Virtual Reality for assembly system 4.0 workplace design. Safety Science, 146, p.105561.
- Paper 6: Simonetto, M., Saporiti, N., (under review). Traditional Kanban, e-Kanban or Digital
 Twin: a Decision Support System for material management solution. Industrial
 Management & Data Systems.

Paper 1

Dolgui, A., Sgarbossa, F. and Simonetto, M., 2022. Design and management of assembly systems 4.0: systematic literature review and research agenda. International Journal of Production Research, 60(1), pp.184-210.

This is an accepted manuscript of an article published in International Journal of Production Research.

The final authenticated version is available online at: https://doi.org/10.1080/00207543.2021.1990433





International Journal of Production Research

ISSN: (Print) (Online) Journal homepage: https://www.tandfonline.com/loi/tprs20

Design and management of assembly systems 4.0: systematic literature review and research agenda

Alexandre Dolgui, Fabio Sgarbossa & Marco Simonetto

To cite this article: Alexandre Dolgui, Fabio Sgarbossa & Marco Simonetto (2021): Design and management of assembly systems 4.0: systematic literature review and research agenda, International Journal of Production Research, DOI: 10.1080/00207543.2021.1990433

To link to this article: https://doi.org/10.1080/00207543.2021.1990433

© 2021 The Author(s). Published by Informa UK Limited, trading as Taylor & Francis Group



Published online: 22 Oct 2021.

-	
	19.
L.	~
-	

Submit your article to this journal



🖸 View related articles 🗹



View Crossmark data 🗹



OPEN ACCESS Check for updates

Design and management of assembly systems 4.0: systematic literature review and research agenda

Alexandre Dolgui ¹^a, Fabio Sgarbossa ¹^b and Marco Simonetto^b

^aIMT Atlantique, Nantes Cedex, France; ^bDepartment of Mechanical and Industrial Engineering, Norwegian University of Science and Technology, Trondheim, Norway

ABSTRACT

Assembly systems (ASs) have moved into the era of mass customisation and Industry 4.0 (I4.0). Mass customisation involves a shift from the production of high quantities of the same product to the production of low quantities of a high number of different products. This is changing the way in which companies assemble their products and has introduced a certain number of challenges. For example, there are increases in the numbers of parts to be moved, the quantities of data to be collected, and the skills of the human workers that companies must manage to meet their customers' demand. The adoption of I4.0 technologies can help companies to face these challenges. However, although companies and researchers have studied possible solutions based on I4.0 technologies for ASs and have introduced the concept of Assembly System 4.0 (AS4.0), no studies have tried to understand how these technologies impact on decision areas at the strategic, tactical, and operational levels. In this paper, we attempt to fill this gap through a systematic literature review that not only offers the opportunity to understand the current situation and the state of the art in this field but also gives an overview of possible future research challenges.

1. Introduction

In recent years, companies have been faced with changes in demand from their customers (Battaïa et al. 2018). In fact, customers are no longer passive clients who buy only the products that companies offer them but are now more often active clients who ask for personalised, customised products that are closer to their needs and desires (Pollard, Chuo, and Lee 2008; Kucukkoc and Zhang 2017). These customisations may involve, for example, a change in the colour (aesthetic design), the shape (measurement), or the technical characteristics of some components (functionality) (Piller 2005). Hence, if companies decide to accept these requests from their customers, they must be ready to manage the production of more complex product models (Otto and Li 2020) and their assembly systems should be agile and reconfigurable (Battaïa et al. 2018). In fact, a reconfigurable system is designed with a certain level of flexibility, that can involve the reconfiguration of human workers, technologies, and equipment, in order to be able to face the rapid change in market conditions (Hashemi-Petroodi et al. 2020a, 2020b; Yelles-Chaouche et al. 2020; Dolgui et al. 2021).

Products can be customised in different phases of a production cycle (Da Silveira, Borenstein, and Fogliatto 2001; Hu et al. 2011), for example during the design, fabrication, or assembly stages, or at the moment that they are bought by the final customers. In this paper, we focus on the assembly systems (ASs) where the mass customisation is especially performed. Such ASs can be composed of the following two parts: the assembly line subsystem (ALS) and the assembly line feeding subsystem (ALFS) (Battini et al. 2009b).

An ALS is a type of production system in which various tasks are executed, at one or more workstations, to create the final product (Rekiek et al. 2002; Dolgui and Proth 2010; Akpinar, Elmi, and Bektaş 2017; Zhong and Ai 2017). Although they have the word 'line' in their name, ALSs can be designed in different shapes, such as a two-sided line, a U-shape line, a system with rotary table, a fixed position, and so on (Becker and Scholl 2006; Battini et al. 2011; Battaïa and Dolgui 2013). However, the straight simple line is the most commonly used layout (Rabbani, Moghaddam, and Manavizadeh 2012; Mukund Nilakantan and Ponnambalam 2016), and, in

CONTACT Marco Simonetto 🖾 marco.simonetto@ntnu.no 💿 Department of Mechanical and Industrial Engineering, Norwegian University of Science and Technology, S P Andersens V 3, Trondheim 7031, Norway

© 2021 The Author(s). Published by Informa UK Limited, trading as Taylor & Francis Group

This is an Open Access article distributed under the terms of the Creative Commons Attribution-NonCommercial-NoDerivatives License (http://creativecommons.org/licenses/ by-nc-nd/4.0/), which permits non-commercial re-use, distribution, and reproduction in any medium, provided the original work is properly cited, and is not altered, transformed, or built upon in any way.

ARTICLE HISTORY Received 2 April 2021

Accepted 26 September 2021

KEYWORDS Assembly; feeding; industry 4.0; design; management; state of the art

		Assembly Line Subsystem (ALS)	Assembly Line Feeding Subsystem (ALFS)
Long term	STRATEGIC	 Configuration of the system Level of automation 	Configuration of the systemLevel of automation
Medium term	TACTICAL	Workplace design Assembly line balancing	Feeding policy selectionScheduling
hort term	OPERATIONAL	Sequencing Control of the ALS	 Routing and material/information management Control of the ALFS

Table 1. Decision areas at different levels of a traditional AS.

fact, all types of layouts can be reconfigured or considered as a line in terms of how they work. Two main activities are executed in an ALS: the process, subassembly and assembly, which is related to the realisation of the final products by integrating together component parts and subassemblies; and control, which is related to monitoring of the quality and performance of the ALS (Hu et al. 2011; Battini et al. 2011).

The ALFS is responsible for the management and delivery of the components to the ALS (Battini et al. 2009b; Sali and Sahin 2016). The right component must be delivered in the right quantity, in the right container and at the right moment, to the right ALS. In order to achieve this, the three main activities of an ALFS are: transportation, preparation, and material management (Schmid and Limère 2019). Transportation involves the process of moving all components or parts from point A, where they are stored, to point B, where they are needed, while preparation relates to the processes of handling and repacking parts into the load carriers used for the corresponding line feeding policy. Finally, material management includes all processes related to the storage of components and products.

These subsystems need to be properly designed and managed in order to give an efficient and flexible AS, and several different decision areas have been defined in the literature, as summarised in Table 1.

These decision areas may be associated with different levels, for example strategic, tactical, or operational levels. Strategic decisions have a long-term impact (measured in years) on the company's operation, while tactical decisions impact operations in the medium term (over weeks), and operational decisions are made daily and have a short-term impact.

At the strategic level, decisions are made on the configuration of the system and the level of automation, for both ALSs and ALFSs. This means that companies must choose the product family has to be assembled and estimate how many resources are required by the family, how many ASs are necessary (for example, one flexible system for the whole product family or several more dedicated ones), the layout of their ASs, and the type of equipment, i.e. whether the different tasks associated with these systems are to be executed manually, automatically, or both (Bassan, Roll, and Rosenblatt 1980; Cormier and Gunn 1992; Roodbergen and Vis 2006; Wänström and Medbo 2009; Battini et al. 2011; Orru' et al. 2019; Fragapane et al. 2021).

At the tactical level, companies may carry out first workforce dimensioning (Dolgui et al. 2018), thus how many workers is necessary to function in worst case (large demand, complex product). Then this information together with time-and-motion and ergonomic analyses are necessary in order to design workplaces for the ALSs, and then assign these tasks to the workplaces using assembly line balancing algorithms (Lindenmeyer 2001; Boysen, Fliedner, and Scholl 2008; Battaïa and Dolgui 2013; Zülch and Zülch 2017). In the ALFS, decisions at this level firstly relate to how to deliver the different components to the ALS (Battini et al. 2009b; Caputo and Pelagagge 2011; Caputo et al., 2018; Arena et al., 2019), workforce dimensioning and then to the scheduling problem of who needs to deliver them (Mei et al. 2005; Dang et al. 2014).

At the operational level, the AS must work correctly, and therefore needs to be controlled. There can be also problems of reconfigurations when the product, production or demand change.

Different forms of data are collected from the AS in order to determine and control this process, for example the sequence of products produced in the ALS, the quantities of the components to stock in various warehouses, and the routes used to deliver these components to the ALS (Agrawal and Cohen 2001; DeCroix and Zipkin 2005; Hu et al. 2008; Battini et al. 2009a; Boysen, Kiel, and Scholl 2011; Gebser et al. 2018).

All these decisions need to be made for the AS in order to be able to generate the varieties of product required to meet customer demand. Despite the efforts of many companies, it is not easy to achieve mass customisation (Piller 2005; Piller 2007; Pollard, Chuo, and Lee 2008). This because companies need to be able to handle the three main elements of elicitation, process flexibility, and logistics if they want to attempt this goal (Radder and Louw 1999; Zipkin 2001; Blecker and Abdelkafi 2006; Roda et al. 2019; Hashemi-Petroodi et al. 2020a).

Process flexibility and logistics can be supported by Industry 4.0 (I4.0) technologies (Lasi et al. 2014; Shrouf, Ordieres, and Miragliotta 2014). I4.0 is the term used to refer to the fourth industrial revolution (Kirazli and Hormann 2015), which offers the opportunity to use new technologies such as collaborative robots (cobots), mobile robots, augmented reality (AR), virtual reality (VR), the Internet of Things (IoT), cloud computing and data analysis, etc. with the promise of increased flexibility, higher levels of automatisation, better quality, and improved productivity (Thames and Schaefer 2016; Zhong et al. 2017).

Companies and researchers are already studying the implementation of these technologies in relation to ASs and have introduced the concept of Assembly System 4.0 (AS4.0) (Bortolini et al. 2017; Cohen et al. 2019b). Bortolini et al. (2017) studied how I4.0 technologies can impact in the AS4.0. In particular, they proposed and described in detail the distinctive characteristics of such technologies in these new systems. With the same purpose, Cohen et al. (2019b) created a road map to understand and investigate the impact of I4.0 technologies on AS4.0, at the three impact levels: strategic, tactical, and operational. On each level they explored the different uses that the technologies can have in the AS4.0 and what is specific for AS4.0 in terms of functionalities and performances.

However, these researchers have mainly focused on the introduction of such systems or part of it, its proofof-concept development, analysis of their characteristics and their performance, while in limited way they have investigated decisions in strategic, tactical, and operational levels applied to AS4.0, as it will be analysed in the following sections.

Thus, in line with the aims and scope of International Journal of Production Research, which is to disseminate research on decision aid on assembly systems among the others, we want to investigate how I4.0 technologies are likely to change the decision areas of an AS, and to provide an original vision of perspectives and challenges to the readers of IJPR and researchers in the production research field.

We pursue the main contribution of the paper answering the following three research questions: How are Industry 4.0 technologies applied to AS4.0? How do these technologies impact on the decision areas of AS4.0? What are the future needs of research into AS4.0? To answer these questions, we conducted a systematic review of the literature. In Section 2, the methodology used in this study is introduced. In Section 3, we present a descriptive analysis of the selected papers, and we analyse these in Section 4 and summarise the results in Section 5. Section 6 reflects on research challenges, and Section 7 concludes the work.

2. Methodology

In this section, following the guidelines outlined by Tranfield, Denyer, and Smart (2003), we explain the process used to select the papers included in our review. In a systematic literature review, the steps applied to select the works need to be clearly defined, in order to create a process that is easily replicable (Seuring and Gold 2012). For this reason, we describe the decisions that were made in order to create a collection of papers, as follows:

(1) Keywords: Based on the literature, two groups of keywords were defined, as shown in Table 2. The keywords in Group A were related to the main topic of these papers, i.e. the ALS and ALFS (Battaïa and Dolgui 2013; Battini et al. 2015; Bortolini et al. 2017; Schmid and Limère 2019). The choice of keywords in Group B was inspired by the work of Winkelhaus and Grosse (2020) and is also supported by the literature on I4.0 (Schwab 2016; Culot et al. 2020) and assembly (Bortolini et al. 2017; Cohen et al. 2019b). We performed the search process, as suggested in Hosseini and Ivanov (2020), based on a combination of keywords. Therefore, we used logical operators 'AND' and 'OR' to create Boolean

Table 2. Groups of keywords related to ASs and I4.0.

Group A	Assembly, Workstation, Feeding, Supermarket
Group B	4.0, Smart, Mass customisation, Individualisation, Human- machine, Assisted operator, Adaptive workplace, Internet of things, IoT, Internet of Services, Cyber-physical, Cybersecurity, Blockchain, Social media, Mobile services, Mobile robot*, Autonomous robot, Cobots, Collaborative robot, Augmented reality, Virtual reality, Big data, Cloud, Exoskeleton, Gamification

4 👄 A. DOLGUI ET AL.

keywords combinations as by Hosseini, Ivanov, and Dolgui (2019), '(keyword of group A) AND (keyword of group B OR another keyword of group B)', in order to generate the queries used in the Scopus database.

- (2) Refinement: We used several different combinations of criteria in the first refinement. We considered articles and reviews written in English as document types, limiting our search from January 2005 to December 2020. Following Winkelhaus and Grosse (2020), we also considered only the subject areas of 'Computer Science', 'Engineering', 'Economics', 'Management', 'Social Science' and 'Decision Science'. From these, we excluded papers in fields that were outside our scope of interest, for example biology, bioinformatics, fusion engineering, and design. A combination of the keywords with these limitations gave us an initial selection of 16,849 papers, which was reduced to 9424 after we removed duplicates and considered only journals with a Scimago Journal Rank (SJR) greater than or equal to 0.5. The SJR is an index used to describe the prestige of a journal (SCIMAGO 2020) and can be used to classify and select relevant journals (Falagas et al. 2008; Delgado-López-Cózar and Cabezas-Clavijo 2013).
- (3) Inclusion and exclusion criteria: From reading the title, abstract and keywords of the papers, we were able to identify works that addressed the use of the different I4.0 technologies in an AS. We were then able to remove all of the papers that were related to topics that were outside the scope of the review, which were not immediately identified as irrelevant in the first refinement step, e.g. material science, nanotechnology, and chemistry. Finally, we were left with 412 potentially relevant papers.
- (4) Second refinement: In this step, we read all of the articles in full. This allowed us to remove papers that appeared relevant from the abstract, but which turned out to be outside our scope once we read them in their entirety, for example papers that were related to AR or VR and which showed how the authors created their solutions but not how these were then implemented. Our vision of the domain as well as our appreciation of the scientific levels of papers were also used at this step. After this refinement, 140 papers remained.
- (5) Snowball search: At this stage, 17 articles were added through a backward snowball search, which was carried out in order to discover relevant papers that were not identified in the first phases. At the end of this step, 157 papers were selected for inclusion in this review.



Figure 1. Flowchart of the different phases of the systematic literature review.

Figure 1 summarises the research process and shows the number of papers remaining at each step.

The final set of 157 papers was divided into two groups, representing the two different parts of the AS. In a few cases, the papers dealt with both subsystems, but we decided to assign each one to a specific subsystem based on the main scope of the research. In total, 90 articles were related to the ALS and 67 to the ALFS. As in Kumar et al. (2020), our analysis is decomposed in a descriptive analysis (Section 3) and a content analysis (Section 4) both are based on these two sets of papers.

3. Descriptive analysis

A descriptive analysis of the papers included in this work is conducted in this section, with the aim of presenting some preliminary quantitative results and to motivate our work. These quantitative results are interesting, because they highlight the need for research of this type.

From Figure 2, we can see that the numbers of papers related to I4.0 technologies for ASs start to increase from 2015. In particular, the numbers of papers almost double each year from 2018, indicating an increase in the interest in this topic.

INTERNATIONAL JOURNAL OF PRODUCTION RESEARCH 😔 5



Figure 2. Number of papers examined by year of publication.

Figure 3. Papers related to the AS, divided by level.











Figure 3 shows that the papers related to the ALFS were well distributed over the strategic, tactical and operational levels, while this was not the case for the ALS. For the latter, researchers focused their attention at the operational level, and the topics were mostly related to the quality control of products or performance evaluations of the system.

From Figure 4, it can be observed that almost all of the papers present a case study, experiments or simulations in order to study the technology that they are considering. Figure 5 shows the I4.0 technologies that are most frequently studied in relation to AS4.0. For the ALS, IoT technologies were the technology most commonly investigated (43 papers), followed by cobots (24 papers), Data Analysis (19 papers), AR (13 papers), and VR (12 papers). The papers related to the family of IoT technologies in our case include sensors, cyber physical systems (CPSs), digital twins and wearables. For the ALFS, we can see that the most widely adopted technologies are mobile robots (53 papers) and IoT technologies (8 papers). In summary, based on this figure, the main technologies related to AS4.0 are cobots, mobile robots, AR, VR, IoT, and cloud computing and data analysis. This categorisation of I4.0 technologies is based on the analysis of the selected papers. Their definitions in the following paragraph will help us in understanding their implementation in ASs and how they impact on decisions at strategic, tactical and operational levels.

Cobots are 'robots intended to physically interact with human workers in a shared workplace' (Djuric, Urbanic, and Rickli 2016). Mobile robots can be defined as 'industrial robots that use a decentralized decision-making process for collision-free navigation to provide a platform for material handling, collaborative activities, and full services within a bounded area' (Fragapane et al. 2021). AR is defined by Carmigniani et al. (2011) as 'a real-time direct or indirect view of a physical real-world environment that has been enhanced/augmented by adding virtual computer-generated information to it'. For Riva (2002) VR is 'A collection of technologies that allow people to interact efficiently with 3D computerized databases in real time using their natural senses and skills'. Xia et al. (2012) state that 'IoT technologies refers to the networked interconnection of everyday objects, which are often equipped with ubiquitous intelligence'. For Armbrust et al. (2010) cloud computing refers 'to both the applications delivered as services over the Internet and the hardware and systems software in the data centers that provide those services'. Oztemel and Gursev (2020) say that data analysis is 'having the capability to handle big amount of data and performing well defined analysis to be able to run the overall system aligned with the manufacturing goals'.

From Figure 5, we can see that there is scientific evidence that the system has changed. However, it is still necessary to study how I4.0 technologies are likely to impact decision areas associated with the design and management of ASs.

4. Content analysis

Krippendorff (2004) defined content analysis as: [a] research technique for making replicable and valid inferences from texts (or other meaningful matter) to the contexts of their use.

Content analysis is a technique that involves specialised procedures; its results should be replicable and valid, and it should help researchers to describe and quantify specific phenomena (Downe-Wamboldt 1992; Krippendorff 2004). In this section, we divide the two sets of papers (on the ALS and ALFS) based on the level (strategic, tactical, and operational) at which they examine these technologies. The main purpose of this analysis is to help us to answer our three research questions, which we repeat here for the reader:

RQ1: How are Industry 4.0 technologies applied to AS4.0?

RQ2: How do these technologies impact on the decision areas of AS4.0?

RQ3: What are the future needs of research into AS4.0?

4.1. Strategic level in assembly system 4.0

A decision made at the strategic level will influence the decisions that can be made at the tactical and operational levels. A wrong decision made at this level can compromise an entire project, and it is therefore important for companies to understand clearly what they want to achieve.

Assembly Line Subsystem: The introduction of new I4.0 technologies such as cobots and assistive technologies to an AS offers the opportunity to create new, more dynamic and flexible configurations. However, it is important to understand how to introduce these technologies into an AS, and whether it is economically beneficial to introduce them. In view of this, two different methodologies for the configuration of a collaborative assembly workplace have been presented by Mateus et al. (2019) and Stadnicka and Antonelli (2019), and an economic evaluation of several different configurations was carried out by Peron, Sgarbossa, and Strandhagen (2020). Mateus et al. (2019) created a four-block procedure that takes into consideration the safety, ergonomics, and performance of the system, whereas the methodology proposed by Stadnicka and Antonelli (2019) instead relies on the implementation of lean methods and tools. Both articles present a case study to demonstrate the validity of the methodologies. Peron, Sgarbossa, and Strandhagen (2020) created a decision tree based on cost models, which was validated through simulation and a case study, to allow the user to understand when it is economically advantageous to create a configuration involving a cobot, AR or both.

The choice of a collaborative human-robot layout requires an understanding of the level of interaction that the human will have with the robot and the skills that will be necessary to make the system work. In order to identify these, Kolbeinsson, Lagerstedt, and Lindblom (2019) proposed several levels of collaboration in their theoretical work, which can help companies that want to shift from automation to collaboration. After studying the state of the art in human-robot collaboration (HRC), Wang et al. (2017) proposed a framework that considered the fundamental elements of an HRC scenario (the actors, work environment, workpieces and operations) to create a symbiotic collaboration between humans and

INTERNATIONAL JOURNAL OF PRODUCTION RESEARCH 😔 7



Figure 5. Industry 4.0 technologies for AS.

robots. For Hashemi-Petroodi et al. (2020b), a welldesigned human-robot collaboration (HRC) system must consider the three features of resource skills, ergonomics, and resource flexibility. These authors evaluated these features to compare the interaction between humans and cobots in two different scenarios: dual resourceconstrained and human-robot collaboration. The cooperation between humans and robots was also carefully investigated by Krüger, Lien, and Verl (2009); through a survey, they studied the organisational and economic aspects of this collaboration. Cyber-physical production systems and human-robot interaction were examined by Yao et al. (2018), and the framework that they proposed was validated based on an assembly case in which a human needed to move heavy parts frequently.

Cobots are not the only assistive technology that can be used in an AS; AR can be used to guide human workers in their tasks, while VR is mainly used for workplace design. However, before implementing these approaches, it is important to know how to use them and whether they will be convenient, and Danielsson, Holm, and Syberfeldt (2020), Miller, Hoover, and Winer (2020), and Marques et al. (2020) therefore attempted to address these two points. Danielsson, Holm, and Syberfeldt (2020) studied the state of the art in relation to the use of AR smart glasses, while Miller, Hoover, and Winer (2020) focused their attention on the Microsoft HoloLens, and tried to mitigate its limitations in terms of input, field of view, tracking and occlusion. They developed a visualisation application in order to overcome these restrictions. Marques et al. (2020) compared three different AR interaction methods for assembly procedures via experiments and a questionnaire. Their results can help in the choice of the best procedure, depending on the aims of the company. The creation of a layout containing an ALS, with or without a robot that can work together with a human, can be facilitated by the development of VR environments. Krishnamurthy and Cecil (2018) created a virtual environment based on data collected by sensors, in which users could interact with the system, and studied different layout alternatives and assembly processes.



Assembly Line Feeding Subsystem: Although other technologies were also applied, mobile robots were the main technology explored in relation to the ALFS. These transportation devices, unlike other equipment such as forklifts, can open up opportunities to create new guide paths using different new methods. Antonelo, Schrauwen, and Van Campenhout (2007) applied a machine learning technique called a reservoir computing network, and Mantegh, Jenkin, and Goldenberg (2010) used harmonic functions and the boundary integral equation method to create different path planning algorithms. A Bézier curve and a genetic algorithm were the approaches applied by Song, Wang, and Sheng (2016) and Elhoseny, Tharwat, and Hassanien (2018), while Unhelkar et al. (2018a) developed a mobile robot that was able to generate its path on a moving floor that was similar to a conveyer belt. Gao et al. (2019) created a new path evaluation function to estimate the localisability of the mobile robots. The importance of spike latency in path planning was highlighted by Koul and Horiuchi (2019), and a global path planning method based on an episodic-cognitive map was presented by Zou et al. (2019). Moysis et al. (2020) reported the performance of their chaotic path planning algorithm, which was enhanced with a pheromone-inspired memory technique. The energy limitations of mobile robots were taken into consideration by Jensen-Nau, Hermans, and Leang (2020) when finding the optimal path based on a Voronoi-based generation algorithm.

The extensive implementation of sensors in mobile robots gives the opportunity to create new path configurations that can guarantee better performance in terms of both time and safety. Indeed, the sensors in mobile robots are controlled by an advanced hardware and control software that allows them to do autonomous operations and communicate with other resources in the dynamic environments in which they perform their tasks (Fragapane et al. 2021). A collision-free path study can be performed in a static environment (Al Al-Dahhan and Schmidt 2020; Das and Jena 2020; Saeed, Recupero, and Remagnino 2020), in a dynamic environment (Qu et al. 2009; Muthukumaran and Sivaramakrishnan 2019; Panda, Das, and Pradhan 2017), or both (Chang et al. 2020). Simulations and experiments were used by these authors to compare their results with other solutions and hence to determine the performance of their models. Al-Dahhan and Schmidt (2020) and Saeed, Recupero, and Remagnino (2020) first found possible solution paths, and then refined them to find the optimal path. In the models created by Muthukumaran and Sivaramakrishnan (2019), Das and Jena (2020), and Panda, Das, and Pradhan (2017), each robot can take the optimal trajectory path without colliding with other robots or obstacles within the environment. Qu et al. (2009) proposed a model that needed to know in advance the environment in which it would be applied, while Chang et al. (2020) proposed a solution that was able to adapt to the unknown environments in which mobile robots perform their activities.

Before implementing a technology, it is essential to understand whether it is convenient to use it and which technology would be best to use, based on what the company aims to do. In view of this, Fragapane et al. (2020) carried out various simulations in an attempt to understand whether or not the use of mobile robots in a material handling system was advantageous. Fager et al. (2019) used experiments in their work to compare paperbased picking, light-based picking and voice-based picking with the use of AR to support picking activities for kit preparation.

The use of IoT and cloud computing can help in creating new configurations for warehouses. Zhou et al. (2019) applied these technologies to collect, store, and share data from different devices in real time in a warehouse, in order to compare two different layout configurations through a simulation.

4.2. Tactical level in assembly system 4.0

At this level, the company has already chosen the technologies that will be used, and now needs to make all the decisions required in order to allow these technologies to work.

Assembly Line Subsystem: The design of a workplace with an ALS may change if a cobot is introduced into it. Faccio et al. (2020) highlighted that it is not only the type of cobot that influences the possible collaboration with human workers and the design of the workstation, but also the characteristics of the products that are being assembled. In their work, Gualtieri et al. (2020) and Prati et al. (2020) presented case studies of the transformation of a manual workplace into a collaborative one, with improvements in its productivity and physical ergonomics. In contrast, Michalos et al. (2018b) presented a method for the transformation of a fully automated line to a collaborative one. This design, with or without a cobot, could be optimised using AR, VR, wearables and sensors. Michalos et al. (2018a) examined the use of VR in a case study with the aim of reducing the time and cost requirements of the process, and achieving the design or redesign of a workplace. Wu et al. (2020) and Yi et al. (2020) proposed two different solutions based on the application of sensors to create a smart workplace. Havard et al. (2019) used both VR and sensors to explore the design of a workplace containing a cobot. The optimisation of a workplace does not only involve the evaluation of production performance, and in order to determine whether it is well designed, an ergonomic evaluation may also be necessary. Enomoto, Yamamoto, and Suzuki (2013), Azizi, Yazdi, and Hashemipour (2019), and Peruzzini et al. (2020a) used VR to design workplaces in which the ergonomic parameters of assembly tasks were respected. Gao, Shao, and Liu (2016), Plantard et al. (2017), Wu et al. (2018), Xiao et al. (2018) and Bortolini et al. (2020) instead applied sensors to collect different forms of data in order to design an ergonomic workplace.

The data collected from time-and-motion and ergonomic analyses of an AS can then allow for the assignment of tasks within a workplace. Papakostas et al. (2016) created a software application that could assign assembly tasks in an AS based on the data available at the moment of the request. Huo et al. (2020a) and Huo, Zhang, and Chan (2020b) developed two real-time monitoring systems that were capable of reassigning tasks based on the states of the different resources in the AS. A lack of data can lead to the incorrect assignment of tasks. Fantoni et al. (2020) created a method of automatic time measurement that could precisely determine the duration of each task. An architecture based on four layers that could help to collect and share data, thus offering the opportunity to flexibly re-assign these tasks if necessary, was proposed by Qian et al. (2020).

The balancing of an AS that implements several cobots requires a knowledge of which tasks can be executed by the cobots and which cannot. A method of assigning tasks to humans or robots based on their skills, regardless of workload balancing, was presented by Bruno and Antonelli (2018), and Bilberg and Malik (2019) created a digital twin for the same purpose. Two models for the assignment of tasks with the aim of minimising the cycle time were put forward by Weckenborg et al. (2020) and Çil et al. (2020). Dalle Mura and Dini (2019) and El Makrini et al. (2019) proposed two frameworks that also took into consideration the ergonomics of the human workers when allocating human-robot assembly tasks. Human workers can be trained before executing their tasks in the workplace using VR and AR (Yuan, Ong, and Nee 2008; Zhang, Ong, and Nee 2011; Hoedt et al. 2017). Galambos et al. (2015) used VR to replicate a work environment in which humans could be trained by interacting with the system. The effects of oral, paper-based and AR-based instructions during assembly tasks were studied and compared by Vanneste et al. (2020). An analysis of literature on workforce reconfiguration strategies in assembly lines, also considering the human-robot collaboration, is given in (Hashemi-Petroodi et al. 2020a).

The introduction of technologies such as IoT, wearables, and cloud computing to AS processes may require new frameworks or guides to enable an understanding of how to achieve this. Guo et al. (2020a) tried to cover this gap by presenting step-by-step guidelines for the implementation and transformation of an AS to a AS4.0. Tan et al. (2019) instead explained how it was possible to successfully implement a CPS in an AS.

Assembly Line Feeding Subsystem: After the choice to adopt mobile robots has been made, companies need to understand and determine how many of them are needed. Choobineh, Asef-Vaziri, and Huang (2012) estimated this number for a fleet of mobile robots. The quantity required can be influenced by the feeding policies used by companies to bring the components to the ALS, and the use of mobile robots offers the opportunity to apply a range of different feeding policies. Boysen, Briskorn, and Emde (2017), Yoshitake, Kamoshida, and Nagashima (2019), Gharehgozli and Zaerpour (2020), and Jiang et al. (2020), for example, proposed different algorithms in which mobile robots were used to move entire shelves from a warehouse to an ALS. Groß and Dorigo (2009) studied the possibility of connecting two or more mobile robots based on the type of object that they need to transport. Three different examples with collaborative robots were presented by Andersen et al. (2017), Unhelkar et al. (2018b), and Fager, Calzavara, and Sgarbossa (2020). In the studies by Andersen et al. (2017) and Fager, Calzavara, and Sgarbossa (2020), a cobot was installed above a mobile robot. This solution was used in the former to feed an ALS, and in the latter to pick components from a warehouse for kit preparation. Unhelkar et al. (2018b) instead studied the possibility of using a cobot with single-axis mobility to pick and transport the components needed to feed the ALS.

When a company has the required number of mobile robots and has decided how to feed the ALS, it is necessary to schedule the robots. Lo, Zhang, and Stone (2020) created an algorithm for the scheduling of mobile robots that took into consideration both the tasks executed by the robots and their movements. Bocewicz, Nielsen, and Banaszak (2014), Dang et al. (2014), Caridá, Morandin, and Tuma (2015), Zeng, Tang, and Yan (2015), Petrović et al. (2016), Zhou and Xu (2018), Dang, Nguyen, and Rudová (2019), Lyu et al. (2019), Kousi et al. (2019), Petrović, Miljković, and Jokić (2019), Hari, Nayak, and Rathinam (2020), and Rahman, Janardhanan, and Nielsen (2020) proposed various algorithms for the scheduling of mobile robots. Cloud computing can help in optimising this scheduling. In their work, Nielsen et al. (2017) created a cloud-based application that was responsible for all communication and exchange of data between mobile robots and the other actors in an ALS. The collection of these real-time data enabled scheduling of the mobile robots based on the current situation of the ALS. Wan et al. (2017) proposed a context-aware cloud robotic entity for advanced material handling. This solution offered the opportunity to schedule mobile robots in a more energy-efficient and cost-saving way.

IoT can be applied to help with the design of a warehouse or the space needed by an ALS to stock materials. Lyu et al. (2020) proposed an IoT solution that could optimise the positioning of the containers in a warehouse based on their dimensions, while Xu et al. (2020) created a system that was aware of the number of goods stocked, and those which could be stocked, in a specific space in a warehouse. The IoT can also improve the performance of the warehouses. Lee et al. (2018) demonstrate with a case study how the implementation of sensors in the warehouse generate a positive impact on the warehouse productivity, picking accuracy and efficiency.

4.3. Operational level in assembly system 4.0

At this level, a company must ensure that the technologies are able to work, and more importantly, it must enable them to continue to work as desired. This means that companies need to control not only whether a technology does what they want, but also how it does it. To achieve this, the AS must be constantly monitored, and if something is not working properly it must be fixed as quickly as possible.

Assembly Line Subsystem: Different sensors were used by Huang et al. (2008), Bauters et al. (2018), Li, Ota, and Dong (2018), Faccio et al. (2019), Alavian et al. (2020), Baumann et al. (2020), and Peruzzini, Grandi, and Pellicciari (2020b) to collect data from machines, human workers, products, and various forms of equipment for use in an AS. These data were used to monitor and control the AS, and allowed the user to create a digital copy or digital twin of the all the actors in an AS. Bao et al. (2020) explained how to create a digital twin application, and such applications were reviewed by Cimino, Negri, and Fumagalli (2019), who then presented a case study of a digital twin for energy consumption monitoring. Zhuang, Liu, and Xiong (2018) and Nikolakis et al. (2019a) proposed two different digital twin solutions for monitoring an AS. The data necessary for this monitoring can be also obtained from videos and pictures of the AS, and Tarallo et al. (2018), Oyekan et al. (2019), and Chen et al. (2020c) collected the data used in their studies from an analysis of videos and pictures of assembly activities. These data allowed the authors to monitor and improve the performance of the assembly processes under study.

Once collected, data need to be stored and analysed. Alexopoulos et al. (2016), Alexopoulos et al. (2018), and Guo et al. (2020b) showed how cloud computing could be used not only to collect data but also to analyse them and to share the information created from them. The information generated in this way is important, because it allows a company to be aware of what is happening in an ALS. Liu, Li, and Wang (2015a), Xu et al. (2016), Liu et al. (2017), Nuzzi et al. (2020), Ruppert and Abonyi (2020) and Tao, Leu, and Yin (2020) used information created from the collection of real-time data as a basis for the operation of their solutions and to understand how to improve them.

Although the performance of an ALS can be improved through the introduction of cobots, it is important that these devices are controlled. Mohammed, Schmidt, and Wang (2017) proposed a solution that used images of human operators taken with a depth camera and a virtual model of a robot for monitoring and collision detection. Nikolakis, Maratos, and Makris (2019b) and Malik and Brem (2020) proposed a CPS and a digital twin, respectively, to guarantee the safety of human workers who operate alongside a robot. Two different AR solutions that showed human workers the spaces in which they could perform their tasks without coming into contact with the robots were proposed by Hietanen et al. (2020).

In their work, Liu et al. (2015b) and Wang, Rizqi, and Nguyen (2020b) studied AR and computer vision, respectively, as methods of guiding human workers in real time in the execution of their tasks, while Chen et al. (2020b) used both AR and computer vision: AR as a guide for the human workers, and computer vision to monitor the assembly process. Although human workers can be guided during their activities, they may still make mistakes, and hence several solutions for greater control over the quality of the products, using different sensors and techniques, were proposed by Yu and Wang (2013), Lei et al. (2017), Colledani et al. (2018), Jiang et al. (2019), Chen et al. (2020a), Negri et al. (2020), Runji and Lin (2020), Wagner et al. (2020), Wang et al. (2020a) and Židek et al. (2020). The collection of information about these errors can allow a company to create databases of similar situations, which can then be studied in order to avoid them happening again. For example, Carvajal Soto, Tavakolizadeh, and Gyulai (2019) and Sassi, Tripicchio, and Avizzano (2019) used machine learning and deep learning, respectively, to analyse data collected from quality inspections of assembled products. A machine learning technique was also used by Kucukoglu et al. (2018) to analyse the data collected by a digital glove and hence to reduce quality errors during assembly. Liu et al. (2017) instead used computer vision to inspect the assembly process and to detect redundant objects that could compromise it.

Assembly Line Feeding Subsystem: As soon as a company has decided which mobile robots to use, based on a scheduling process, it then needs to decide on their routes. In view of this, Nishi and Tanaka (2012), Ohnishi and Imiya (2013), and Moussa and ElMaraghy (2019) proposed three different methodologies for routeing mobile robots to enable them to carry out their activities. Two methods that can allow mobile robots to follow certain routes in dynamic environments were proposed by Walker, Garrett, and Wilson (2006) and Posadas et al. (2008).

Armesto et al. (2008), Li et al. (2014), Chien, Wang, and Hsu (2017), Zhao et al. (2018), Bencherif and Chouireb (2019), Filotheou et al. (2020), and Halawa et al. (2020) focused their attention on the localisation of mobile robots in order to control them and to understand whether they were following the correct routes. Tracking the positions of mobile robots can help in increasing the performance of the ALFS, and the addition of other sensors can also improve also its safety. A range of different sensors were therefore applied to mobile robots and their environments by Huang et al. (2015), Indri et al. (2019), Keung et al. (2020), and Luo et al. (2019), to avoid collisions between them.

Mobile robots and human workers involved in transportation and preparation activities cannot do their work if they cannot find the components that they need to transport or pick. Tejesh and Neeraja (2018) implemented a warehouse inventory management system that used sensors to monitor the mobility and storage of products throughout a company. The use of sensors in a warehouse and in an ASL can control the quantities of components present in each container in real time. The data generated by such sensors were used in works by Kartal et al. (2016), Lolli et al. (2017), and Lolli et al. (2019), who applied three different machine learning techniques to analyse the collected data in order to optimise the inventory levels of the warehouses and the ALSs studied.

	AL	S	ALFS			
Technology	Assembly	Control	Transportation	Preparation	Material management	
Collaborative robots	Х			Х		
Mobile robots			Х	Х	Х	
Augmented Reality	Х	х		Х		
Virtual Reality	Х	х				
IoT technologies	Х	х	Х	х	Х	
Cloud computing and Data Analysis	Х	Х	Х	х	Х	

Table 3. I4.0 technologies for an AS, based on ALS and ALFS activities.

5. Insights

In this section, we aim to answer our first two research questions.

14.0 technologies applications in AS4.0: The ways in which I4.0 technologies are applied in AS4.0, and hence the answer to RQ1, can be observed from Table 3 which shows these different technologies based on the activities in which they have been implemented. For the ALS, the technologies are divided into the two activities of assembly and control, while for the ALFS, they are divided into the three activities of transportation, preparation, and material management. Table 3, in comparison with the results of previous works, like for example, those that focus more on how these technologies are generally used in AS4.0 (Cohen et al. 2019a; Cohen et al. 2019b), allows us to see where these technologies can create new opportunities or applications in the activities of the AS4.0, and particularly those areas in which they are not being used at the moment.

From Table 3, we can see that there are two groups of technologies (IoT/cloud computing and data analysis) that are used in all the activities of an AS4.0. This result is as expected, due to the nature of I4.0. IoT technologies allow companies to collect all the available data from the AS4.0 (Huang et al. 2008; Lee et al. 2018; Alavian et al. 2020; Xu et al. 2020). These data then need to be stored and shared, and cloud computing is very valuable in this regard (Papakostas et al. 2016; Alexopoulos et al. 2018; Guo et al. 2020b). Cloud computing can simultaneously offer both the resources needed to save and transform these data into information, using data analysis techniques, and then to deliver them to the users who need them (Lolli et al. 2017; Carvajal Soto, Tavakolizadeh, and Gyulai 2019).

Differences were seen for VR and mobile robots, as these two technologies were used only in the activities of the ALS (Galambos et al. 2015; Hoedt et al. 2017; Havard et al. 2019) and the ALFS, respectively (Choobineh, Asef-Vaziri, and Huang 2012; Kousi et al. 2019). Although this result was expected for the use of mobile robots, we did not expect this in the case of VR. In fact, we did not find any work that addressed the use of VR in an ALFS, despite the fact that it can be used in simulations, for example in the study of human workers working within the same space as mobile robots.

The use of cobots and AR was found in both ALSs and ALFSs. In particular, they were used in the preparation activities of ALFs, where they were applied to carry out picking and kitting (Andersen et al. 2017; Kim, Nussbaum, and Gabbard 2019; Fager, Calzavara, and Sgarbossa 2020; Fang and An 2020). In this scenario, a cobot can be installed above a mobile robot so it can do its tasks by following a human worker. For ALSs, it is interesting to note that AR was used to train and guide human assembly workers and to control the quality of the assembly (Westerfield, Mitrovic, and Billinghurst 2015; Michalos et al. 2018b; Hietanen et al. 2020; Vanneste et al. 2020).

I4.0 technologies impact on the decision areas of AS4.0: The ways in which these technologies impact on the decision areas of the AS4.0, and hence the answer to RQ2, can be seen from Tables 4 and 5, which summarise the decisions that companies need to make when they decide to introduce a specific technology to their AS. These decisions are divided based on the different levels (strategic, tactical, and operational) at which they need to be made. They are also related to the decisions made in a traditional AS (see Table 1 above). This allows us to see where each specific decision for each technology fits in with respect to more traditional methods. Tables 4 and 5 also allow us to answer RQ3, which relates to the future needs of research into AS4.0, and this is discussed in the next section.

Strategic: When we talk about the configuration of the system for a technology in an AS4.0, we are referring to what it is necessary to know before deciding which technology would be best to implement in the AS. For example, the integration of a cobot into an AS can change the spaces that are necessary to allow the robot to work alongside humans, if a company wants to create instructions to train their human workers in a virtual environment, without the need for a real one, it may prefer to use VR. In contrast, if the company wants to guide their human workers during the execution of their tasks within

12 🛞 A. DOLGUI ET AL.

			Industry 4.0 Technologies in Assembly Line Subsystem (ALS)					
	Decision areas		Collaborative robots	Augmented Reality	Virtual Reality	IoT technologies	Cloud computing and Data Analysis	
Medium term Long term	Strategic	Configuration of the system	Type of cobot	Decision of which AR instructions create	Decision of which VR instructions create	 Type of IoT technologies 	 Data to storage and analysis 	
		Level of automation	 Type of cobot Skills to use the cobot 	 Type of AR devices Skills to use the AR devices 	 Type of VR devices Skills to use the VR devices 	 Type of IoT technologies Skills to use the IoT technologies 	 Type of Cloud Computing system and/or Data Analytics techniques Skills to use these technologies 	
	Tactical	Workplace design	Workplace design	 How to create and share the instructions 	Workplace design	Workplace design	 Information to create and share How to create the information 	
		Assembly line balancing	 Balancing of the assembly line with the cobots 	 Number of AR devices 	 Number of VR devices 	 Number of IoT tecnologies 	 Number of information to create 	
Short term	Operational	Sequencing	Sequencing of the operations	 Real-time information sharing → assembly instructions 	 Real-time information sharing → assembly environment and instructions 	 Real-time control of the quality of the assembled products, machines, workers, and environment 	 Real-time sharing and displaying of the information 	
		Control of the ALS	Control of the system	Control of the system	Control of the system	Control of the system	Control of the system	

Table 4.	Decision	areas divided	into diffe	ent levels f	or each I4.	0 technoloc	y in an ALS.
----------	----------	---------------	------------	--------------	-------------	-------------	--------------

Table 5. Decision	n areas divided into	different levels for	r each I4.0 technolo	ogy in the ALFS.
-------------------	----------------------	----------------------	----------------------	------------------

			Industry 4.0 Technologies in Assembly Line Feeding Subsystem (ALFS						
	Decision areas		Mobile robots	Augmented Reality	IoT technologies	Cloud computing and Data Analysis			
Long term		Configuration of the system	 Warehouse design Type of mobile robots Design the guide path 	 Decision of which AR instructions create Type of AR devices 	Type of containers	 Data to storage and analysis 			
	Strategic	Level of automation	 Type of mobile robots Skills to use the mobile robots 	 Type of AR devices Skills to use the AR devices 	 Type of IoT technologies Skills to use the IoT technologies 	 Type of Cloud Computing system and/or Data Analytics techniques Skills to use these technologies 			
Medium term	Tactical Scheduling	Feeding policy selection	Feeding policies	 How to create and share the instructions 	 Positioning of the IoT technologies 	 Information to create and share How to create the information 			
		Scheduling	 Scheduling of the mobile robots Number of mobile robots 	Number of AR devices	 Number of containers in the warehouses and in the ALS 	 How to share the information 			
Short term	Operational -	Routing and material/information management	Routes of the mobile robots	 Real-time information sharing → sequencing of kitting and picking information 	 Real-time control of the capacity of the container 	 Real-time sharing and displaying of the information 			
		Control of the ALFS	Control of the system	Control of the system	Control of the system	Control of the system			

the environment where they work, AR may be the best solution. Moreover, the decisions that are made in regard to the types of containers and the types of warehouse in which these will be used may influence the kind of sensors that can be applied to them and the transportation devices that can transport them. In view of all the data that can be collected, companies also now need to decide on the level of automation of their AS. This because the different forms of data that can be collected require different sensors and different techniques to analyse them. Moreover, on the type of information generated by the data analysis, the selection of the most proper device is very important. This information can be as simple as just numbers or text, but can also be more complex, such as pictures or videos. This choice depends not only on the type of information to deliver but also on who will receive it and which tasks have to be performed. For example, in the case of large products, wearable devices will be more preferable than fixed ones more useful in the case of small products assembled in standard workstations. Furthermore, the main purpose of using such technology will also affect its implementation and hence the level of automation of the ASs. The selected technology will influence the skills that will be required to use it. The company then needs to know whether it already has someone with these skills or whether it will have to hire or train someone. In the case where the company can rely on internal resources with such skills, the technology selection will be facilitated, as well as their implementation. This will also increase the general technological knowledge of the whole company in the case where these resources share their expertise. Furthermore, even if a company is not planning to implement any new technology in its AS4.0, a strategic decision may be to hire skilled human workers, with previous experience in I4.0 projects. These workers are valuable resources that could assist and guide the company toward the possible implementation of I4.0 technologies.

Tactical: When the technologies have been chosen, the company needs to design the workplace in which they will be used and to determine how they will be used. Some forms of technology, for example VR, can help in designing the workplaces where other technologies will be used. Indeed, VR provides the possibility of creating virtual environments where human workers can interact with the virtual elements and also where the technologies, like cobots and mobile robots, can be modelled. This facilitates the study of different workplace designs. As a result of the virtual environment different alternatives between the human workers and the technologies can be studied before their implementation in real workplaces. This helps to create optimal workplaces where technologies and human workers can interact with optimal results. The feeding policies determined by the company can also influence the design of the workplace, since the transportation of a container, like a pallet, may require a different space when it is delivered to the workplace, compared to the transportation of a shelf. A time analysis and decision aid may be necessary at this level in order to identify the numbers of cobots, workers, AR devices, or mobile robots that are required to execute the different tasks in the AS. In conjunction with this time analysis, it may be necessary to evaluate the ergonomic impact of the technologies while they are being used, in order to understand where and when it is best to use them.

The time analysis and ergonomic impact can be evaluated due to the sensors applied in the technologies and by the human workers, such as IoT sensors or wearable devices. Another possibility is the combination of cameras that monitored the movements of the technologies and the human workers as a result of computer vision algorithms. At this level, a company is aware of which data it must collect and now needs to know where to install the sensors to collect these data. The sensors need to be installed in protected and proper places in order to reliably and safely collect the required data. Therefore, the significant amount of data that companies can collect from the sensors needs to be analysed and validated. This is because the next step is to decide what information the company is interested in creating from the collected data and how this information will be created. The final decision also relates to the issue of the final recipient of such information. It is important to create specific information for each user in order to avoid possible misunderstandings that can result in the execution of incorrect activities or decisions.

Operational: At this level, the technologies have been implemented and are working. The employees now need to know how to do their jobs, and the company needs to control them to ensure that they are doing them to the best of their ability. The daily activities that are assigned to cobots need to be sequenced based on the products that they are required to produce, and the routes of the mobile robots must be generated according to the paths that they need to take. The AR and VR equipment must also receive the information necessary for the human workers to carry out their activities. This information needs to be sequenced based on what activities the human workers will carry out when they are using the technology and it needs to be easily comprehensible and personalised. Indeed, there is a negative impact in the case of incorrect information, but also if the information is not understandable or clearly visible to the user. Therefore, a good control loop of accuracy and usability is important for continuously improving their utilisation. Moreover, the correct information, at the appropriate time and place, not only needs to be sent to the AR and VR devices, but also to all the actors in the AS4.0 that require it. This needs to be carried out efficiently whilst avoiding a possible loss of information during the communication process. Cloud computing can provide a way of sharing all this information, without the need for physical connections, across all of the devices in the AS that have the capacity to give them in output. All of this information can be generated using data analysis techniques and algorithms, and these need to be checked periodically to ensure that the parameters that they are using are still valid; alternatively, they may need


Figure 6. Future research opportunities at the different levels.

to be updated, for example due to variation in customer demand. These techniques and algorithms also need to be updated if new techniques or algorithms that are more efficient are developed, or if they fail to work properly. It is not only the software aspects of the technologies that need to be checked; the hardware parts of technologies such as mobile robots, cobots, AR and VR devices also need to be periodically maintained and updated in order to avoid problems during the execution of their tasks. For example, the malfunctioning of the batteries that power the mobile robots affects their performance in terms of speed and autonomy, thereby having a negative impact on the performance of the AS4.0. Checking the hardware of the technologies can help to avoid not only losses in the performance of the AS4.0 but also in its security. It is therefore clear that appropriate maintenance measures are important for strictly controlling the reliability and availability of the hardware of the technologies.

6. Future research opportunities

In this section, we explore possible future research into AS4.0, reported in Figure 6, to answer our third and final research question, RQ3. In order to achieve this, we use the knowledge acquired from the analysis presented in this work. This represents an attempt to inspire future researchers in the field of AS4.0 to address new research challenges that go beyond the study of the use of these technologies. We use the same classification as in Section 4, with future research opportunities divided based on strategic, tactical and operational levels. At the strategic level, the implementation of emerging technologies is investigated mainly through simulation. Indeed, simulation gives the opportunity to study, before the actual implementation, the benefits of the technologies under investigation, which ones performs better and how they perform and should be used. Given a positive outcome from simulation, the next step is to implement such technologies. The use of simulation allows for the investigation of a potentially infinite number of scenarios with a limited knowledge of the technology and its effects on the system. Consequently, decision makers, practitioners, engineers and managers can be made more aware of the behaviour of the system and its interaction with technologies. Moreover, it is mandatory to validate the simulation results, so some data collected from ad-hoc experiments or observations from previous applications are very helpful as input for the simulation, as well as validating its results.

At this point, at the tactical level, pilots are used to recreate real scenarios where technologies are under investigation mainly through experiments and in some cases with the observation of similar cases in order to tune their implementation by improving their performance. Simulation can still be used to study different situations starting from the data derived from the experiments. If the results of the experiments are positive, pilots are then extended and industrialised to be integrated with the rest of the facility. Finally, at the operational level, it is possible to study it through observations. These observations give the opportunity to obtain data to see if everything is working as it should be and to create simulations of possible future scenarios based on real data. In this case, experiments can still be used to investigate new ad-hoc solutions and so to feed simulation models with dedicated data. At the operational level, simulation is mainly used in order to exploit the potentialities of digital twins and so of predictive data-analytics, thereby gaining knowledge from data.

6.1. Strategic level

6.1.1. Research opportunities SL1: models of dynamic reconfigurable AS4.0

We observed that mobile robots and cobots were the two of the most frequently studied technologies. One aspect identified in our analysis that was not deeply investigated was related to the high level of reconfigurability that can be enabled from these two technologies, and especially from the use of mobile robots. In a reconfigurable system all the equipment, human workers, or material handling systems should be rapidly added, removed, modified or interchange in response to changing needs and opportunities. Due to their flexibility, mobile robots can facilitate companies in creating AS4.0 configurations (primarily networks rather than line configurations) that change dynamically based on the current demand. In fact, using mobile robots as a workplace in which human workers perform the assembly tasks, can give the opportunity to increase or decrease the number of workplaces that are needed based on the demand and the same mobile robots can also be used to transport all the components that the human workers need for each task (Battaïa et al. 2018). Moreover, a mobile robot can be equipped with a cobot to create a collaborative mobile robot that can pick and transport components and then execute assembly tasks based on the same components. Although these approaches can be easily adopted to create different AS4.0 configurations, it is essential to understand which is the best configuration, in terms of performance and cost, based on the situation under study. Future studies should therefore focus on mathematical and simulation modelling of the different configurations created by these two technologies. These models will allow us to determine the parameters that can affect the reconfigurability of the system. Sensitivity analyses will also be useful to study how these parameters can change the reconfigurability

and the performance of the different configurations. In the end, decision support systems, like decision trees and evolutionary algorithms, can be created in order to support decision makers in determining which is the best configuration to adopt based on their needs.

6.1.2. Research opportunities SL2: models to support the selection of a suitable level of automation

I4.0 technologies can be used to execute a task or to collect and share data, for example to assist operators or to support managers. Hence, the choice of a technology is not only related to the help that it can give in terms of the physical and cognitive aspects of the execution of a task, but also to the data that can be collected and shared with it. A huge amount of data can be collected and sent to the AS4.0, and new models are needed to support decision makers in regard to how to automatise the AS4.0. These models can provide a methodological toolbox that can guarantee a structured implementation of the technologies to ensure a suitable level of automation for AS4.0. For example, decisions that need to be made on the most suitable amount of information to give to the operator through AR or assistive technologies (which relates to the level of collaboration between operators and collaborative robots), or on the configuration of the systems, can be made by managers based on the available data. To create these models, further research is required specifically in order to deepen our existing knowledge of the case where multiple technologies work together to complete different tasks. Simulations and experiments are needed in order to validate these models. The results from these methods, together with the possibility to do sensitive analysis, give the opportunity to evaluate different cases with advanced data analytics tools that in the end can create decisional support systems. Companies can benefit from these results since they will help them simplify the decision process of the level of automation to adopt in their AS4.0.

6.1.3. Research opportunities SL3: models for dynamic assignment of technologies on AS4.0 configurations

We have seen that at this level, companies have to choose the configuration and the level of automation of their AS4.0. It can be that the choice of a configuration compromises the possibility of using a specific technology and instead that technology is most appropriate for another kind of configuration. Moreover, the performance of the AS4.0 is correlated to the combination of configurations and technologies, with synergic effects in some cases. Therefore, at this level, to determine the goals of the AS4.0 that the companies want to obtain, it is necessary to have a guide that supports decision makers in

16 🛞 A. DOLGUI ET AL.

the correct assignment of the different technologies to the configurations under analysis. The decision of the assignment of the technologies on the different configurations will facilitate other decisions, such as, for example, the number of devices to buy and the skills that are necessary in the different configurations. In order to understand if a technology is good or not for a specific configuration, future research should consider its behaviour and performance in different configurations. Case studies, experiments and simulations represent opportunities to collect the necessary data to create new decisional support systems to guide companies in better deciding which configurations to implement for the different technologies. It will also be important to study the behaviour and performance of more technologies together and the interaction between the technologies and the human workers to see if their implementation together generates issues in the configuration or not.

6.2. Tactical level

6.2.1. Research opportunities TL1: methods for ergo-efficient workplace design

When a human worker is involved in the execution of activities in the AS4.0, he or she should be able to do these without worrying about the workplace in which these activities need to take place. To enable this, the workplace needs to be optimised, and VR can be used in this instance to help create virtual environments that replicate the real AS, to allow a user to design and study a range of configurations of workplaces without needing a physical model. These virtual solutions need to be further investigated, and new parameters such as the ages of the human workers or the integration of additional technologies into virtual environments should be evaluated. Different human characteristics may require different workplace design solutions in order to facilitate high levels of ergonomic comfort for manual processes and to increase efficiency. Data collected from human workers while they are executing their activities can be used to adapt the workplace based on the tasks that need to be executed within the workplace and the characteristics of the human worker that is executing them. Motion capture (mocap) systems can be used to collect these data. A mocap system gives the opportunity to create a virtual copy of the object that is under study by utilising sensors or cameras to recreate its movements. Experiments are first needed in order to determine the relevant parameters for optimising a workstation, and case studies will then be required to verify whether these parameters are correct, in order to design a workplace that is not just optimal in terms of performance but is also efficient in terms of human well-being. In relation to the acceptance

of technology, further studies could provide insight into the factors affecting success. Data analysis and Artificial Intelligence techniques can be useful at this step.

6.2.2. Research opportunities TL2: models of new feeding policies

The use of mobile robots, with or without a cobot, can open up the possibility of creating new feeding policies, since if these are used to move shelves, each level of the shelves can be designed to store different components in different ways. For example, although a kit can be stored at one level, while only one type of component can be stored at another level, the shelf may be delivered to one or more workplaces. Instead, if mobile robots are used in conjunction with a cobot, these collaborative mobile robots can be used to pick the individual components from warehouses or supermarkets and to transport them to the ALSs that need them. In order to study these new opportunities, advanced modelling and simulations and then real case studies will be needed, to validate the results of the models. Moreover, AR can be used to support the human workers involved in feeding tasks by giving them instructions such as which component to pick, where to put the components, and how many components to pick. When AR is used together with mobile robots, the two technologies need to be integrated. This means that the AR instructions should not only able to change based on the demand for products but also based on the characteristics of the mobile robots that are used to execute the feeding policies. In addition, these AR instructions need to change based on the components that the human workers have to pick and must be visible and in the correct position for the particular type of mobile robot that is used. Further investigation and testing through experiments that replicate real scenarios can better highlight the strengths, weaknesses, and the potential advantages that the use of AR with mobile robots can generate in terms of developing new feeding policies.

6.2.3. Research opportunities TL3: models for real-time multi-objective balancing of ALS and scheduling of ALFS

The balancing of the new ALS needs to take into account not only the assignment of the different tasks within the different workplaces, but also the technologies that will be used in the ALS. However, no research has yet studied how these technologies can affect line balancing. For example, the choice to adopt AR to support operators in executing the tasks is not yet well-investigated and in particular the impact on the balancing of the ALS. In any case, the use of AR is very interesting since it can help to reduce line balancing effort for the ALS by creating flexible, dynamic, self-balancing ALSs in which the

number of workplaces may change based on the current demand from customers. Hence, if demand is low, a single operator may be able to do all the tasks in one workplace, thanks to the use of AR, whereas if demand grows, the number of workplaces can increase, and the AR instructions will need to be divided between the workstations. One avenue for future research would be to investigate the number of AR devices that are needed and the ways in which the instructions that they display to the human workers need to change. The various types of mobile robots, which may or may not have a cobot mounted on them, allow for new tasks to be carried out in the ALS and in the ALFS. For example, we saw that a mobile robot with a cobot can pick up a component from the warehouse, transport it to the ALFS, and then execute a task based on this component in the ALS. The new dynamic network configurations that have become possible require new assembly line balancing methods that consider the evolutionary aspects of such configurations, with multi-period and multi-objective models that can be applied in real time to adapt the workload of the workstations without affecting the work of the operators. Economic, performance, and ergonomic objectives should be considered together.

A mobile robot can be used solely as a workplace or as a transportation device. Different mobile robots can carry out different tasks, giving rise not only to new opportunities but also to a more complex problem in relation to scheduling these mobile robots. A company needs to optimise the use of its mobile robots according to the tasks that they are able to do, and new scheduling models therefore need to be created that can consider all the possible tasks that these mobile robots can carry out. These models first need to be studied through simulations, and then with experiments or case studies in order to verify the results. A connection between the different actors in an AS4.0, thanks to all the sensors and hence the collected data, can allow for synchronised assembly line balancing and scheduling in order to simultaneously optimise both the assembly and the activities needed to deliver the components. Future investigations into methodologies for dynamic and synchronised assembly line balancing and scheduling of the AS4.0 are needed in order to allow companies to use their technologies in an optimal way.

6.3. Operational level

6.3.1. Research opportunities OL1: smart and real-time methods for sequencing of ALS and routeing of ALFS

ALSs need to know the sequence of products that they have to assemble on a daily basis, and mobile robots need

to know the routes to follow to arrive at their destinations. Algorithms developed in the future to solve these problems will need to take into consideration the new information generated by the AS4.0 in a predictive and prescriptive way, which will involve not only reacting to the changes required but also anticipating them and adapting in a proactive way. For example, a new sequencing algorithm may consider the fatigue of the human workers when deciding which products to assemble or which workers should execute the tasks to make them. If the human workers are tired, it may be possible to sequence products that are easy to assemble, or to let the workers take a break to recover their energy. Sequencing can be done at the same time as routeing when the products to be assembled are determined with the goal of minimising the routes travelled by the mobile robots. Sequencing should also take into account the quantity of components stored in the warehouses and supermarkets, i.e. to optimise inventory management by reducing stocked quantities by as much as possible. Sequencing and routeing can be synchronised to allow products in an ALS to be assembled using only the components that are closest to that ALS. To achieve this, the routeing of the mobile robots must consider their positions in real time; only the mobile robots closest to the components will be used to transport them to the ALS. This synchronised sequencing and routeing can reduce the travelling time of the mobile robots and can also increase the productivity of the ALS, since the components are delivered more quickly.

6.3.2. Research opportunities OL2: methods for efficient control of AS4.0

Today, it has become possible for the first time to control the quality of the products, the assembly activity, and all the activities of an AS4.0. This is facilitated by the huge amounts of data that can be collected from an AS4.0 if its activities are monitored, including data from each step of the assembly tasks, from the products being assembled, from the mobile robots while they are working and so on. After collection, these data need to be understood and analysed, in order to allow companies to use them. Data analysis techniques can be applied to these data to give useful information as output. New machine learning and artificial intelligence algorithms that transform data into information can be developed by studying the different activities of the AS4.0. Simulations will be useful in order to understand the performance of the methods, and experiments with real scenarios should then be carried out. It is important that the right data are collected to ensure the success of these methods and to give valid results, and more research is needed to identify the factors that make a data analysis technique reliable.

Surveys of practitioners can help in developing an understanding of the factors that can increase reliability. More reliable are the results of a data analysis technique and more precise can be the control of the AS4.0. At the same time higher can be the benefits that can be generate from these analyses. In the material management activity, for example, the adoption of data analysis techniques can open the opportunity to optimise the stock of materials in the warehouses and workplaces, and at the same time the flows of these materials in the AS4.0 in a dynamic and real-time way, such as dynamic Kanban systems, dynamic replenishment, integrated replenishment policies. New objective functions, for a dynamic and synchronised stock and flow of the material, can be defined in order to improve the performance of material management in an AS4.0. Benefitting from all these data and information, it is possible to model all, or parts of, the AS4.0 in order to create its digital twin. The digital twin of the AS4.0 gives the opportunity to control in real-time what is happening in the AS4.0 and to simulate what can happen in the future using predictive analysis models. Indeed, it allows us to see, for example, how mobile robots work in a single day and then simulate their performance for an entire week. This simulation can be useful for understanding whether the mobile robots will decrease their performance or not after a period of continuous work. Although digital twins have significant potential, it is important to understand what it is relevant to monitor. Therefore, future studies can be orientated towards understanding which digital copies are most important to create based on the characteristics of the AS4.0. Case studies, experiments and interviews with the practitioners can help to create frameworks to follow to create the digital copies whilst simultaneously verifying their reliability.

6.3.3. Research opportunities OL3: models and methods for maintenance of technologies

As discussed in the previous section, both the software and hardware aspects of these technologies need to be controlled and updated. All of the sensors in the AS4.0, the mobile robots, and the AR devices must operate without problems over their lifetimes. For example, the sensors must collect all of the appropriate data without missing any, and the mobile robots must run at a certain speed to execute their tasks. The development of new models and methods for the maintenance of these technologies, both predictive and proactive, should therefore be a subject for future study. These models and methods can help to avoid a loss of performance of the AS4.0 due to hardware problems with the technologies. It is important that these methods recognise when the performance of a particular technology is reduced, even if problems are not evident, and when it must be replaced by a new model. Technology must also be replaced when it becomes obsolete. New data analysis techniques should be created, creating predictive analytics models, to predict when both of these situations are likely to occur. The information created by these techniques can then be sent to the technologies that are recognised as obsolete or that are not working properly to make them stop working. Once they stop, they can be evaluated in order to understand whether to fix or dispose of them.

7. Conclusion

The systematic literature review presented here reveals that despite the benefits that I4.0 technologies can bring to ASs, it is necessary to carry out more research in order to exploit their full potential. Researchers should not limit themselves to studies of how a technology works. Although these are important, they are not sufficient, since this technology will be implemented in a complex environment such as an AS. In an AS, each technology does not operate alone but is integrated into a process that involves several actors, such as human workers, machinery, and other technologies. Each actor has certain tasks to complete, which may require cooperation from two or more other actors. Moreover, companies need to know which decisions they must make when they want to plan to use a technology (strategic level), when they have to decide which technology to use (tactical level), and when they are using this technology (operational level). For these reasons, we first reviewed how these different technologies were used in ASs as part of their activities, and then proposed a future research agenda that can be followed in order to start to reduce the gaps highlighted here. In particular, have seen that the creation of the new AS4.0 requires the study of new advanced solutions at three different levels (strategic, tactical and operational). At the strategic level, one of the most interesting challenges in the AS4.0 is to understand which technologies to use in order to create reconfigurable AS4.0s. It is important to understand first what is required to create a reconfigurable system and then how the technologies can help in developing one. Mathematical optimisation models can be created and then simulated to generate multi-case scenarios that can help the development of decision support systems to help to choose the configuration of the AS4.0 and which technologies to use in it. At the tactical level, the AS4.0 is characterised by the multi-purpose use of the technologies. Therefore, at this level, the challenges are related with the creation of models and methods that can solve multi-objective problems (time, ergonomics, costs, energy and others)

that can involve the design of the AS4.0 and their feeding, balancing and scheduling. At the operational level, the major challenges that we individuated are related to the daily control of the technologies in order to make them work and to keep them working with the same efficiency in order to do not compromise the performance of the AS4.0. Therefore, to make this happen, it is necessary to create data-driven models and methods to control and maintain the technologies and the AS4.0. Moreover, the development of predictive analytics models based on the huge amount of data available from sensors that can forecast the behaviour of the technologies and the performances of the AS4.0 can be an interesting challenge for future studies. We hope that researchers who are already studying or intend to study this topic can find useful ideas in this work and will be inspired to create new ones.

DAS statement

'The authors confirm that the data supporting the findings of this study are available within the article'.

Disclosure statement

No potential conflict of interest was reported by the author(s).

Notes on contributors



Marco Simonetto is a PhD candidate at the Department of Mechanical and Industrial Engineering (MTP) at NTNU (Norway). He has a master's degree in Management Engineering from University of Padova (Italy). His research interests include assembly systems, Industry 4.0, and material feeding.



Fabio Sgarbossa is Full Professor of Industrial Logistics at the Department of Mechanical and Industrial Engineering (MTP) at NTNU (Norway) from October 2018. He was Associate Professor at University of Padova (Italy) where he also received his PhD in Industrial Engineering in 2010. He is leader of the Production Management

Group at MTP, and he is responsible of the logistics 4.0 Lab at NTNU. He has been and he is involved in several European and National Projects. He is author and co-author of about 120 publications in relevant international journals, about industrial logistics, material handling, materials management, supply chain. He is member of Organizing and Scientific Committees of several International Conferences, and he is member of editorial boards and associate editor in relevant International Journals.



Alexandre Dolgui is an IISE Fellow, Distinguished Professor, and the Head of Automation, Production and Computer Sciences Department at the IMT Atlantique, France. His research focuses on manufacturing line design, production planning and supply chain optimisation. His main results are based on the exact

mathematical programming methods and their intelligent coupling with heuristics and metaheuristics algorithms. He is author or co-author of about 270 publications in refereed international journals, and has received several international research awards. He has been Program or General chair of many leading international conferences. He is Member of the Editorial Board of many journals, including the International Journal of Production Economics. He is the Editor-in-Chief of the International Journal of Production Research.

ORCID

Alexandre Dolgui D http://orcid.org/0000-0003-0527-4716 Fabio Sgarbossa D http://orcid.org/0000-0002-9541-3515

References

- Agrawal, N., and M. A. Cohen. 2001. "Optimal Material Control in an Assembly System with Component Commonality." *Naval Research Logistics (NRL)* 48 (5): 409–429.
- Akpinar, S., A. Elmi, and T. Bektaş. 2017. "Combinatorial Benders Cuts for Assembly Line Balancing Problems with Setups." *European Journal of Operational Research* 259 (2): 527–537.
- Al-Dahhan, M. R. H., and K. W. Schmidt. 2020. "Voronoi Boundary Visibility for Efficient Path Planning." *IEEE Access* 8: 134764–134781.
- Alavian, P., Y. Eun, S. M. Meerkov, and L. Zhang. 2020. "Smart Production Systems: Automating Decision-Making in Manufacturing Environment." *International Journal of Production Research* 58 (3): 828–845.
- Alexopoulos, K., S. Makris, V. Xanthakis, K. Sipsas, and G. Chryssolouris. 2016. "A Concept for Context-Aware Computing in Manufacturing: The White Goods Case." *International Journal of Computer Integrated Manufacturing* 29 (8): 839–849.
- Alexopoulos, K., K. Sipsas, E. Xanthakis, S. Makris, and D. Mourtzis. 2018. "An Industrial Internet of Things Based Platform for Context-Aware Information Services in Manufacturing." *International Journal of Computer Integrated Manufacturing* 31 (11): 1111–1123.
- Andersen, R. E., E. B. Hansen, D. Cerny, S. Madsen, B. Pulendralingam, S. Bøgh, and D. Chrysostomou. 2017. "Integration of a Skill-Based Collaborative Mobile Robot in a Smart Cyber-Physical Environment." *Procedia Manufacturing* 11: 114–123.
- Antonelo, E. A., B. Schrauwen, and J. Van Campenhout. 2007. "Generative Modeling of Autonomous Robots and Their Environments Using Reservoir Computing." *Neural Processing Letters* 26 (3): 233–249.
- Arena, S., F. Doneddu, and M. T. Pilloni. 2019. "Optimization of Brewing and Beverage Packaging Process using V-Graph Analysis." XXIV Summer School "Francesco Turco"-Industrial Systems Engineering 1: 307–313.

- Armbrust, M., A. Fox, R. Griffith, A. D. Joseph, R. Katz, A. Konwinski, G. Lee, et al. 2010. "A View of Cloud Computing." *Communications of the ACM* 53 (4): 50–58.
- Armesto, L., G. Ippoliti, S. Longhi, and J. Tornero. 2008. "Probabilistic Self-Localization and Mapping-an Asynchronous Multirate Approach." *IEEE Robotics & Automation Magazine* 15 (2): 77–88.
- Azizi, A., P. G. Yazdi, and M. Hashemipour. 2019. "Interactive Design of Storage Unit Utilizing Virtual Reality and Ergonomic Framework for Production Optimization in Manufacturing Industry." *International Journal on Interactive Design and Manufacturing (IJIDeM)*) 13 (1): 373–381.
- Bao, Q., G. Zhao, Y. Yu, S. Dai, and W. Wang. 2020. "Ontology-Based Modeling of Part Digital Twin Oriented to Assembly." *Proceedings of the Institution of Mechanical Engineers, Part B: Journal of Engineering Manufacture.* doi:10.1177/0954405420941160.
- Bassan, Y., Y. Roll, and M. J. Rosenblatt. 1980. "Internal Layout Design of a Warehouse." AIIE Transactions 12 (4): 317–322.
- Battaïa, O., and A. Dolgui. 2013. "A Taxonomy of Line Balancing Problems and Their Solution Approaches." *International Journal of Production Economics* 142 (2): 259–277.
- Battaïa, O., A. Otto, F. Sgarbossa, and E. Pesch. 2018. "Future Trends in Management and Operation of Assembly Systems: From Customized Assembly Systems to Cyber-Physical Systems." Omega 78: 1–4.
- Battini, D., M. Faccio, A. Persona, and F. Sgarbossa. 2009a. "Balancing–Sequencing Procedure for a Mixed Model Assembly System in Case of Finite Buffer Capacity." *The International Journal of Advanced Manufacturing Technology* 44 (3–4): 345–359.
- Battini, D., M. Faccio, A. Persona, and F. Sgarbossa. 2009b. "Design of the Optimal Feeding Policy in an Assembly System." *International Journal of Production Economics* 121 (1): 233–254.
- Battini, D., M. Faccio, A. Persona, and F. Sgarbossa. 2011. "New Methodological Framework to Improve Productivity and Ergonomics in Assembly System Design." *International Journal of Industrial Ergonomics* 41 (1): 30–42.
- Battini, D., M. Gamberi, A. Persona, and F. Sgarbossa. 2015. "Part-Feeding with Supermarket in Assembly Systems: Transportation Mode Selection Model and Multi-Scenario Analysis." Assembly Automation 35: 149–159.
- Baumann, D., F. Mager, U. Wetzker, L. Thiele, M. Zimmerling, and S. Trimpe. 2020. "Wireless Control for Smart Manufacturing: Recent Approaches and Open Challenges." *Proceed*ings of the IEEE 109: 441–467.
- Bauters, K., J. Cottyn, D. Claeys, M. Slembrouck, P. Veelaert, and H. Van Landeghem. 2018. "Automated Work Cycle Classification and Performance Measurement for Manual Work Stations." *Robotics and Computer-Integrated Manufacturing* 51: 139–157.
- Becker, C., and A. Scholl. 2006. "A Survey on Problems and Methods in Generalized Assembly Line Balancing." European Journal of Operational Research 168 (3): 694–715.
- Bencherif, A., and F. Chouireb. 2019. "A Recurrent TSK Interval Type-2 Fuzzy Neural Networks Control with Online Structure and Parameter Learning for Mobile Robot Trajectory Tracking." *Applied Intelligence* 49 (11): 3881–3893.
- Bilberg, A., and A. A. Malik. 2019. "Digital Twin Driven Human-Robot Collaborative Assembly." *CIRP Annals* 68 (1): 499–502.

- Blecker, T., and N. Abdelkafi. 2006. "Mass Customization: State-of-the-Art and Challenges." In Mass Customization: Challenges and Solutions. Vol. 87, 1–25. Boston, MA: Springer.
- Bocewicz, G., I. Nielsen, and Z. Banaszak. 2014. "Automated Guided Vehicles Fleet Match-Up Scheduling with Production Flow Constraints." *Engineering Applications of Artificial Intelligence* 30: 49–62.
- Bortolini, M., M. Faccio, M. Gamberi, and F. Pilati. 2020. "Motion Analysis System (MAS) for Production and Ergonomics Assessment in the Manufacturing Processes." *Computers & Industrial Engineering* 139: 105485.
- Bortolini, M., E. Ferrari, M. Gamberi, F. Pilati, and M. Faccio. 2017. "Assembly System Design in the Industry 4.0 Era: A General Framework." *IFAC-PapersOnLine* 50 (1): 5700–5705.
- Boysen, N., D. Briskorn, and S. Emde. 2017. "Parts-to-Picker Based Order Processing in a Rack-Moving Mobile Robots Environment." *European Journal of Operational Research* 262 (2): 550–562.
- Boysen, N., M. Fliedner, and A. Scholl. 2008. "Assembly Line Balancing: Which Model to Use When?" *International Jour*nal of Production Economics 111 (2): 509–528.
- Boysen, N., M. Kiel, and A. Scholl. 2011. "Sequencing Mixed-Model Assembly Lines to Minimise the Number of Work Overload Situations." *International Journal of Production Research* 49 (16): 4735–4760.
- Bruno, G., and D. Antonelli. 2018. "Dynamic Task Classification and Assignment for the Management of Human-Robot Collaborative Teams in Workcells." *The International Journal of Advanced Manufacturing Technology* 98 (9): 2415–2427.
- Caputo, A. C., and P. M. Pelagagge. 2011. "A Methodology for Selecting Assembly Systems Feeding Policy." Industrial Management & Data Systems 111: 84–112.
- Caputo, A. C., P. M. Pelagagge, and P. Salini. 2018. "Selection of Assembly Lines Feeding Policies Based on Parts Features and Scenario Conditions." *International Journal of Production Research* 56 (3): 1208–1232.
- Caridá, V. F., O. Morandin, and C. C. M. Tuma. 2015. "Approaches of Fuzzy Systems Applied to an AGV Dispatching System in a FMS." *The International Journal of Advanced Manufacturing Technology* 79 (1–4): 615–625.
- Carmigniani, J., B. Furht, M. Anisetti, P. Ceravolo, E. Damiani, and M. Ivkovic. 2011. "Augmented Reality Technologies, Systems and Applications." *Multimedia Tools and Applications* 51 (1): 341–377.
- Carvajal Soto, J. A., F. Tavakolizadeh, and D. Gyulai. 2019. "An Online Machine Learning Framework for Early Detection of Product Failures in an Industry 4.0 Context." *International Journal of Computer Integrated Manufacturing* 32 (4–5): 452–465.
- Chang, L., L. Shan, C. Jiang, and Y. Dai. 2020. "Reinforcement Based Mobile Robot Path Planning with Improved Dynamic Window Approach in Unknown Environment." *Autonomous Robots*, 1–26.
- Chen, B., Y. Liu, C. Zhang, and Z. Wang. 2020a. "Time Series Data for Equipment Reliability Analysis with Deep Learning." *IEEE Access* 8: 105484–105493.
- Chen, C., Z. Tian, D. Li, L. Pang, T. Wang, and J. Hong. 2020b. "Projection-based Augmented Reality System for Assembly Guidance and Monitoring." *Assembly Automation*.

- Chen, C., C. Zhang, T. Wang, D. Li, Y. Guo, Z. Zhao, and J. Hong. 2020c. "Monitoring of Assembly Process Using Deep Learning Technology." *Sensors* 20 (15): 4208.
- Chien, C. H., W. Y. Wang, and C. C. Hsu. 2017. "Multi-Objective Evolutionary Approach to Prevent Premature Convergence in Monte Carlo Localization." *Applied Soft Computing* 50: 260–279.
- Choobineh, F. F., A. Asef-Vaziri, and X. Huang. 2012. "Fleet Sizing of Automated Guided Vehicles: A Linear Programming Approach Based on Closed Queuing Networks." *International Journal of Production Research* 50 (12): 3222–3235.
- Çil, Z. A., Z. Li, S. Mete, and E. Özceylan. 2020. "Mathematical Model and Bee Algorithms for Mixed-Model Assembly Line Balancing Problem with Physical Human–Robot Collaboration." *Applied Soft Computing* 93: 106394.
- Cimino, C., E. Negri, and L. Fumagalli. 2019. "Review of Digital Twin Applications in Manufacturing." *Computers in Industry* 113: 103130.
- Cohen, Y., M. Faccio, F. Pilati, and X. Yao. 2019a. "Design and Management of Digital Manufacturing and Assembly Systems in the Industry 4.0 Era." *International Journal of Advanced Manufacturing Technolology* 105 (9): 3565–3577.
- Cohen, Y., H. Naseraldin, A. Chaudhuri, and F. Pilati. 2019b. "Assembly Systems in Industry 4.0 Era: A Road Map to Understand Assembly 4.0." *The International Journal of Advanced Manufacturing Technology* 105 (9): 4037–4054.
- Colledani, M., D. Coupek, A. Verl, J. Aichele, and A. Yemane. 2018. "A Cyber-Physical System for Quality-Oriented Assembly of Automotive Electric Motors." CIRP Journal of Manufacturing Science and Technology 20: 12–22.
- Cormier, G., and E. A. Gunn. 1992. "A Review of Warehouse Models." European Journal of Operational Research 58 (1): 3–13.
- Culot, G., G. Nassimbeni, G. Orzes, and M. Sartor. 2020. "Behind the Definition of Industry 4.0: Analysis and Open Questions." *International Journal of Production Economics* 226: 107617.
- Dalle Mura, M., G. Dini. 2019. "Designing Assembly Lines with Humans and Collaborative Robots: A Genetic Approach." *CIRP Annals* 68 (1): 1–4.
- Dang, Q. V., C. T. Nguyen, and H. Rudová. 2019. "Scheduling of Mobile Robots for Transportation and Manufacturing Tasks." *Journal of Heuristics* 25 (2): 175–213.
- Dang, Q. V., I. Nielsen, K. Steger-Jensen, and O. Madsen. 2014. "Scheduling a Single Mobile Robot for Part-Feeding Tasks of Production Lines." *Journal of Intelligent Manufacturing* 25 (6): 1271–1287.
- Danielsson, O., M. Holm, and A. Syberfeldt. 2020. "Augmented Reality Smart Glasses in Industrial Assembly: Current Status and Future Challenges." *Journal of Industrial Information Integration* 20: 100175.
- Das, P. K., and P. K. Jena. 2020. "Multi-Robot Path Planning Using Improved Particle Swarm Optimization Algorithm Through Novel Evolutionary Operators." *Applied Soft Computing* 92: 106312.
- Da Silveira, G., D. Borenstein, and F. S. Fogliatto. 2001. "Mass Customization: Literature Review and Research Directions." International Journal of Production Economics 72 (1): 1–13.
- DeCroix, G. A., and P. H. Zipkin. 2005. "Inventory Management for an Assembly System with Product or Component Returns." *Management Science* 51 (8): 1250–1265.

- Delgado-López-Cózar, E., and Á Cabezas-Clavijo. 2013. "Ranking Journals: Could Google Scholar Metrics be an Alternative to Journal Citation Reports and Scimago Journal Rank?" *Learned Publishing* 26 (2): 101–114.
- Djuric, A. M., R. J. Urbanic, and J. L. Rickli. 2016. "A Framework for Collaborative Robot (CoBot) Integration in Advanced Manufacturing Systems." SAE International Journal of Materials and Manufacturing 9 (2): 457–464.
- Dolgui, A., S. E. Hashemi-Petroodi, S. Kovalev, and M. Y. Kovalyov. 2021. "Profitability of a Multi-Model Manufacturing Line Versus Multiple Dedicated Lines." *International Journal* of Production Economics 236: 108113.
- Dolgui, A., S. Kovalev, M. Kovalyov, S. Malyutin, and A. Soukhal. 2018. "Optimal Workforce Assignment to Opertions of a Paced Assembly Line." *European Journal of Operational Research* 264 (1): 200–211.
- Dolgui, A., and J.-M. Proth. 2010. Supply Chain Engineering: Useful Methods and Techniques. London: Springer.
- Downe-Wamboldt, B. 1992. "Content Analysis: Method, Applications, and Issues." *Health Care for Women International* 13 (3): 313–321.
- Elhoseny, M., A. Tharwat, and A. E. Hassanien. 2018. "Bezier Curve Based Path Planning in a Dynamic Field Using Modified Genetic Algorithm." *Journal of Computational Science* 25: 339–350.
- El Makrini, I., K. Merckaert, J. De Winter, D. Lefeber, and B. Vanderborght. 2019. "Task Allocation for Improved Ergonomics in Human-Robot Collaborative Assembly." *Interaction Studies* 20 (1): 102–133.
- Enomoto, A., N. Yamamoto, and T. Suzuki. 2013. "Automatic Estimation of the Ergonomics Parameters of Assembly Operations." *CIRP Annals* 62 (1): 13–16.
- Faccio, M., E. Ferrari, M. Gamberi, and F. Pilati. 2019. "Human Factor Analyser for Work Measurement of Manual Manufacturing and Assembly Processes." *The International Journal of Advanced Manufacturing Technology* 103 (1–4): 861– 877.
- Faccio, M., R. Minto, G. Rosati, and M. Bottin. 2020. "The Influence of the Product Characteristics on Human-Robot Collaboration: A Model for the Performance of Collaborative Robotic Assembly." The International Journal of Advanced Manufacturing Technology 106 (5): 2317–2331.
- Fager, P., M. Calzavara, and F. Sgarbossa. 2020. "Modelling Time Efficiency of Cobot-Supported kit Preparation." *The International Journal of Advanced Manufacturing Technology* 106 (5): 2227–2241.
- Fager, P., R. Hanson, L. Medbo, and M. I. Johansson. 2019. "Kit Preparation for Mixed Model Assembly–Efficiency Impact of the Picking Information System." *Computers & Industrial Engineering* 129: 169–178.
- Falagas, M. E., V. D. Kouranos, R. Arencibia-Jorge, and D. E. Karageorgopoulos. 2008. "Comparison of SCImago Journal Rank Indicator with Journal Impact Factor." *The FASEB Journal* 22 (8): 2623–2628.
- Fang, W., and Z. An. 2020. "A Scalable Wearable AR System for Manual Order Picking Based on Warehouse Floor-Related Navigation." *The International Journal of Advanced Manufacturing Technology* 109 (7): 2023–2037.
- Fantoni, G., S. Q. Al-Zubaidi, E. Coli, and D. Mazzei. 2020. "Automating the Process of Method-Time-Measurement." *International Journal of Productivity and Performance Man*agement 70 (4): 958–982.

22 🛞 A. DOLGUI ET AL.

- Filotheou, A., E. Tsardoulias, A. Dimitriou, A. Symeonidis, and L. Petrou. 2020. "Pose Selection and Feedback Methods in Tandem Combinations of Particle Filters with Scan-Matching for 2D Mobile Robot Localisation." Journal of Intelligent & Robotic Systems 100 (3): 925–944.
- Fragapane, G., R. de Koster, F. Sgarbossa, and J. O. Strandhagen. 2021. "Planning and Control of Autonomous Mobile Robots for Intralogistics: Literature Review and Research Agenda." *European Journal of Operational Research* 294 (2): 405–426.
- Fragapane, G., D. Ivanov, M. Peron, F. Sgarbossa, and J. O. Strandhagen. 2020. "Increasing Flexibility and Productivity in Industry 4.0 Production Networks with Autonomous Mobile Robots and Smart Intralogistics." Annals of Operations Research, 1–19. doi:10.1007/s10479-020-03526-7.
- Galambos, P., Á Csapó, P. Zentay, I. M. Fülöp, T. Haidegger, P. Baranyi, and I. J. Rudas. 2015. "Design, Programming and Orchestration of Heterogeneous Manufacturing Systems Through VR-Powered Remote Collaboration." *Robotics and Computer-Integrated Manufacturing* 33: 68–77.
- Gao, Y., J. Liu, M. Q. Hu, H. Xu, K. P. Li, and H. Hu. 2019.
 "A New Path Evaluation Method for Path Planning With Localizability." *IEEE Access* 7: 162583–162597.
- Gao, W., X. D. Shao, and H. L. Liu. 2016. "Enhancing Fidelity of Virtual Assembly by Considering Human Factors." *The International Journal of Advanced Manufacturing Technology* 83 (5–8): 873–886.
- Gebser, M., P. Obermeier, T. Schaub, M. Ratsch-Heitmann, and M. Runge. 2018. "Routing Driverless Transport Vehicles in car Assembly with Answer set Programming." *Theory and Practice of Logic Programming* 18 (3–4): 520–534.
- Gharehgozli, A., and N. Zaerpour. 2020. "Robot Scheduling for Pod Retrieval in a Robotic Mobile Fulfillment System." *Transportation Research Part E: Logistics and Transportation Review* 142: 102087.
- Groß, R., and M. Dorigo. 2009. "Towards Group Transport by Swarms of Robots." *International Journal of Bio-Inspired Computation* 1 (1–2): 1–13.
- Gualtieri, L., I. Palomba, F. A. Merati, E. Rauch, and R. Vidoni. 2020. "Design of Human-Centered Collaborative Assembly Workstations for the Improvement of Operators' Physical Ergonomics and Production Efficiency: A Case Study." Sustainability 12 (9): 3606.
- Guo, D., R. Y. Zhong, P. Lin, Z. Lyu, Y. Rong, and G. Q. Huang. 2020b. "Digital Twin-Enabled Graduation Intelligent Manufacturing System for Fixed-Position Assembly Islands." *Robotics and Computer-Integrated Manufacturing* 63: 101917.
- Guo, D., R. Y. Zhong, S. Ling, Y. Rong, and G. Q. Huang. 2020a. "A Roadmap for Assembly 4.0: Self-Configuration of Fixed-Position Assembly Islands Under Graduation Intelligent Manufacturing System." *International Journal of Production Research* 58 (15): 4631–4646.
- Halawa, F., H. Dauod, I. G. Lee, Y. Li, S. W. Yoon, and S. H. Chung. 2020. "Introduction of a Real Time Location System to Enhance the Warehouse Safety and Operational Efficiency." *International Journal of Production Economics* 224: 107541.
- Hari, S. K. K., A. Nayak, and S. Rathinam. 2020. "An Approximation Algorithm for a Task Allocation, Sequencing and Scheduling Problem Involving a Human-Robot Team." *IEEE Robotics and Automation Letters* 5 (2): 2146–2153.

- Hashemi-Petroodi, S. E., A. Dolgui, S. Kovalev, M. Kovalyov, and S. Thevenin. 2020a. "Workforce Reconfiguration Strategies in Manufacturing Systems: A State of the art." *International Journal of Production Research*, doi:10.1080 00207543.2020.1823028.
- Hashemi-Petroodi, S. E., S. Thevenin, S. Kovalev, and A. Dolgui. 2020b. "Operations Management Issues in Design and Control of Hybrid Human-Robot Collaborative Manufacturing Systems: A Survey." *Annual Reviews in Control* 49: 264–276.
- Havard, V., B. Jeanne, M. Lacomblez, and D. Baudry. 2019. "Digital Twin and Virtual Reality: A Co-Simulation Environment for Design and Assessment of Industrial Workstations." *Production & Manufacturing Research* 7 (1): 472–489.
- Hietanen, A., R. Pieters, M. Lanz, J. Latokartano, and J. K. Kämäräinen. 2020. "AR-based Interaction for Human-Robot Collaborative Manufacturing." *Robotics and Computer-Integrated Manufacturing* 63: 101891.
- Hoedt, S., A. Claeys, H. Van Landeghem, and J. Cottyn. 2017. "The Evaluation of an Elementary Virtual Training System for Manual Assembly." *International Journal of Production Research* 55 (24): 7496–7508.
- Hosseini, S., and D. Ivanov. 2020. "Bayesian Networks for Supply Chain Risk, Resilience and Ripple Effect Analysis: A Literature Review." *Expert Systems with Applications* 161: 113649.
- Hosseini, S., D. Ivanov, and A. Dolgui. 2019. "Review of Quantitative Methods for Supply Chain Resilience Analysis." *Transportation Research Part E: Logistics and Transportation Review* 125: 285–307.
- Hu, S. J., J. Ko, L. Weyand, H. A. ElMaraghy, T. K. Lien, Y. Koren, H. Bley, G. Chryssolouris, N. Nasr, and M. Shpitalni. 2011. "Assembly System Design and Operations for Product Variety." *CIRP Annals* 60 (2): 715–733.
- Hu, S. J., X. Zhu, H. Wang, and Y. Koren. 2008. "Product Variety and Manufacturing Complexity in Assembly Systems and Supply Chains." *CIRP Annals* 57 (1): 45–48.
- Huang, G. Q., Y. F. Zhang, X. Chen, and S. T. Newman. 2008. "RFID-Enabled Real-Time Wireless Manufacturing for Adaptive Assembly Planning and Control." *Journal of Intelligent Manufacturing* 19 (6): 701–713.
- Huang, Z., J. Zhu, L. Yang, B. Xue, J. Wu, and Z. Zhao. 2015. "Accurate 3-D Position and Orientation Method for Indoor Mobile Robot Navigation Based on Photoelectric Scanning." *IEEE Transactions on Instrumentation and Measurement* 64 (9): 2518–2529.
- Huo, J., F. T. Chan, C. K. Lee, J. O. Strandhagen, and B. Niu. 2020a. "Smart Control of the Assembly Process with a Fuzzy Control System in the Context of Industry 4.0." Advanced Engineering Informatics 43: 101031.
- Huo, J., J. Zhang, and F. T. Chan. 2020b. "A Fuzzy Control System for Assembly Line Balancing with a Three-State Degradation Process in the era of Industry 4.0." *International Journal of Production Research* 58 (23): 7112–7129.
- Indri, M., L. Lachello, I. Lazzero, F. Sibona, and S. Trapani. 2019. "Smart Sensors Applications for a new Paradigm of a Production Line." Sensors 19 (3): 650.
- Jensen-Nau, K. R., T. Hermans, and K. K. Leang. 2020. "Near-Optimal Area-Coverage Path Planning of Energy-Constrained Aerial Robots With Application in Autonomous Environmental Monitoring." *IEEE Transactions on Automation Science and Engineering* 18 (3): 1453–1468.

- Jiang, M., K. H. Leung, Z. Lyu, and G. Q. Huang. 2020. "Picking-Replenishment Synchronization for Robotic Forward-Reserve Warehouses." *Transportation Research Part E: Logistics* and *Transportation Review* 144: 102138.
- Kartal, H., A. Oztekin, A. Gunasekaran, and F. Cebi. 2016. "An Integrated Decision Analytic Framework of Machine Learning with Multi-Criteria Decision Making for Multi-Attribute Inventory Classification." Computers & Industrial Engineering 101: 599–613.
- Keung, K. L., C. K. Lee, P. Ji, and K. K. Ng. 2020. "Cloud-Based Cyber-Physical Robotic Mobile Fulfillment Systems: A Case Study of Collision Avoidance." *IEEE Access* 8: 89318–89336.
- Kim, S., M. A. Nussbaum, and J. L. Gabbard. 2019. "Influences of Augmented Reality Head-Worn Display Type and User Interface Design on Performance and Usability in Simulated Warehouse Order Picking." *Applied Ergonomics* 74: 186–193.
- Kirazli, A., and R. Hormann. 2015. "A Conceptual Approach for Identifying Industrie 4.0 Application Scenarios." In *IIE Annual Conference. Proceedings*, 862–871. Institute of Industrial and Systems Engineers (IISE). https://www.proquest. com/scholarly-journals/conceptual-approach-identifyingindustrie-4-0/docview/1791989039/se-2?accountid = 12870.
- Kolbeinsson, A., E. Lagerstedt, and J. Lindblom. 2019. "Foundation for a Classification of Collaboration Levels for Human-Robot Cooperation in Manufacturing." Production & Manufacturing Research 7 (1): 448–471.
- Koul, S., and T. K. Horiuchi. 2019. "Waypoint Path Planning With Synaptic-Dependent Spike Latency." *IEEE Trans*actions on Circuits and Systems I: Regular Papers 66 (4): 1544–1557.
- Kousi, N., S. Koukas, G. Michalos, and S. Makris. 2019. "Scheduling of Smart Intra–Factory Material Supply Operations Using Mobile Robots." *International Journal of Production Research* 57 (3): 801–814.
- Krippendorff, K. 2004. Content Analysis: An Introduction to Its Methodology. Thousand Oaks, CA: Sage.
- Krishnamurthy, R., and J. Cecil. 2018. "A Next-Generation IoT-Based Collaborative Framework for Electronics Assembly." The International Journal of Advanced Manufacturing Technology 96 (1–4): 39–52.
- Krüger, J., T. K. Lien, and A. Verl. 2009. "Cooperation of Human and Machines in Assembly Lines." *CIRP Annals* 58 (2): 628–646.
- Kucukkoc, I., and D. Z. Zhang. 2017. "Balancing of Mixed-Model Parallel U-Shaped Assembly Lines Considering Model Sequences." *International Journal of Production Research* 55 (20): 5958–5975.
- Kucukoglu, I., H. Atici-Ulusu, T. Gunduz, and O. Tokcalar. 2018. "Application of the Artificial Neural Network Method to Detect Defective Assembling Processes by Using a Wearable Technology." *Journal of Manufacturing Systems* 49: 163–171.
- Kumar, R., L. Ganapathy, R. Gokhale, and M. K. Tiwari. 2020. "Quantitative Approaches for the Integration of Production and Distribution Planning in the Supply Chain: A Systematic Literature Review." *International Journal of Production Research* 58 (11): 3527–3553.

- Lasi, H., P. Fettke, H. G. Kemper, T. Feld, and M. Hoffmann. 2014. "Industry 4.0." Business & Information Systems Engineering 6 (4): 239–242.
- Lee, C. K. M., Y. Lv, K. K. H. Ng, W. Ho, and K. L. Choy. 2018. "Design and Application of Internet of Things-Based Warehouse Management System for Smart Logistics." *International Journal of Production Research* 56 (8): 2753–2768.
- Lei, P., L. Zheng, L. Wang, Y. Wang, C. Li, and X. Li. 2017. "MTConnect Compliant Monitoring for Finishing Assembly Interfaces of Large-Scale Components: A Vertical Tail Section Application." *Journal of Manufacturing Systems* 45: 121–134.
- Li, L., Y. H. Liu, K. Wang, and M. Fang. 2014. "Estimating Position of Mobile Robots from Omnidirectional Vision Using an Adaptive Algorithm." *IEEE Transactions on Cybernetics* 45 (8): 1633–1646.
- Li, L., K. Ota, and M. Dong. 2018. "Deep Learning for Smart Industry: Efficient Manufacture Inspection System with Fog Computing." *IEEE Transactions on Industrial Informatics* 14 (10): 4665–4673.
- Lindenmeyer, C. 2001. How to Design and Conduct a Computer-Integrated Time Study with Active Element Performance Rating (CITS APR). Copyright, C-FOUR July 8, 2001. Page 1 of 16.
- Liu, Y., S. Li, and J. Wang. 2015a. "Assembly Auxiliary System for Narrow Cabins of Spacecraft." *Chinese Journal of Mechanical Engineering* 28 (5): 1080–1088.
- Liu, Y., S. Li, J. Wang, H. Zeng, and J. Lu. 2015b. "A Computer Vision-Based Assistant System for the Assembly of Narrow Cabin Products." *The International Journal of Advanced Manufacturing Technology* 76 (1–4): 281– 293.
- Liu, M., J. Ma, L. Lin, M. Ge, Q. Wang, and C. Liu. 2017. "Intelligent Assembly System for Mechanical Products and key Technology Based on Internet of Things." *Journal of Intelligent Manufacturing* 28 (2): 271–299.
- Lo, S. Y., S. Zhang, and P. Stone. 2020. "The PETLON Algorithm to Plan Efficiently for Task-Level-Optimal Navigation." *Journal of Artificial Intelligence Research* 69: 471–500.
- Lolli, F., E. Balugani, A. Ishizaka, R. Gamberini, B. Rimini, and A. Regattieri. 2019. "Machine Learning for Multi-Criteria Inventory Classification Applied to Intermittent Demand." *Production Planning & Control* 30 (1): 76–89.
- Lolli, F., A. Ishizaka, R. Gamberini, E. Balugani, and B. Rimini. 2017. "Decision Trees for Supervised Multi-Criteria Inventory Classification." *Procedia Manufacturing* 11: 1871–1881.
- Luo, J., Y. Wan, W. Wu, and Z. Li. 2019. "Optimal Petri-Net Controller for Avoiding Collisions in a Class of Automated Guided Vehicle Systems." *IEEE Transactions on Intelligent Transportation Systems* 21 (11): 4526–4537.
- Lyu, Z., P. Lin, D. Guo, and G. Q. Huang. 2020. "Towards Zero-Warehousing Smart Manufacturing from Zero-Inventory Just-In-Time Production." *Robotics and Computer-Integrated Manufacturing* 64: 101932.
- Lyu, X., Y. Song, C. He, Q. Lei, and W. Guo. 2019. "Approach to Integrated Scheduling Problems Considering Optimal Number of Automated Guided Vehicles and Conflict-Free Routing in Flexible Manufacturing Systems." *IEEE Access* 7: 74909–74924.

24 👄 A. DOLGUI ET AL.

- Malik, A. A., and A. Brem. 2020. Digital Twins for Collaborative Robots: A Case Study in Human-Robot Interaction." *Robotics and Computer-Integrated Manufacturing* 68: 102092.
- Mantegh, I., M. R. Jenkin, and A. A. Goldenberg. 2010. "Path Planning for Autonomous Mobile Robots Using the Boundary Integral Equation Method." *Journal of Intelligent & Robotic Systems* 59 (2): 191–220.
- Marques, B., J. Alves, M. Neves, I. Justo, A. Santos, R. Rainho, R. Maio, et al. 2020. "Interaction with Virtual Content Using Augmented Reality: A User Study in Assembly Procedures." *Proceedings of the ACM on Human-Computer Interaction* 4 (ISS): 1–17.
- Mateus, J. C., D. Claeys, V. Limère, J. Cottyn, and E. H. Aghezzaf. 2019. "A Structured Methodology for the Design of a Human-Robot Collaborative Assembly Workplace." *The International Journal of Advanced Manufacturing Technology* 102 (5–8): 2663–2681.
- Mei, Y., Y. H. Lu, Y. C. Hu, and C. G. Lee. 2005, July. "A Case Study of Mobile Robot's Energy Consumption and Conservation Techniques." In *ICAR'05. Proceedings*, 12th International Conference on Advanced Robotics, 2005. Seattle, WA, USA, 492–497. IEEE. doi:10.1109/ICAR.2005.1507454.
- Michalos, G., A. Karvouniari, N. Dimitropoulos, T. Togias, and S. Makris. 2018a. "Workplace Analysis and Design Using Virtual Reality Techniques." CIRP Annals 67 (1): 141–144.
- Michalos, G., N. Kousi, P. Karagiannis, C. Gkournelos, K. Dimoulas, S. Koukas, K. Mparis, A. Papavasileiou, and S. Makris. 2018b. "Seamless Human Robot Collaborative Assembly-An Automotive Case Study." *Mechatronics* 55: 194–211.
- Miller, J., M. Hoover, and E. Winer. 2020. "Mitigation of the Microsoft HoloLens' Hardware Limitations for a Controlled Product Assembly Process." *The International Journal of Advanced Manufacturing Technology* 109 (5): 1741–1754.
- Mohammed, A., B. Schmidt, and L. Wang. 2017. "Active Collision Avoidance for Human–Robot Collaboration Driven by Vision Sensors." *International Journal of Computer Integrated Manufacturing* 30 (9): 970–980.
- Moussa, M., and H. ElMaraghy. 2019. "Master Assembly Network for Alternative Assembly Sequences." *Journal of Man*ufacturing Systems 51: 17–28.
- Moysis, L., E. Petavratzis, C. Volos, H. Nistazakis, and I. Stouboulos. 2020. "A Chaotic Path Planning Generator Based on Logistic map and Modulo Tactics." *Robotics and Autonomous Systems* 124: 103377.
- Mukund Nilakantan, J., and S. G. Ponnambalam. 2016. "Robotic U-Shaped Assembly Line Balancing Using Particle Swarm Optimization." *Engineering Optimization* 48 (2): 231–252.
- Muthukumaran, S., and R. Sivaramakrishnan. 2019. "Optimal Path Planning for an Autonomous Mobile Robot Using Dragonfly Algorithm." *International Journal of Simulation Modelling* 18 (3): 397–407.
- Negri, E., S. Berardi, L. Fumagalli, and M. Macchi. 2020. "MESintegrated Digital Twin Frameworks." *Journal of Manufacturing Systems* 56: 58–71.
- Nielsen, I., Q. V. Dang, G. Bocewicz, and Z. Banaszak. 2017. "A Methodology for Implementation of Mobile Robot in Adaptive Manufacturing Environments." *Journal of Intelligent Manufacturing* 28 (5): 1171–1188.

- Nikolakis, N., K. Alexopoulos, E. Xanthakis, and G. Chryssolouris. 2019a. "The Digital Twin Implementation for Linking the Virtual Representation of Human-Based Production Tasks to Their Physical Counterpart in the Factory-Floor." *International Journal of Computer Integrated Manufacturing* 32 (1): 1–12.
- Nikolakis, N., V. Maratos, and S. Makris. 2019b. "A Cyber Physical System (CPS) Approach for Safe Human-Robot Collaboration in a Shared Workplace." *Robotics and Computer-Integrated Manufacturing* 56: 233–243.
- Nishi, T., and Y. Tanaka. 2012. "Petri Net Decomposition Approach for Dispatching and Conflict-Free Routing of Bidirectional Automated Guided Vehicle Systems." *IEEE Transactions on Systems, Man, and Cybernetics-Part A: Systems and Humans* 42 (5): 1230–1243.
- Nuzzi, C., S. Pasinetti, R. Pagani, S. Ghidini, M. Beschi, G. Coffetti, and G. Sansoni. 2020. MEGURU: A Gesture-Based Robot Program Builder for Meta-Collaborative Workstations." *Robotics and Computer-Integrated Manufacturing* 68: 102085.
- Ohnishi, N., and A. Imiya. 2013. "Appearance-Based Navigation and Homing for Autonomous Mobile Robot." *Image* and Vision Computing 31 (6-7): 511–532.
- Orru', P F, S Arena, C Loi, V Giussani, and M Natalini. 2019. "Production line optimization in an Engineering to Order Business (petrochemical industry) using Value-stream mapping method." XXIV Summer School "Francesco Turco 1: 300–306.
- Otto, A., and X. Li. 2020. "Product Sequencing in Multiple-Piece-Flow Assembly Lines." *Omega* 91: 102055.
- Oyekan, J., A. Fischer, W. Hutabarat, C. Turner, and A. Tiwari. 2019. "Utilising Low Cost RGB-D Cameras to Track the Real Time Progress of a Manual Assembly Sequence." *Assembly Automation* 40 (6): 925–939.
- Oztemel, E., and S. Gursev. 2020. "Literature Review of Industry 4.0 and Related Technologies." *Journal of Intelligent Manufacturing* 31 (1): 127–182.
- Panda, M. R., P. Das, and S. Pradhan. 2017. "Hybridization of IWO and IPSO for Mobile Robots Navigation in a Dynamic Environment." *Journal of King Saud University-Computer* and Information Sciences 32 (9): 1020–1033.
- Papakostas, N., G. Pintzos, C. Giannoulis, and G. Chryssolouris. 2016. "An Agent-Based Collaborative Platform for the Design of Assembly Lines." *International Journal of Computer Integrated Manufacturing* 29 (4): 374–385.
- Peron, M., F. Sgarbossa, and J. O. Strandhagen. 2020. "Decision Support Model for Implementing Assistive Technologies in Assembly Activities: A Case Study." *International Journal* of Production Research, 1–27. doi:10.1080/00207543.2020. 1856441.
- Peruzzini, M., F. Grandi, S. Cavallaro, and M. Pellicciari. 2020a. "Using Virtual Manufacturing to Design Human-Centric Factories: An Industrial Case." *The International Journal of Advanced Manufacturing Technology* 115: 873–887.
- Peruzzini, M., F. Grandi, and M. Pellicciari. 2020b. "Exploring the Potential of Operator 4.0 Interface and Monitoring." *Computers & Industrial Engineering* 139: 105600.
- Petrović, M., Z. Miljković, and A. Jokić. 2019. "A Novel Methodology for Optimal Single Mobile Robot Scheduling Using Whale Optimization Algorithm." *Applied Soft Computing* 81: 105520.

- Petrović, M., N. Vuković, M. Mitić, and Z. Miljković. 2016. "Integration of Process Planning and Scheduling Using Chaotic Particle Swarm Optimization Algorithm." *Expert Systems with Applications* 64: 569–588.
- Piller, F. T. 2005. "Mass Customization: Reflections on the State of the Concept." *International Journal of Flexible Manufacturing Systems* 16 (4): 313–334.
- Piller, F. T. 2007. "Observations on the Present and Future of Mass Customization." *International Journal of Flexible Manufacturing Systems* 19 (4): 630–636.
- Plantard, P., H. P. Shum, A. S. Le Pierres, and F. Multon. 2017. "Validation of an Ergonomic Assessment Method Using Kinect Data in Real Workplace Conditions." *Applied Ergonomics* 65: 562–569.
- Pollard, D., S. Chuo, and B. Lee. 2008. "Strategies for Mass Customization." *Journal of Business & Economics Research* (*JBER*) 6 (7): 77–86. doi:10.19030/jber.v6i7.2447.
- Posadas, J. L., J. L. Poza, J. E. Simó, G. Benet, and F. Blanes. 2008. "Agent-based Distributed Architecture for Mobile Robot Control." *Engineering Applications of Artificial Intelligence* 21 (6): 805–823.
- Prati, E., M. Peruzzini, M. Pellicciari, and R. Raffaeli. 020. "How to Include User EXperience in the Design of Human-Robot Interaction." *Robotics and Computer-Integrated Manufactur*ing 68: 102072.
- Qian, C., Y. Zhang, C. Jiang, S. Pan, and Y. Rong. 2020. "A Real-Time Data-Driven Collaborative Mechanism in Fixed-Position Assembly Systems for Smart Manufacturing." Robotics and Computer-Integrated Manufacturing 61: 101841.
- Qu, H., S. X. Yang, A. R. Willms, and Z. Yi. 2009. "Real-Time Robot Path Planning Based on a Modified Pulse-Coupled Neural Network Model." *IEEE Transactions on Neural Net*works 20 (11): 1724–1739.
- Rabbani, M., M. Moghaddam, and N. Manavizadeh. 2012. "Balancing of Mixed-Model two-Sided Assembly Lines with Multiple U-Shaped Layout." *The International Journal of Advanced Manufacturing Technology* 59 (9–12): 1191–1210.
- Radder, L., and L. Louw. 1999. "Mass Customization and Mass Production." *The TQM Magazine* 11 (1): 35–40.
- Rahman, H. F., M. N. Janardhanan, and P. Nielsen. 2020. "An Integrated Approach for Line Balancing and AGV Scheduling Towards Smart Assembly Systems." Assembly Automation 40 (2): 219–234.
- Rekiek, B., A. Dolgui, A. Delchambre, and A. Bratcu. 2002. "State of art of Optimization Methods for Assembly Line Design." *Annual Reviews in Control* 26 (2): 163–174.
- Riva, G. 2002. "Virtual Reality for Health Care: The Status of Research." *Cyberpsychology & Behavior* 5 (3): 219–225.
- Roda, I., S. Arena, M. Macchi, and P. F. Orru'. 2019. "Total Cost of Ownership Driven Methodology for Predictive Maintenance Implementation in Industrial Plants.." *IFIP International Conference on Advances in Production Management Systems* 566: 315–322.
- Roodbergen, K. J., and I. F. Vis. 2006. "A Model for Warehouse Layout." *IIE Transactions* 38 (10): 799–811.
- Runji, J. M., and C. Y. Lin. 2020. "Switchable Glass Enabled Contextualization for a Cyber-Physical Safe and Interactive Spatial Augmented Reality PCBA Manufacturing Inspection System." Sensors 20 (15): 4286.

- Ruppert, T., and J. Abonyi. 2020. "Integration of Real-Time Locating Systems into Digital Twins." *Journal of Industrial Information Integration* 20: 100174.
- Saeed, R. A., D. R. Recupero, and P. Remagnino. 2020. "A Boundary Node Method for Path Planning of Mobile Robots." *Robotics and Autonomous Systems* 123: 103320.
- Sali, M., and E. Sahin. 2016. "Line Feeding Optimization for Just in Time Assembly Lines: An Application to the Automotive Industry." *International Journal of Production Economics* 174: 54–67.
- Sassi, P., P. Tripicchio, and C. A. Avizzano. 2019. "A Smart Monitoring System for Automatic Welding Defect Detection." *IEEE Transactions on Industrial Electronics* 66 (12): 9641–9650.
- Schmid, N. A., and V. Limère. 2019. "A Classification of Tactical Assembly Line Feeding Problems." *International Journal of Production Research* 57 (24): 7586–7609.
- Schwab, K. 2016. *The Fourth Industrial Revolution*. New York: Crown Publishing Group.
- SCIMAGO. 2020. Scimago Journal and Country Rank. Accessed 25 January 2021. https://www.scimagojr.com/.
- Seuring, S., and S. Gold. 2012. "Conducting Content-Analysis Based Literature Reviews in Supply Chain Management." Supply Chain Management: An International Journal 17 (5): 544–555. doi:10.1108/13598541211258609.
- Shrouf, F., J. Ordieres, and G. Miragliotta. 2014, December. "Smart Factories in Industry 4.0: A Review of the Concept and of Energy Management Approached in Production Based on the Internet of Things Paradigm." In 2014 IEEE International Conference on Industrial Engineering and Engineering Management. Selangor, Malaysia, 697–701. IEEE. doi:10.1109/IEEM.2014.7058728.
- Song, B., Z. Wang, and L. Sheng. 2016. "A new Genetic Algorithm Approach to Smooth Path Planning for Mobile Robots." Assembly Automation 36 (2): 138–145.
- Stadnicka, D., and D. Antonelli. 2019. "Human-robot Collaborative Work Cell Implementation Through Lean Thinking." *International Journal of Computer Integrated Manufacturing* 32 (6): 580–595.
- Tan, Q., Y. Tong, S. Wu, and D. Li. 2019. "Modeling, Planning, and Scheduling of Shop-Floor Assembly Process with Dynamic Cyber-Physical Interactions: A Case Study for CPS-Based Smart Industrial Robot Production." *The International Journal of Advanced Manufacturing Technology* 105 (9): 3979–3989.
- Tao, W., M. C. Leu, and Z. Yin. 2020. "Multi-modal Recognition of Worker Activity for Human-Centered Intelligent Manufacturing." *Engineering Applications of Artificial Intelligence* 95: 103868.
- Tarallo, A., R. Mozzillo, G. Di Gironimo, and R. De Amicis. 2018. "A Cyber-Physical System for Production Monitoring of Manual Manufacturing Processes." *International Journal* on Interactive Design and Manufacturing (IJIDeM 12 (4): 1235–1241.
- Tejesh, B. S. S., and S. J. A. E. J. Neeraja. 2018. "Warehouse Inventory Management System Using IoT and Open Source Framework." *Alexandria Engineering Journal* 57 (4): 3817–3823.
- Thames, L., and D. Schaefer. 2016. "Software-defined Cloud Manufacturing for Industry 4.0." *Procedia Cirp* 52: 12–17.

26 👄 A. DOLGUI ET AL.

- Tranfield, David, David Denyer, and Palminder Smart. 2003. "Towards a Methodology for Developing Evidence-Informed Management Knowledge by Means of Systematic Review." British Journal of Management 14: 207–222.
- Unhelkar, V. V., S. Dörr, A. Bubeck, P. A. Lasota, J. Perez, H. C. Siu, J. C. Boerkoel, et al. 2018a. "Mobile Robots for Moving-Floor Assembly Lines: Design, Evaluation, and Deployment." *IEEE Robotics & Automation Magazine* 25 (2): 72–81.
- Unhelkar, V. V., P. A. Lasota, Q. Tyroller, R. D. Buhai, L. Marceau, B. Deml, and J. A. Shah. 2018b. "Human-Aware Robotic Assistant for Collaborative Assembly: Integrating Human Motion Prediction with Planning in Time." *IEEE Robotics and Automation Letters* 3 (3): 2394–2401.
- Vanneste, P., Y. Huang, J. Y. Park, F. Cornillie, B. Decloedt, and W. Van den Noortgate. 2020. "Cognitive Support for Assembly Operations by Means of Augmented Reality: An Exploratory Study." *International Journal of Human-Computer Studies* 143: 102480.
- Wagner, R., B. Haefner, M. Biehler, and G. Lanza. 2020. "Digital DNA in Quality Control Cycles of High-Precision Products." *CIRP Annals* 69: 373–379.
- Walker, J. H., S. M. Garrett, and M. S. Wilson. 2006. "The Balance Between Initial Training and Lifelong Adaptation in Evolving Robot Controllers." *IEEE Transactions on Systems*, *Man, and Cybernetics, Part B (Cybernetics* 36 (2): 423–432.
- Wan, J., S. Tang, Q. Hua, D. Li, C. Liu, and J. Lloret. 2017. "Context-aware Cloud Robotics for Material Handling in Cognitive Industrial Internet of Things." *IEEE Internet of Things Journal* 5 (4): 2272–2281.
- Wang, X. V., Z. Kemény, J. Váncza, and L. Wang. 2017. "Human–Robot Collaborative Assembly in Cyber-Physical Production: Classification Framework and Implementation." *CIRP Annals* 66 (1): 5–8.
- Wang, K., D. Liu, Z. Liu, Q. Wang, and J. Tan. 2020a. "An Assembly Precision Analysis Method Based on a General Part Digital Twin Model." *Robotics and Computer-Integrated Manufacturing* 68: 102089.
- Wang, K. J., D. A. Rizqi, and H. P. Nguyen. 2020b. "Skill Transfer Support Model Based on Deep Learning." Journal of Intelligent Manufacturing 32: 1129–1146.
- Wänström, C., and L. Medbo. 2009. "The Impact of Materials Feeding Design on Assembly Process Performance." *Journal* of Manufacturing Technology Management 20: 30–51.
- Weckenborg, C., K. Kieckhäfer, C. Müller, M. Grunewald, and T. S. Spengler. 2020. "Balancing of Assembly Lines with Collaborative Robots." *Business Research* 13 (1): 93–132.
- Westerfield, G., A. Mitrovic, and M. Billinghurst. 2015. "Intelligent Augmented Reality Training for Motherboard Assembly." *International Journal of Artificial Intelligence in Education* 25 (1): 157–172.
- Winkelhaus, S., and E. H. Grosse. 2020. "Logistics 4.0: A Systematic Review Towards a New Logistics System." International Journal of Production Research 58 (1): 18–43.
- Wu, S. Q., B. Shen, Y. Z. Tang, J. H. Wang, and D. T. Zheng. 2018. "Ergonomic Layout Optimization of a Smart Assembly Workbench." Assembly Automation 38: 314–322.
- Wu, S., Z. Wang, B. Shen, J. H. Wang, and L. Dongdong. 2020. "Human-computer Interaction Based on Machine Vision of a Smart Assembly Workbench." *Assembly Automation* 40: 475–482.

- Xia, F., L. T. Yang, L. Wang, and A. Vinel. 2012. "Internet of Things." *International Journal of Communication Systems* 25 (9): 1101.
- Xiao, H., Y. Duan, Z. Zhang, and M. Li. 2018. "Detection and Estimation of Mental Fatigue in Manual Assembly Process of Complex Products." Assembly Automation 38: 239–247.
- Xu, P., H. Mei, L. Ren, and W. Chen. 2016. "ViDX: Visual Diagnostics of Assembly Line Performance in Smart Factories." *IEEE Transactions on Visualization and Computer Graphics* 23 (1): 291–300.
- Xu, S., F. Xiao, N. Si, L. Sun, W. Wu, and V. H. C. de Albuquerque. 2020. "GQM: Autonomous Goods Quantity Monitoring in IIoT Based on Battery-Free RFID." *Mechanical Systems and Signal Processing* 136: 106411.
- Yao, B., Z. Zhou, L. Wang, W. Xu, J. Yan, and Q. Liu. 2018. "A Function Block Based Cyber-Physical Production System for Physical Human–Robot Interaction." *Journal of Manufacturing Systems* 48: 12–23.
- Yelles-Chaouche, A. R., E. Gurevsky, N. Brahimi, and A. Dolgui. 2020. "Reconfigurable Manufacturing Systems from an Optimisation Perspective: A Focused Review of Literature." *International Journal of Production Research*, 1–19. doi:10.1080/00207543.2020.1813913.
- Yi, Y., Y. Yan, X. Liu, Z. Ni, J. Feng, and J. Liu. 2020. "Digital Twin-Based Smart Assembly Process Design and Application Framework for Complex Products and its Case Study." *Journal of Manufacturing Systems* 58: 94–107.
- Yoshitake, H., R. Kamoshida, and Y. Nagashima. 2019. "New Automated Guided Vehicle System Using Real-Time Holonic Scheduling for Warehouse Picking." *IEEE Robotics* and Automation Letters 4 (2): 1045–1052.
- Yu, Q., and K. Wang. 2013. "3D Vision Based Quality Inspection with Computational Intelligence." Assembly Automation 33: 240–246.
- Yuan, M. L., S. K. Ong, and A. Y. C. Nee. 2008. "Augmented Reality for Assembly Guidance Using a Virtual Interactive Tool." *International Journal of Production Research* 46 (7): 1745–1767.
- Zeng, C., J. Tang, and C. Yan. 2015. "Job-shop Cell-Scheduling Problem with Inter-Cell Moves and Automated Guided Vehicles." *Journal of Intelligent Manufacturing* 26 (5): 845–859.
- Zhang, J., S. K. Ong, and A. Y. C. Nee. 2011. "RFID-assisted Assembly Guidance System in an Augmented Reality Environment." *International Journal of Production Research* 49 (13): 3919–3938.
- Zhao, Z., M. Zhang, C. Yang, J. Fang, and G. Q. Huang. 2018. "Distributed and Collaborative Proactive Tandem Location Tracking of Vehicle Products for Warehouse Operations." *Computers and Industrial Engineering* 125: 637–648.
- Zhong, Y. G., and B. Ai. 2017. "A Modified ant Colony Optimization Algorithm for Multi-Objective Assembly Line Balancing." Soft Computing 21 (22): 6881–6894.
- Zhong, R. Y., X. Xu, E. Klotz, and S. T. Newman. 2017. "Intelligent Manufacturing in the Context of Industry 4.0: A Review." *Engineering* 3 (5): 616–630.
- Zhou, L., J. Liu, X. Fan, D. Zhu, P. Wu, and N. Cao. 2019. "Design of V-Type Warehouse Layout and Picking Path Model Based on Internet of Things." *IEEE Access* 7: 58419–58428.

- Zhou, B. H., and J. H. Xu. 2018. "An Adaptive Large Neighbourhood Search-Based Optimisation for Economic co-Scheduling of Mobile Robots." *European Journal of Industrial Engineering* 12 (6): 832–854.
- Zhuang, C., J. Liu, and H. Xiong. 2018. "Digital Twin-Based Smart Production Management and Control Framework for the Complex Product Assembly Shop-Floor." *The International Journal of Advanced Manufacturing Technology* 96 (1–4): 1149–1163.
- Židek, K., J. Piteľ, M. Adámek, P. Lazorík, and A. Hošovský. 2020. "Digital Twin of Experimental Smart Manufacturing

Assembly System for Industry 4.0 Concept." *Sustainability* 12 (9): 3658.

- Zipkin, P. 2001. "The Limits of Mass Customization." *MIT* Sloan Management Review 42 (3): 81.
- Zou, Q., M. Cong, D. Liu, and Y. Du. 2019. "Robotic Path Planning Based on Episodic-Cognitive Map. International Journal of Control." *Automation and Systems* 17 (5): 1304– 1313.
- Zülch, M., and G. Zülch. 2017. "Production Logistics and Ergonomic Evaluation of U-Shaped Assembly Systems." International Journal of Production Economics 190: 37–44.

Paper 2

Simonetto, M. and Sgarbossa, F., 2021, September. Straight and U-Shaped Assembly Lines in Industry 4.0 Era: Factors Influencing Their Implementation. In IFIP International Conference on Advances in Production Management Systems (pp. 414-422). Springer, Cham.

This is an accepted manuscript of an article published in IFIP International Conference on Advances in Production Management Systems.

The final authenticated version is available online at: https://doi.org/10.1007/978-3-030-57993-7_19



Introduction to Material Feeding 4.0: Strategic, Tactical, and Operational Impact

Marco Simonetto^(⊠) and Fabio Sgarbossa

Department of Mechanical and Industrial Engineering, Norwegian University of Science and Technology, S P Andersens V 3, 7031 Trondheim, Norway marco.simonetto@ntnu.com

Abstract. Mass customization, the process of producing a low-volume highvariety of products, is changing production environments. In Material Feeding (MF) this means a huge increment in the number of parts, and information, that need to be managed during the different MF activities. If companies want to get high performances from their MF activities, they need to be able to manage these changes in the best manner. Industry 4.0 technologies are introducing new opportunities to help companies in the execution and control of the MF activities. It is important for companies to be able to understand how to implement these technologies in their processes and how to take these opportunities. In order to facilitate this, in this paper the concept of Material Feeding 4.0 (MF 4.0) is presented for the first time, as Material Feeding where the Industry 4.0 technologies are introduced. The impact of the identified technologies is studied at a strategic, tactical and operational level.

Keywords: Material Feeding · Industry 4.0 · Assembly system · Strategic · Tactical · Operational

1 Introduction

The demand for customized products is increasing. A high demand of customized products implies that the production and assembly lines require to be fed with an enormous amount of different parts. Material feeding (MF) describes all the activities that are responsible for the provision of components, parts, equipment, etc. to production and assembly systems when needed [1]. If companies want to satisfy the demand of their customers these activities need to be continuously improved and optimized. In this paper the focus will be in the assembly system (AS). MF, in AS, can be performed in different ways, by storing all parts near the assembly line or preprocessing parts and delivering them when needed [2]. In the last years with the socalled Industry 4.0 revolution, both AS and MF are considering the introduction of the technologies that this revolution is offering in their planning and control processes. However, although other works already exist that try to study the effects of these technologies on the AS, [3, 4], still no work exists about their effects in the planning and control processes of the MF systems. Although in literature the terms Supply Chain 4.0 and Logistics 4.0 have been already discussed with a broader perspective, [5, 6], in this paper we focus on the specific process and set of activities related to MF. The idea is to understand which are the emerging technologies that can be implemented in the MF activities and how their implementation can impact at three levels: strategic, tactical, and operational. The remainder of this paper is structured as follows. Section 2 provides a review of the existing literature concentrating on the MF activities. Section 3 illustrates and discusses the results obtained from the review of the literature. Finally, in Sect. 4 the paper is shortly summarized.

2 Review of the Existing Literature

Examples of Industry 4.0 technologies in AS are: Internet of Things (IoT), Big data, Cloud Computing, Augmented Reality, Collaborative Robots, and Additive Manufacturing [3]. These technologies affect the performance of the AS activities. In MF it is not yet defined which emerging technologies can be used. The main activities of MF are transportation, preparation, and material management [7]. Transportation implies the process of moving all the components, parts, from a point A where they are stocked to a point B where they are requested. The preparation implies the processes of handling and repacking parts into load carriers used for the corresponding line feeding policy [7]. Material management instead includes all the process related to the storage of components and products. For the storage of components, two of the possible solutions are the central receiving stores and the supermarkets [8]. An extensive review of the existing literature on the subject has been carried out to see what is possible to find about Industry 4.0 technologies that can have an impact on these three activities. We used Scopus database as search engine limiting the research from January 2006 to March 2020, English as language, and journal and conference papers as sources.

2.1 Transportation

In the planning and control of the transportation activity one of the most important decisions is the choice of the transportation devices. Examples of traditional transportation devices are the forklift, tow train, or feeder line [8]. Thanks to the technology development, new solutions have appeared in the last few years [9]. First the Automatic Guided Vehicles (AGVs), driverless vehicles that follow a fixed path in order to move from a point to another [10], and now Autonomous Mobile Robots (AMRs) are getting more popular due to their flexibility [11]. These mobile robots can move without following a fixed path and use cameras and sophisticated software to identify their surroundings and take the most efficient route to their destination [12]. AGVs and AMRs are platforms in which you can implement different additional equipment/resources such as lifting systems and collaborative robot. They can be used as "workstations" to assemble the products [13] or to move shelves from the warehouse to the AS [11]. Based on how companies decide to use them, they will face different problems in order to optimize the performances. For example, if the AGVs/AMRs are used to move the shelves for the picking operations, an example of a possible problem might be when non-completed shelves have to be moved using AGVs/AMRs that usually move complete shelves. A solution to this problem is presented in [14] where the authors proposed a scheduling method to move the shelves even if they are not complete. In [15] a cloud-based architecture that controls and improves the collaboration of the AMRs is presented. Simulations showed that using this cloud-based architecture gives the opportunity to improve energy efficiency and save costs. AMRs seem to be a very prominent technology in order to improve the transportation activity. This is also thanks to the development of more powerful Data Analytics techniques.

2.2 Preparation

In the preparation activity for AS we can find the picking and kitting process [1]. The process of picking can be described as manually seeking a particular item in storage according to a list, taking that object and putting it into a bin/container the appropriate transport vehicle, and bringing the objects to the required location for processing [16]. The kit preparation process instead supplies AS with kits of components. The kits are prepared at a kit preparation workspace apart from the AS [17]. Picking and kitting operation can be facilitated using a AMRs to transport the necessary components from the warehouse to the AS [11]. AMRs are not used only for the transportation. In the last few years many manufacturers have started to develop mobile robots with various functions [9]. If a collaborative robot is integrated with a mobile robot this can be called collaborative mobile robots [18]. In [18] the collaborative mobile robot is used to perform a part-feeding task. The validity of the solution is granted by the successfully performed tests. Another example of a collaborative robot mounted in a mobile robot is presented in [17]. The collaborative robot sorts the components that have been picked by an operator into a batch of kits. The mobile robot gives the opportunity to the collaborative robot to do its operations moving together with the human operator. This is done safely and relieves the human operator of activities associated with sorting components into kits. Kitting preparation in the context of material supply to AS can be performed utilizing augmented reality (AR) [19]. AR is used in single kit-preparation and in batch preparation. The experiment demonstrates how AR is competitive both in terms of time-efficiency and picking accuracy. AR is used also in picking activities. The main improvements that AR can give to picking operations are the errors reduction and time efficiency [16, 20, 21]. The preparation of the different components and parts to deliver at the AS is a very important activity. A mistake in this activity means a reduction in the performances of the AS. The introduction of AR in this activity is made in order to reduce the probability of making mistakes.

2.3 Material Management

Data Analytics techniques, such as data mining algorithms and machine learning algorithm, are becoming more popular and powerful with the advent of Industry 4.0 [22]. These techniques can be used to improve the material management activities of a company. For example, in [23] a machine learning algorithm is proposed to optimize the warehouse storage location allocation. The solution can improve the efficiency of warehouse operations in case of weak correlation between the stock keeping units. Whereas in [24] a positioning big data forecasting model is used in order to predict the trajectory of the mobile robots. This can improve the safety, reliability and stability of the mobile robot navigation. The introduction of mobile robots in the warehouses can

mean a change in their design. Mobile robots allow spatial flexibility and expandability. If a warehouse needs more capacity, one simply adds more pods, drives, and stations [11]. In order to manage the inventories of warehouses and ASs it is possible to adopt new smart solutions. Internet of Things (IoT) and Cloud computing can be used to automate inventory systems [25]. An example of a solution that can improve this automation of inventory systems is presented in [26] with the concept of selfoptimizing Kanban. Self-optimizing Kanban systems autonomously adjust their capacities as well as the quantity of cards in circulation according to predefined performances. In order to use these systems, it is important to collect all the data of the quantities inside the warehouses in real-time and to use data analytic algorithms to manage the collected data. The data are the most important resource in material management. It is important to know which data need to be collected and how. Once the data are collected, they need to be analyzed in order to obtain valuable information. IoT technologies and Data Analytics techniques will help companies in this MF activity.

3 Material Feeding 4.0

In Fig. 1 we see the results obtained from the review of the literature. From the figure we can see that the most adopted technologies in MF activities are: mobile robots, augmented reality, IoT technologies, Cloud Computing and Data Analytics. These are the technologies that can be present in a MF 4.0 system.



Fig. 1. Industry 4.0 technologies for Material Feeding activities.

Table 1 summarizes where the technologies impact in the different activities at a strategic, tactical, and operational level. The table helps to think about new research challenges that can appear if some of the technologies identified are introduced in the MF system. In order to state the impact of the different technologies at a strategic, tactical, and operational levels in MF 4.0, the scope of every level is shortly explained in the following. Decisions of different levels affect each other and should be considered in an integrated way regarding the intra-organizational decision levels as well as the inter-organizational hierarchy [27]. Due to page restriction, we will use the mobile robots as an example.

	Impact level					
Industry 4.0	Strategic	Tactial	Operational			
technology						
Mobile robots	 Type of mobile robots Design the guide path system Warehouse design Skills to use the sytem 	 Feeding polices Number of mobile robots Schedulling of the mobile robots Human-mobile robot interaction 	 Routes of the mobile robots Contol of the system			
Augmented reality	 Type of AR devices Skills to use the AR devices 	Number of workersErgonomic of the system	 Real time information sharing → sequencing of kitting and picking information Control of the system 			
IoT technologies	 Type of IoT technologies Type of containers Skills to use the IoT technologies 	 Number of containers in the warehouses and in the AS Position of the IoT technologies 	 Real time control of the capacity of the container Control of the system			
Cloud computing and Data Analysis	 Type of Cloud Computing system and/or Data Analytic techniques Data to storage and analysis Skills to use these technologies 	 Information to create and share How to create the information 	 Real-time sharing and displaying of the information Control of the system 			

 Table 1. Decision areas of Material Feeding 4.0. Impact of the different technologies in planning and control.

3.1 Strategic Level in MF 4.0

Strategic decisions are those decisions that have an influence over the years and a longterm impact on the performance of the MF. Once a strategic decision is made, it is very unlikely to be altered in the short term. They are usually taken at the highest levels of management, include a wide range of uncertainties and carry higher levels of risk. In MF 4.0, the most important decision at this level for all the MF 4.0 activities, is the choice of the technologies to adopt. It is possible to decide gradually which are the technologies to implement in the different activities. Which type of mobile robots to buy will influence all the decisions that a company has to face after their purchase. A mobile robot can be used only as a transportation system or it can be integrated with collaborative robot for picking activities [9, 17]. This together with the flexibility of the guidance system of a mobile robot and its dimensions will influence the design of the possible paths and of the warehouses (central receiving stores or the supermarkets) where it will work [10, 12]. The design of the warehouses can be also influenced by the different types of containers that the mobile robots can transport [11]. The investment that a company makes when it decides to buy the mobile robots is not related only to the purchase of the robots. Someone needs to know how to use them. If the mobile robots are introduced in one or more of the MF activities, it is important to know which skills are needed to use them and what is necessary to do in order to best implement them. This can influence the performance given by the mobile robots. For example, if the company decide to not hire new workers that already know how to use them, it is possible that during the first period, their performance will be lower than the expected. This because a certain amount of time is needed to learn how to use them.

3.2 Tactical Level in MF 4.0

Tactical decisions are decisions and plans that concern the more detailed implementation of the strategic decisions, usually with a medium-term impact on a company. At the tactical level, the decisions about which are the technologies to implement are already made. For the mobile robots this means that the company has decided which type to buy. At this level, for all the MF 4.0 activities, it is important to prepare the different technologies to be used. The first thing to understand is which are the feeding policies to be adopted [1]. The flexibility of mobile robots can change the application of the feeding policies making one policy more convenient than another. For example, it can be more convenient to prepare kits than move entire pallets. Mobile robots are also more flexible with respect to the transported volumes and they can adapt themselves with the variation of the material flow. The chosen policies can influence the number of mobile robots that the company needs and respect with traditional transportation devices it is possible that new algorithms are necessaries to calculate this number. In the scheduling phase of the mobile robots, the company decides when, where and how a mobile robot should act to perform tasks [10]. A new problem to solve during this phase is which mobile robot to use based on the operations needed to perform [9]. This because there are different types of mobile robots that are going to be implemented based on the activities that they must do. In this phase is important to consider also the ergonomic aspect of the system. In fact, the safety precautions that one has to implement can be different in order to use the different type of mobile robots. These can change from a mobile robot that is used only as transportation system from another that works together with human workers, for example when it is used as a workstation or when it helps the human operator in the preparation of the kits [13, 17].

3.3 Operational Level in MF 4.0

Operational decisions are related to day-to-day operations of the companies. They have a short-term horizon as they are taken repetitively. At this level, for all the MF 4.0 activities, it is important to ensure that technology works as best as possible. The vehicle routing problem decides the route a mobile robot should take and the sequence of loads (or jobs) that this vehicle should visit [10]. The routes that a mobile robot can travel are different from those of a traditional transportation devise [8]. The routes change also depending on whether the mobile robots must follow a fixed path or not. Regardless of the technology, at the operational level it is possible to find the control of the system. This will give the opportunity to continuously improve the MF 4.0 system and to avoid that it stops working. Checking the operation of a mobile robot is more difficult and requires new data to be collected and new knowledge. This is because they are not guided by human workers, and if something is not working properly, no one can be aware of it if a proper control system is not implemented. It is important to create new solutions that can control the mobile robots during the execution of their activities [15]. The data collected from the mobile robots need to be analyzed, and once these data become information these need to be understood before then take any decisions. Not understand the information generated by the execution of the different activities from the mobile robots means not being able to understand if the system is working properly. This means having a system that is not working with the desired performance and that is not possible to change it in order to improve them.

4 Conclusion

This paper focused on the individualization of the most common Industry 4.0 technologies that can be used during the execution of the MF activities. The considered MF activities are transportation, preparation, and material management. This gives us the opportunity to introduce the concept of MF 4.0. MF systems where the different activities are done with the help of the new technologies of Industry 4.0. Moreover, we give some suggestions about the decisions that need to be taken in MF 4.0 with respect to a strategic, tactical and operational impact level. This is only an introduction in the topic of MF 4.0. There seems to be a high potential for future works in this research stream. Future research for example should focus on understanding how to measure the performance of the different technologies implemented in the MF 4.0 activities. This is related to another possible work that is understanding which are the data that need to be collected from the MF 4.0 activities and how to collect them. The technologies are becoming always more powerful and user friendly and their introduction will increase in the next few years. It will be important to know in advance which will be possible issues that companies will face when they decide to implement one of these technologies.

References

- 1. Battini, D., Faccio, M., Persona, A., Sgarbossa, F.: Design of the optimal feeding policy in an assembly system. Int. J. Prod. Econ. **121**(1), 233–254 (2009)
- 2. Sali, M., Sahin, E.: Line feeding optimization for just in time assembly lines: an application to the automotive industry. Int. J. Prod. Econ. **174**, 54–67 (2016)
- Cohen, Y., Naseraldin, H., Chaudhuri, A., Pilati, F.: Assembly systems in industry 4.0 era: a road map to understand assembly 4.0. Int. J. Adv. Manuf. Technol. 105(9), 4037–4054 (2019)
- Bortolini, M., Ferrari, E., Gamberi, M., Pilati, F., Faccio, M.: Assembly system design in the industry 4.0 era: a general framework. IFAC-PapersOnLine 50(1), 5700–5705 (2017)

- 5. Frederico, G.F., Garza-Reyes, J.A., Anosike, A.I., Kumar, V.: Supply chain 4.0: concepts, maturity and research agenda. Int. J. Supply Chain Manag. 1–21 (2019)
- 6. Winkelhaus, S., Grosse, E.H.: Logistics 4.0: a systematic review towards a new logistics system. Int. J. Prod. Res. **58**(1), 18–43 (2020)
- Schmid, N.A., Limère, V.: A classification of tactical assembly line feeding problems. Int. J. Prod. Res. 57(24), 7586–7609 (2019)
- 8. Battini, D., Boysen, N., Emde, S.: Just-in-time supermarkets for part supply in the automobile industry. J. Manag. Control **24**(2), 209–217 (2013)
- 9. Lottermoser, A., Berger, C., Braunreuther, S., Reinhart, G.: Method of usability for mobile robotics in a manufacturing environment. Procedia CIRP **62**, 594–599 (2017)
- Le-Anh, T., De Koster, M.B.M.: A review of design and control of automated guided vehicle systems. Eur. J. Oper. Res. 171(1), 1–23 (2006)
- 11. Wurman, P.R., D'Andrea, R., Mountz, M.: Coordinating hundreds of cooperative, autonomous vehicles in warehouses. AI Mag. **29**(1), 9 (2008)
- MiR autonomous mobile robot. https://www.mobile-industrial-robots.com. Accessed 30 Mar 2020
- 13. Here's how Audi plans to scrap the assembly line. https://www.autoguide.com/auto-news/ 2017/07/here-s-how-audi-plans-to-scrap-the-assembly-line.html. Accessed 30 Mar 2020
- Yoshitake, H., Kamoshida, R., Nagashima, Y.: New automated guided vehicle system using real-time holonic scheduling for warehouse picking. IEEE Robot. Autom. Lett. 4(2), 1045– 1052 (2019)
- Wan, J., Tang, S., Hua, Q., Li, D., Liu, C., Lloret, J.: Context-aware cloud robotics for material handling in cognitive industrial internet of things. IEEE Internet Things J. 5(4), 2272–2281 (2017)
- Regenbrecht, H., Baratoff, G., Wilke, W.: Augmented reality projects in the automotive and aerospace industries. IEEE Comput. Graph. Appl. 25(6), 48–56 (2005)
- 17. Fager, P., Calzavara, M., Sgarbossa, F.: Modelling time efficiency of cobot-supported kit preparation. Int. J. Adv. Manuf. Technol. **106**(5), 2227–2241 (2020)
- 18. Andersen, R.E., et al.: Integration of a skill-based collaborative mobile robot in a smart cyber-physical environment. Procedia Manuf. **11**, 114–123 (2017)
- Hanson, R., Falkenström, W., Miettinen, M.: Augmented reality as a means of conveying picking information in kit preparation for mixed-model assembly. Comput. Ind. Eng. 113, 570–575 (2017)
- Krajcovic, M., Gabajova, G., Micieta, B.: Order picking using augmented reality. Commun.-Sci. lett. Univ. Zilina 16(3A), 106–111 (2014)
- 21. Schwerdtfeger, B., Reif, R., Günthner, W.A., Klinker, G.: Pick-by-vision: there is something to pick at the end of the augmented tunnel. Virtual Reality **15**(2–3), 213–223 (2011)
- 22. Choi, T.M., Wallace, S.W., Wang, Y.: Big data analytics in operations management. Prod. Oper. Manag. 27(10), 1868–1883 (2018)
- Xin, C., Liu, X., Deng, Y., Lang, Q.: An optimization algorithm based on text clustering for warehouse storage location allocation. In: 1st International Conference on Industrial Artificial Intelligence (IAI), pp. 1–6. IEEE, Shenyang (2019)
- Liu, H., Xu, Y., Wu, X., Lv, X., Zhang, D., Zhong, G.: Big data forecasting model of indoor positions for mobile robot navigation based on apache spark platform. In: 4th International Conference on Cloud Computing and Big Data Analysis (ICCCBDA), pp. 378–382. IEEE, Chengdu (2019)
- Alwadi, A., Gawanmeh, A., Parvin, S., Al-Karaki, J.N.: Smart solutions for RFID based inventory management systems: a survey. Scalable Comput.: Pract. Experience 18(4), 347– 360 (2017)

- Buer, S.V., Fragapane, G.I., Strandhagen, J.O.: The data-driven process improvement cycle: using digitalization for continuous improvement. IFAC-PapersOnLine 51(11), 1035–1040 (2018)
- Bhatnagar, R., Chandra, P., Goyal, S.K.: Models for multi-plant coordination. Eur. J. Oper. Res. 67(2), 141–160 (1993)

Paper 3

Simonetto, M. and Sgarbossa, F., 2021, September. Straight and U-Shaped Assembly Lines in Industry 4.0 Era: Factors Influencing Their Implementation. In IFIP International Conference on Advances in Production Management Systems (pp. 414-422). Springer, Cham.

This is an accepted manuscript of an article published in IFIP International Conference on Advances in Production Management Systems.

The final authenticated version is available online at: https://doi.org/ 10.1007/978-3-030-85914-5_44



Straight and U-Shaped Assembly Lines in Industry 4.0 Era: Factors Influencing Their Implementation

Marco Simonetto^(⊠) and Fabio Sgarbossa

Department of Mechanical and Industrial Engineering, Norwegian University of Science and Technology, S P Andersens V 3, 7031 Trondheim, Norway marco.simonetto@ntnu.com

Abstract. The increasing of products variety is moving companies in rethinking the configuration of their Assembly Systems (ASs). An AS is where the products are being configured in their different variants and its configuration needs to be chosen in order to handle such high variety. Two of the most common AS configurations are the straight-line and the U-shaped line. In this paper we want to first determine and define, through a narrative literature research, which are the factors that are relevant in order to compare these two AS configurations. Additionally, with the advent of Industry 4.0 (I4.0) technologies it is possible that these new technologies have an impact on these factors. For this reason, first, we investigated how I4.0 technologies impact these straight-t line and U-shaped line, change based on the impact of the technologies. The seven factors here defined open the opportunity for future research challenges.

Keywords: Assembly system \cdot Configuration \cdot Straight-line \cdot U-shaped line \cdot Industry 4.0

1 Introduction

An AS configuration is a sequence of several workstations organized in a certain way in order to assemble different parts to final products. The assembly process consists of a sequence of typically ordered tasks which cannot be subdivided and should be completed at its assigned workstation [1]. The two AS configurations more used are the straight-lines and the U-shaped lines [2, 3], Fig. 1. A straight-line configuration is represented by a sequence of workstations in a line. Simply rearranging the straight-line into a U-shaped it is possible to obtain the U-shaped line. In both cases, there can be a human worker in each workstation or there can be single workers managing more workstations [4, 5]. However, in U-shaped lines, it is easier for human workers to move from one workstation to another since they can move across both sides of the line, allowing more flexibility and degrees of freedom in the number of workers that can be assigned in these configurations with respect of the straight-line configurations.

Traditionally, a straight-line is suitable for mass-production of standardized products [6]. Because of the growing demand for customized products, in the last few years,



Fig. 1. Straight-line configuration (a) and U-shaped line configuration (b).

the products' variety is increased [7]. The U-shaped line is an example of layout that can handle this increase of variety [8]. However, recently also straight-lines are gained more importance in low volume production of customized products [9]. This can complicate the decision process of which layout configuration to choose. Moreover, the introduction of I4.0 technologies in these configurations can have an impact in this decision.

Industry 4.0 emerges from the synergy of the availability of innovative technologies and the demand by consumers for high quality and customized products and it aims at connecting resources, services, and humans in real-time throughout the production in order to attain an advanced level of operational effectiveness and productivity, as well as a higher level of automatization [10]. Its economic impact is supposed to be huge [11]. It holds the promise of increased flexibility, better quality, and improved productivity [12]. Industry 4.0 technologies are starting to be an integrated part of the AS creating a new generation of AS, the so-called Assembly System 4.0 (AS4.0) [13]. The technologies that can be adopted from AS4.0 can be grouped into Collaborative Robots, Augmented Reality (AR), Internet of Things (IoTs), and Cloud Computing and Data Analysis [13].

In literature is it possible to find some qualitative guidelines about when to choose an AS configuration respect to another [14] and a work that gives an idea of factors that can be considered is the one of [8]. The factors that [8] used are the number of tasks, network density, and the cycle time. However, these are not the only factors that need to be taken into consideration to compare straight-line and U-shaped line. Therefore, we are going to look through a narrative literature review, at which are other factors that need to be taken into consideration when one wants to compare the two configurations.

This research is done in order to answer the following research questions:

- 1. Which are the factors that have to be considered in order to compare the two AS configurations straight-line and U-shaped line?
- 2. How Industry 4.0 technologies impact on these factors?

The research wants to help companies that are facing the problem of the increase in product variety, or that are simply thinking in changing their AS configuration, in understanding which are the factors that they need to know in order to be able to compare the two AS configurations. This to facilitate their decision process. The implementation of the right Industry 4.0 technologies can give further help to companies to reach their production performance but can also influence the factors. Therefore, it is important to know if and how the technologies are going to impact on the factors. The remainder of this paper is structured as follows. In Sect. 2 we explained the methodology applied in the paper. In Sect. 3, we answered at the first

research question. In Sect. 4, we answered at the second research question and finally, Sect. 5 concludes the work.

2 Methodology

In this paper, to answer at our two research questions, we conducted a narrative literature review. The keywords that we used for the research of the papers are presented in Table 1. We used the Scopus database, searching only result in English as language without limits in the document type and in the year of publications of the papers.

First group of	Assembly*, Assembly system, Line*, Straight line, U shape, U		
keywords	shape*		
Second group of	Augmented reality, Cloud computing, Collaborative robo*, Data		
keywords	analysi*, Internet of thing, Assembly*, Assembly time, Quality, *		
	time, Products quality, Products variety		

Table 1. The two groups of keywords used in the research

These keywords were used as a single keyword or, most of the time, as a combination of two, or more, words thanks to the operator "AND". For example, one combined research can be: "Assembly*" AND "Collaborative robo*".

3 Factor Influencing the Configuration Selection

In this section with an inductive approach, we derived the relevant factors that a company has to know when it has to decide which is the best configuration for their AS between straight-line and U-shaped line. Here below there is the list of identified factors followed by an analysis of the selected papers which helped us to derived them.

Cycle Time: indicates the time elapsed between completions of two consequent units.

Number of Tasks: it is the number of different activities that each operator must do in each workstation and is a measure used to indicate the problem size for combinatorial problems such as the SALB (Simple assembly line balancing) and UALB (U-shaped assembly line balancing) problems.

Network Density: calculating network density entails dividing the number of actual precedence relationships by the theoretical maximum number of relationships that could exist for a problem of that size. Extreme network density values of 0.0 and 1.0 correspond to flexible and rigid assembly sequences, respectively [8].

Products Variety: is the number of different products that are assembled in an assembly line.

Volumes Variety: is the difference between the quantity to produce of a product one day and the quantity to produce of the same product the following day.

Quality: is the quality of the final assemble products that come out from the assembly line.

Complexity of Tasks: is the complexity of each task of the assembly process.

Following the definitions here above introduced, the papers found by the literature are here presented in order to better explain the different factors.

In straight-line it is not easy as in U-shaped line to adapt to changes in cycle time. This because of the high potential of rebalancing the U-shaped line with a new cycle time and the reallocation of workers [6]. The productivity in straight-line is lower than the one of U-shaped line when network density is high. Indeed, when the network density is low is better to use straight-lines [8]. Straight-lines can guarantee less flexibility in manufacturing volumes respect to U-shaped lines since in U-shaped lines you can increase or decrease the number of operators on the line [15] and the quality of final assembled products is usually better in U-shape lines [2]. The quality in U-shaped is higher thanks to the improved visibility and communication between operators on the line, which facilitates problem-solving and quality improvement [16]. In U-shaped lines usually, a human worker does more tasks than one in a straight-line. As operator responsibilities expand to include more assembly tasks, operators are used more efficiently when using a straight-line configuration [8]. This because more are the different assembly tasks that one human worker must remember and higher will be his mental workload. When a human worker has to walk from a workstation to another one to do his work, the things to keep in mind are not only the time lost by the worker to walk and his physical strain but there is also the possible congestion that can be generate in the line by all the workers that have to move [4, 8]. This congestion can be more problematic in the U-shaped line due to the limited space within the line generating a lost in performance. The walking time of human workers not only influence the physical demand of workers and the congestion of the lines but adding it at the processing time significantly increases the number of workers that are requested in order to execute the assembly tasks [17]. U-shaped lines are a good choice when the workers have to do repetitive actions and avoid keeping wrong postures during the execution of their activities [18]. Moreover, the U-shape lines are more flexible than the straight-lines, balancing by reducing the precedence relations thanks also to the multiskilled workers that perform various tasks in different workstations along the assembly line [19]. Ushaped workers are required to possess more skills than on straight-lines [20].

In order to decide if a configuration is better than another one companies need Key Performance Indicators (KPIs) or objective functions to measure (e.g. costs, flexibility, ergonomics, productivity). Therefore, companies need first to understand how these functions are going to be influenced by the factors and then how to do multi objectives analysis. It will be important to define one or more objective functions and see how these factors impact the decision of the AS configuration. Future works can be indeed related to the definition of these objective functions. Moreover, sensitivity analyses can help to see how the variations of the different factors can change the final decision of which configuration to choose.

4 The Impact of Industry 4.0 Technologies

In this section we are going to investigate the literature to understand how the I4.0 technologies impact the factors defined in the previous section. Collaborative robots do not only improve the performance of the assembly lines [21], but they can also help the human workers in reducing potential occupational risks like awkward postures, mental workload and repetitive actions [22]. This means better ergonomic conditions in the assembly lines. Moreover, collaborative robots can reduce the surface used and so decrease the distances traveled by the human workers [23]. The main improvements that AR can give to assembly operations are the errors reduction and time efficiency. However, researchers have different opinions about these two improvements [24, 25]. There are different opinions also in terms of usability of AR technology. [26] showed how for the participants of their experiments were easier to use paper instructions instead of the AR instructions displayed through Head-mounted display. In [24], experiment's participants instead, in terms of cognitive load, usability and performance, preferred the AR-based instruction than the paper ones. IoT technologies can help companies with the management of the different variants and of the different volumes of products to produce since the consumers can change their orders until the last second and different consumers have different requests [27]. The data collected from the IoT technologies can be analyzed thanks to Data analysis techniques, like machine learning algorithms, and together with the Cloud Computing, that can store data, manage data, and share data and information, the quality of the final products can be guaranteed reducing also the time to do the inspection of the products [28]. These technologies cannot only increase the quality of the final products but can also decrease the amount of scrap parts in a real industrial scenario, consequently saving valuable resources such as energy and raw materials [29] increasing the efficiency of the systems [30].

In Table 2 we can see which are the I4.0 technologies applied in the AS and we summarized the effect of each technology in each factor.

From Table 2 one interesting result that we can see is that there is a technology, AR, that have "Decreases/Increases" as result. This result means that is not well known yet the real impact of this technology with respect to the cycle time. In fact, at the moment the research with this technology regarding this factor are showing different results. This can be related for example to the experience of the human workers in using the technology or on how the AR instructions are designed and presented to the human workers.

Based on the results of Table 2 we created Table 3. Table 3 gives the idea of how the choice of the configuration can change based on the decrease or increase of the factors. In Table 3 we also did the column, based on the review of the literature in the previous section, of the case in which there is no I4.0 technology in the AS.0. This column helps to see that an I4.0 technology can open the opportunity to move to another configuration with respect to the case without technology.

	Industry 4.0 technology				
Factor	Collaborative	Augmented	IoT	Cloud computing and	
	robot	reality	technologies	data analysis	
Cycle time	Decreases	Decreases/	Decreases	Decreases	
		Increases			
Number of tasks	/	/	/	/	
Network density	/	/	/	/	
Products variety	Increases	Increases	Increases	Increases	
Volumes variety	Increases	/	Increases	Increases	
Quality	Increases	Increases	Increases	Increases	
Complexity of	Decreases	Decreases	Decreases	Decreases	
tasks					

Table 2. Impact of the I4.0 technologies on the factors.

Table 3. How the choice of the AS configuration change based on the impact of a I4.0 technology on a factor (U = U-shaped; S = Straight-line; / = not found in literature).

			Industry 4.0 technology				
Factor		Without I4.0 technology	Collaborative robot	Augmented reality	IoT technologies	Cloud computing and data analysis	
Cycle time	Low	S	S	/	S	S	
	Medium	U/S	U		U	U	
	High	U					
Number of	Low	U	/	/	/	/	
tasks	Medium	U/S					
	High	S					
Network	Low	S	/	/	/	/	
density	Medium	U/S					
	High	U					
Products variety	Low	S	S	S	S	S	
	Medium	U/S					
	High	U	U	U	U	U	
Volumes variety	Low	S	/	S	S	S	
	Medium	U/S					
	High	U		U	U	U	
Quality	Low	S	S	S	S	S	
	Medium	U/S					
	High	U	U	U	U	U	
Complexity of tasks	Low	S	S	S	S	S	
	Medium	U/S					
	High	U	U	U	U	U	

From Table 3 we can see that, for example, when the quality of the final products increases thanks to the collaborative robots from low to medium, the choice of the configuration of the AS can go to the straight-line configuration instead of the U-shaped line configuration. Indeed, the increase in quality thanks to the collaborative robot gives the opportunity to use the straight-line configuration that usually, without the technology, guarantees a lower quality with respect to the U-shaped line configuration. Of course, it is not only the focus on one factor that defines the choice of the configuration of the AS4.0. Therefore, the combined behavior of these factors with regard to the choice of the configuration can be a matter of interesting future research. Moreover, it will be interesting to do a deeper investigation of the literature to see what it is possible to find about the cases that in Table 3 are now presenting the "/". Especially for all the technologies in the case of the factor "Number of tasks", for the technology "Cloud computing and Data Analysis" for the factor "Network density" and for the technology "Collaborative robot" for the factor "Volumes variety".

5 Conclusion

In this paper, we first determined, based on the literature, the factors that a company has to know when it is time to decide which is the best AS configuration to adopt between the straight-line and the U-shaped line and second we saw how the Industry 4.0 technologies, that can be adopted in these AS configurations, impact in these factors. Being this an exploratory research, our conclusions need to be support by some quantitative results. This is the reason why this work will be used as a starting point for the creation of a mathematical model that will help us validate the assumptions made here.

References

- Battini, D., Faccio, M., Persona, A., Sgarbossa, F.: Balancing-sequencing procedure for a mixed model assembly system in case of finite buffer capacity. Int. J. Adv. Manuf. Technol. 44(3–4), 345–359 (2009)
- Rabbani, M., Moghaddam, M., Manavizadeh, N.: Balancing of mixed-model two-sided assembly lines with multiple U-shaped layout. Int. J. Adv. Manuf. Technol. 59(9–12), 1191– 1210 (2012)
- Mukund Nilakantan, J., Ponnambalam, S.G.: Robotic U-shaped assembly line balancing using particle swarm optimization. Eng. Optim. 48(2), 231–252 (2016)
- Calzavara, M., Faccio, M., Persona, A., Zennaro, I.: Walking worker vs fixed worker assembly considering the impact of components exposure on assembly time and energy expenditure. Int. J. Adv. Manuf. Technol. 112(9), 2971–2988 (2021)
- 5. Chen, Q., Liao, S., Wu, Z., Yi, S.: Comparative analysis of the performance of a novel U-shaped 'chasing-overtaking' production line. Int. J. Prod. Res. **54**(12), 3677–3690 (2016)
- Erel, E., Sabuncuoglu, I., Aksu, B.A.: Balancing of U-type assembly systems using simulated annealing. Int. J. Prod. Res. 39(13), 3003–3015 (2001)

- Battaïa, O., Otto, A., Sgarbossa, F., Pesch, E.: Future trends in management and operation of assembly systems: from customized assembly systems to cyber-physical systems. Omega 78, 1–4 (2018)
- Aase, G.R., Olson, J.R., Schniederjans, M.J.: U-shaped assembly line layouts and their impact on labor productivity: an experimental study. Eur. J. Oper. Res. 156(3), 698–711 (2004). https://doi.org/10.1016/S0377-2217(03)00148-6
- Boysen, N., Fliedner, M., Scholl, A.: A classification of assembly line balancing problems. Eur. J. Oper. Res. 183(2), 674–693 (2007)
- Thames, L., Schaefer, D.: Software-defined cloud manufacturing for industry 4.0. Procedia cirp. 52, 12–17 (2016)
- 11. Kagermann, H., Helbig, J., Hellinger, A., Wahlster, W.: Recommendations for implementing the strategic initiative INDUSTRIE 4.0: securing the future of German manufacturing industry; final report of the Industrie 4.0 working group. Forschungsunion (2013)
- Zhong, R.Y., Xu, X., Klotz, E., Newman, S.T.: Intelligent manufacturing in the context of industry 4.0: a review. Engineering 3(5), 616–630 (2017)
- Cohen, Y., Naseraldin, H., Chaudhuri, A., Pilati, F.: Assembly systems in industry 4.0 era: a road map to understand assembly 4.0. Int. J. Adv. Manuf. Technol. 105(9), 4037–4054 (2019). https://doi.org/10.1007/s00170-019-04203-1
- Battini, D., Faccio, M., Persona, A., Sgarbossa, F.: New methodological framework to improve productivity and ergonomics in assembly system design. Int. J. Ind. Ergon. 41(1), 30–42 (2011)
- Şahin, M., Kellegöz, T.: Increasing production rate in U-type assembly lines with sequencedependent set-up times. Eng. Optim. 49(8), 1401–1419 (2017)
- Guerriero, F., Miltenburg, J.: The stochastic U-line balancing problem. Naval Res. Logist. (NRL) 50(1), 31–57 (2003)
- 17. Sirovetnukul, R., Chutima, P.: The impact of walking time on U-shaped assembly line worker allocation problems. Eng. J. **14**(2), 53–78 (2010)
- Zhang, Z., Tang, Q., Ruiz, R., Zhang, L.: Ergonomic risk and cycle time minimization for the U-shaped worker assignment assembly line balancing problem: a multi-objective approach. Comput. Oper. Res. 118, 104905 (2020)
- 19. Tiacci, L.: The assembly line simulator project: modeling U-shaped, un-paced, mixed model assembly lines. XIX Summer School Impianti Industriali, The role of industrial engineering in a global sustainable economy, Senigallia (AN–ITA) (2014)
- 20. Hwang, R., Katayama, H.: A multi-decision genetic approach for workload balancing of mixed-model U-shaped assembly line systems. Int. J. Prod. Res. 47(14), 3797–3822 (2009)
- Bloss, R.: Collaborative robots are rapidly providing major improvements in productivity, safety, programing ease, portability and cost while addressing many new applications. Ind. Robot Int. J. (2016)
- Mateus, J.C., Claeys, D., Limère, V., Cottyn, J., Aghezzaf, E.-H.: A structured methodology for the design of a human-robot collaborative assembly workplace. Int. J. Adv. Manuf. Technol. **102**(5–8), 2663–2681 (2019). https://doi.org/10.1007/s00170-019-03356-3
- Gil-Vilda, F., Sune, A., Yagüe-Fabra, J.A., Crespo, C., Serrano, H.: Integration of a collaborative robot in a U-shaped production line: a real case study. Procedia Manuf. 13, 109–115 (2017)
- Lampen, E., Teuber, J., Gaisbauer, F., Bär, T., Pfeiffe, T., Wachsmuth, S.: Combining simulation and augmented reality methods for enhanced worker assistance in manual assembly. Procedia CIRP 81, 588 (2019)
- 25. Longo, F., Nicoletti, L., Padovano, A.: Smart operators in industry 4.0: a human-centered approach to enhance operators' capabilities and competencies within the new smart factory context. Comput. Ind. Eng. **113**, 144–159 (2017)
- Büttner, S., Funk, M., Sand, O., Röcker, C.: Using head-mounted displays and in-situ projection for assistive systems: a comparison. In: Proceedings of the 9th ACM international conference on pervasive technologies related to assistive environments, pp. 1–8 (2016)
- 27. Thramboulidis, K., Kontou, I., Vachtsevanou, D.C.: Towards an IoT-based framework for evolvable assembly systems. IFAC-PapersOnLine **51**(11), 182–187 (2018)
- Schmitt, J., Bönig, J., Borggräfe, T., Beitinger, G., Deuse, J.: Predictive model-based quality inspection using machine learning and edge cloud computing. Adv. Eng. Inf. 45, 101101 (2020)
- Coupek, D., Lechler, A., Verl, A.: Cloud-based control strategy: downstream defect reduction in the production of electric motors. IEEE Trans. Ind. Appl. 53(6), 5348–5353 (2017)
- Carvajal Soto, J.A., Tavakolizadeh, F., Gyulai, D.: An online machine learning framework for early detection of product failures in an industry 4.0 context. Int. J. Comput. Integr. Manuf. 32(4–5), 452–465 (2019)

Paper 4

Simonetto, M., Peron, M., Fragapane, G. and Sgarbossa, F., 2020, October. Digital assembly assistance system in Industry 4.0 era: a case study with projected augmented reality. In International Workshop of Advanced Manufacturing and Automation (pp. 644-651). Springer, Singapore.

This is an accepted manuscript of an article published in International Workshop of Advanced Manufacturing and Automation.

The final authenticated version is available online at: https://doi.org/ 10.1007/978-981-33-6318-2_80



Digital Assembly Assistance System in Industry 4.0 Era: A Case Study with Projected Augmented Reality

Marco Simonetto^(⊠), Mirco Peron, Giuseppe Fragapane, and Fabio Sgarbossa

Department of Mechanical and Industrial Engineering, Norwegian University of Science and Technology, S P Andersens V 3, 7031 Trondheim, Norway marco.simonetto@ntnu.no

Abstract. Automation has increased more and more in manufacturing companies over the last decades. However, manual labor is still used in a variety of complex tasks and is currently irreplaceable, especially in assembly operations. Assisting and supporting the human worker during potentially complex assembly tasks is very relevant. Clear and easy-to-read assembly instructions, error-proofing methods, and an intuitive user interface for the worker have the potential to not only reduce the cognitive workload of the operator but also increase the productivity, improve the quality, reduce defects, and consequently reduce costs. Industry 4.0 technologies, in particular Augmented Reality and motion recognition sensors, can help companies in reaching these goals. However, there are currently only a few works that show how to implement these technologies with real case studies, especially with the Projected Augmented Reality (PAR). This is the reason why, in this paper an example of prototype of a smart workstation equipped with a Kinect-projector assistance system for manual assembly is presented.

Keywords: Digitalization \cdot Assembly \cdot Industry 4.0 \cdot Projector \cdot Augmented reality \cdot Case study

1 Introduction

In the las few years, manufacturers are dealing with an increasingly demand for customized products [1]. With such a market demand, the variety of products is increasing, dictating the necessity for more flexible and adaptable assembly lines in order to be responsive to the needs of individuals [2]. Zhong and Ai [3] defined assembly lines as continuous production lines, consisting of materials and workstations combined by conveyor belts, contacting workers and machines closely and efficiently. These represent the last phase of production processes and are the place where the products are customized [4].

In assembly lines activities are still performed mainly by human operators due to their characteristics like flexibility, adaptability, decision-making skills, and creativity [5]. Despite these qualities, humans workers can compromise the performance of an assembly system due to the over-workload during the execution of the different

assembly tasks [6]. This over-workload can be high since the assembly tasks are growing in number due to the increased products varieties and they are getting highly demanded in terms of required skills [7].

In order to help human workers to reduce their workload during the executions of assembly tasks, new solutions have started being introduced thanks to the advent of Industry 4.0. Besides their effects on the human workload, these technologies can also increase the performance of the assembly lines and the quality of the products. An example of such solutions are the Augmented Reality (AR) instructions [8]. These AR instructions compared to the traditional ones, such as paper-based and computerassisted instructions [9, 10], that can be used as assistance system to guide the human workers during the execution of assembly activities seems to be very promising [11]. Augmented Reality instructions can be presented to human workers through a screen, like a smart phone [12] or a tablet [13], through head-mounted devices, like smart glasses, or through images projected directly in the working area [14]. In the paper we will refer to this last possibility as Projected Augmented Reality (PAR). Despite the high attention that AR solutions have attracted in the last years among researchers and practitioners, the majority of these works are qualitative studies. In a perspective of implementation of such solutions, it is required to understand which is their impact on the execution of assembly activities in terms of task completion time, quality and mental workload. However, very few papers present quantitative studies, especially when dealing with PAR, despite the high interest that is recently gaining among companies. To fill this gap, in this paper, we presented a case study of a prototype of PAR solution, comparing its performances in terms of task completion time, quality, and mental workload with those of paper-based and computer-assisted instructions. The remainder of this paper is structured as follows. Section 2 provides a review of the existing literature on the AR solutions used to assist human workers. Section 3 introduces our case study, while Sect. 4 illustrates the results obtained from the case study. Finally, in Sect. 5 the results are discussed and the conclusions are drwan.

2 Literature Review

In this section the results of a literature review carried out using Scopus database as search engine and limiting the research from January 2005 to August 2020 and English as language. Due to the need and importance of understanding and quantifying the impact of AR solutions on assembly operations, we considered only the papers reporting quantitative results.

Värno et al. [15] used AR to present the instructions to the assembly workers through a tablet. The parameters studied during the tests are the assembly time and the errors committed during the execution of the assembly tasks. The presented prototype is compared with paper-based instructions and resulted in better performing with respect to both the parameters. Koning et al. [16] presented an AR assistance system, with smart glasses and gloves, intended to assist an inexperienced worker during a manual assembly. They validate their solution through a user test that proved that the system is suitable for assisting in the manual assembly, reducing the number of errors respect to the case where AR is not used. Moreover, it provided a learning effect to its users while

also offering a positive user experience. Two experiments with 50 participants were conducted in [9] to compared an animated AR system and the paper-based manual system. They studied the task completion time, the number of errors and the cognitive workload. The results of the experiments revealed how the AR system yielded better performances than the paper-based manual with respect to all the three considered parameters. The same parameters were also considered by Hou, Wang and Truijens [17], where the authors investigated how much improvement in assembly productivity and performance can be achieved by lowering cognitive workload via AR. AR instructions are presented through a screen and they are confronted with isometric drawings. A prototype developed based on the Ocolus Rift platform is presented and compared to paper-based instructions in Syberfeldt et al. [14]. Contrarily to other works, paper-based instructions were characterized by better feedback from the human workers resulted in lower assembly time. The authors stated that this was due to the fact that the AR solution was perceived complex by the operators. A Cognition-based interactive Augmented Reality Assembly Guidance System (CARAGS) is bench marched against a LCD screen-based digital documentation and a traditional AR assembly guidance system in [18]. After the experiments, CARAGS has been shown to be the most intuitive, easy to use, and satisfactory guidance system among the three guidance systems based on the average values. It also gives the opportunities to complete the assembly tasks in less time respect to the other two solutions. The results in [19] demonstrated how the presented AR solution provides a more mental-relaxed solution to support the operators in an assembly task. This, even if the AR solution less performed in terms of assembly time with respect to the pape-based instructions, can bring an important advantage considering the highly customized products that are demanded now a day. Two different sets of visual features through AR, concrete AR (CAR) and abstract AR (AAR) are presented and compared with paper-based instructions in [20]. The experiments showed how AAR had the highest average error rates, as well as the longest average completion time, and that the paper-based instructions obtained the best results. However, both the AR solutions help operators in being more confident in performing assembly tasks. A system with AR together with data analytics is introduced by Lai et al. [21]. The experimental results showed how this system, compared to the paper-based instructions, can reduce the assembly completion time and the number of errors committed during the assembly activities. Performance, ease of use and acceptance of two AR-based methods (mobile and spatial AR), are compared in [22]. The results of the experiments demonstrated how participants were faster and made fewer error using the spatial AR solution. This solution gave the opportunity to project the assembly information of a puzzle directly on the workplace where the assembly activities were executed. The findings of Vanneste et al. [23] suggested that projection-based AR has the potential to cognitively support operators during assembly tasks and can hence contribute to better quality, a lower stress level, a higher degree of independence and a lower perceived complexity. A work with projection-based AR is presented in [24]. Their tests indicate that this system is perceived as more helpful and more engaging compared with a similar system using AR smart glasses or an assembly station without assistance. However, they do not show any numerical results that demonstrate their results. Two examples of projected AR solutions to help impaired workers are introduced in [25, 26]. In both, the

authors compared different solutions and showed how AR is able to help the impaired workers during the executions of the assembly tasks.

From the literature it emerges two facts. First, that AR instructions can be beneficial for the performance of an assembly line, and, second, that PAR solutions are scarcely investigated with quantitative studies. PAR can overcome the other AR instructions performances for example giving the opportunity to the human workers to not wear, or grab, any kind of devices to perform the assembly activities. These devices, like the smart glasses for example, can be uncomfortable and can create blind spots even for people that wear glasses daily [7]. These reasons move us to create in the Logistic4.0 Lab the prototype of a smart workstation equipped with a Kinect-projector assistance system for manual assembly presented, with a case study, in the next chapter.

3 Case Study Description

To assess the potential of the assembly assistance system in terms of productivity and mental workload of the workers, the Logistics 4.0 Laboratory at the Department of Mechanical and Industrial Engineering at the Norwegian University of Science and Technology developed a smart assembly workstation. The smart assembly workstation consists of a Kinect-camera, motion recognition software, work-instruction program and a projector. While the cameras of the Kinect and the motion recognition software track the movement and trigger action in the work-instruction program, the projector can visualize necessary information for the worker to pick and assemble a product (Fig. 1). This innovative, assistive work assembly process can be classified as Projected Augmented Reality (PAR). In most production environments paper-based manuals and workplace-mounted monitors provide work instructions. Therefore, the PAR was compared to paper-based and computer-assisted instruction method in times of productivity and mental workload.



Fig. 1. The smart workstation with Projected Augmented Reality

The test population consisted of 15 participants, who carried out a series of assembly tests with different workstation configuration in the Logistics 4.0 Laboratory.

The participants must assemble a LEGO model following the provided assembly information.

In this paper, due to pages restriction, we are going to show only the results obtained from the assembly of two components (Front wing and Side pod, Fig. 2) of a LEGO product These two components are characterized by different degrees of complexity: the Front wing has a lower degree of complexity (3 different parts, 4 work steps), while the Side pod has a higher degree of complexity (8 different parts, 8 work steps).



Fig. 2. The Front wing (on the left) and the Side pod (on the right)

4 Results

In the experiment, we the first measured the time for completing the assembly tasks and the number of errors performed during the assembly. The results allowed to assess the performance and accuracy of the assistance system prototype. In the second part, after completing the assembly tests at the smart workstation, were asked to the participants of the experiment to fill out an questionnaire.

The computer-assisted instructions achieved the best results in the Task Completion Times compared to the other alternatives (see Fig. 3, left). The "Front wing" was assembled in 29.98 s with SD = 5.35 s and the "Side pod" in 46.19 s with SD of 6.79 s.



Fig. 3. Task Completion Time (left) and number of errors (right) of the two assembled components

The quality of the assembly activities were measured by counting the number of assembly and picking errors (see Fig. 3, right). The results show that the PAR system can reduce the possibility of committed errors during the especially for the more complex assembly task of the "Side pod".

The mental workload was measured thanks to the compilation of a simplified NASA-TLX questionnaire. The questionnaire asked the participants to give feedback on perceived enjoyment, frustration, perceived ease of use, effort, and mental demand for each work instruction method. The PAR system obtained the overall best results and scored especially high in perceived enjoinment (86.7%) and ease of use (86%).



Fig. 4. Simplified NASA-TLX

5 Discussion and Conclusion

In this paper, a prototype of a PAR system has been developed in order to quantify the benefits that an assembly line can obtain from its implementation. The results of the case study suggest that a PAR assistance system has several advantages over the paperbased and computers-assisted methods. Although PAR did not lead to the lowest Tasks Completion Time, Fig. 3, the use of this solution can still be beneficial in situations where workers have to wear gloves or they have to use tools during the execution of the assembly tasks since with PAR they do not have to drag their finger on a touch screen or to move paper pages in order to move to the next assembly task. These operations can be problematic when workers have to wear gloves or they have to use tools during the execution of the assembly tasks. The PAR system can guarantee a higher quality of the final products since the numbers of mistakes made by the human workers are lower, Fig. 3. This means fewer products that need to be reassembled with a saving in time and money. Moreover, the results from the questionnaires showed how a PAR systems leads workers to do their work with less effort then the other two solutions, Fig. 4. In such a way, workers' health can be preserved resulting in a higher productivity thanks to the reduced absenteeism related to sickness leaves.

Despite the just mentioned advantages, our solution is limited by some limitations, such as the inability to detect the correctness of the assembly operations. Another limitation that hinders our PAR from being the fastest solution is the fact that each task after being executes need to be confirmed, since there is not a system that can detect the execution of the assembly process step by step and understand when a task is over and the next has to start.

However, despite the two mentioned limitations that need to be solved in the future development of the system, the benefits provided by the PAR system are several. In order to study in more detail these benefits, as possible future work, our solutions can be tested in a real industry case study. Another future work can be the study of our solution in a business case point of view, comparing, for example, the costs of its implementation and maintenance with the same costs of the other two studied solutions in the case study (paper-based and computer-assisted instructions).

References

- Battaïa, O., Otto, A., Sgarbossa, F., Pesch, E.: Future trends in management and operation of assembly systems: from customized assembly systems to cyber-physical systems. Omega (U. K.) 78, 1–4 (2018)
- Hu, S.J., Ko, J., Weyand, L., Elmaraghy, H.A., Lien, T.K., Koren, Y., et al.: Assembly system design and operations for product variety. CIRP Ann. – Manuf. Technol. 60(2), 715– 733 (2011)
- 3. Zhong, Y.-G., Ai, B.: A modified ant colony optimization algorithm for multi-objective assembly line balancing. Soft Comput. **21**(22), 6881–6894 (2017)
- Akpinar, S., Elmi, A., Bektaş, T.: Combinatorial Benders cuts for assembly line balancing problems with setups. Eur. J. Oper. Res. 259(2), 527–537 (2017)
- Michalos, G., Makris, S., Spiliotopoulos, J., Misios, I., Tsarouchi, P., Chryssolouris, G.: ROBO-PARTNER: seamless human-robot cooperation for intelligent, flexible and safe operations in the assembly factories of the future. Procedia CIRP [Internet] 23(C), 71–76 (2014). http://dx.doi.org/10.1016/j.procir.2014.10.079
- Battini, D., Calzavara, M., Otto, A., Sgarbossa, F.: Preventing ergonomic risks with integrated planning on assembly line balancing and parts feeding. Int. J. Prod. Res. [Internet] 55(24), 7452–7472. (2017). https://doi.org/10.1080/00207543.2017.1363427
- Joundi, J., Conradie, P., van Den Bergh, J., Saldien, J.: Understanding and exploring operator needs in mixed model assembly. CEUR Workshop Proc. 2503, 130–138 (2019)
- Ong, S.K., Yuan, M.L., Nee, A.Y.C.: Augmented reality applications in manufacturing: a survey. Int. J. Prod. Res. 46(10), 2707–2742 (2008)
- Hou, L., Wang, X., Bernold, L., Love, P.E.D.: Using animated augmented reality to cognitively guide assembly. J. Comput. Civ. Eng. 27(5), 439–451 (2013)
- Watson, G., Curran, R., Butterfield, J., Craig, C.: The effect of using animated work instructions over text and static graphics when performing a small scale engineering assembly. In: Collaborative Product and Service Life Cycle Management for a Sustainable World – Proceedings of the 15th ISPE International Conferences Concurr Engineering CE 2008, vol. 44, no. 0, pp. 541–550 (2008)
- 11. Van Krevelen, D.W.F., Poelman, R.: A survey of augmented reality technologies, applications and limitations. Int. J. Virtual Real. 9(2), 1–20 (2010)

- Bagamaspad, A.M., Caingles, J.I., Koa, K.A., Simeon, J.L., Ruiz, C.: A mobile augmented reality-based assembly guidance application for LEGO. In: MCCSIS 2018 - Multi Conference on Computer Science and Information Systems. Proceedings of the International Conferences Interfaces Human Computer Interacting 2018, Game Entertainment Technology 2018 Computer Graph Visual Comparative 2018, pp. 237–244, July 2018
- 13. Servan, J., Mas, F., Menéndez, J.L., Ríos, J.: Assembly work instruction deployment using augmented reality. Key Eng. Mater. **502**(February), 25–30 (2012)
- Syberfeldt, A., Danielsson, O., Holm, M., Wang, L.: Visual assembling guidance using augmented reality. Procedia Manuf. [Internet] 1, 98–109 (2015). http://dx.doi.org/10.1016/j. promfg.2015.09.068
- 15. Värno, K., Mahmood, K., Otto, T., Kuts, V.: Development of a smart workstation by using ar technology. Ann. DAAAM Proc. Int. DAAAM Symp. **30**(1), 1061–1067 (2019)
- Konig, M., Stadlmaier, M., Rusch, T., Sochor, R., Merkel, L., Braunreuther, S., et al.: MA2RA - manual assembly augmented reality assistant. In: IEEE International Conference on Industrial Engineering and Engineering Management, pp. 501–505 (2019)
- 17. Hou, L., Wang, X., Truijens, M.: Using augmented reality to facilitate piping assembly: an experiment-based evaluation. J. Comput. Civ. Eng. **29**(1), 1–12 (2015)
- Wang, X., Ong, S.K., Nee, A.Y.C.: Multi-modal augmented-reality assembly guidance based on bare-hand interface. Adv. Eng. Inform. [Internet] 30(3), 406–421 (2016). http://dx. doi.org/10.1016/j.aei.2016.05.004
- Mourtzis, D., Zogopoulos, V., Xanthi, F.: Augmented reality application to support the assembly of highly customized products and to adapt to production re-scheduling. Int. J. Adv. Manuf. Technol. 105(9), 3899–3910 (2019)
- Radkowski, R., Herrema, J., Oliver, J.: Augmented reality-based manual assembly support with visual features for different degrees of difficulty. Int. J. Hum. Comput. Interact. [Internet] 31(5), 337–349 (2015). http://dx.doi.org/10.1080/10447318.2014.994194
- Lai, Z.H., Tao, W., Leu, M.C., Yin, Z.: Smart augmented reality instructional system for mechanical assembly towards worker-centered intelligent manufacturing. J. Manuf. Syst. [Internet] 55(July 2019), 69–81 (2020). https://doi.org/10.1016/j.jmsy.2020.02.010
- Alves, J., Marques, B., Oliveira, M., Araújo, T., Dias, P., Santos, B.S.: Comparing spatial and mobile augmented reality for guiding assembling procedures with task validation. In: 19th IEEE International Conference on Autonomous Robot Systems and Competitions ICARSC 2019 (2019)
- Vanneste, P., Huang, Y., Park, J.Y., Cornillie, F., Decloedt, B., Van den Noortgate, W.: Cognitive support for assembly operations by means of augmented reality: an exploratory study. Int. J. Hum. Comput. Stud. [Internet] 143(October 2019), 102480 (2020). https://doi. org/10.1016/j.ijhcs.2020.102480
- Büttner, S., Sand, O., Röcker, C.: Extending the design space in industrial manufacturing through mobile projection. In: Mobile HCI 2015 - Proceedings of the 17th International Conference on Human-Computer Interaction with Mobile Devices and Services Adjunct, pp. 1130–1133 (2015)
- 25. Korn, O., Schmidt, A., Hörz, T.: Augmented manufacturing: a study with impaired persons on assistive systems using in-situ projection. In: ACM International Conference Proceeding Ser (2013)
- 26. Funk, M., Bächler, A., Bächler, L., Korn, O., Krieger, C., Heidenreich, T., et al.: Comparing projected in-situ feedback at the manual assembly workplace with impaired workers. In: Proceedings of the 8th ACM International Conference on PErvasive Technologies Related to Assistive Environments PETRA (2015)

Paper 5

Simonetto, M., Arena, S. and Peron, M., 2022. A methodological framework to integrate motion capture system and Virtual Reality for assembly system 4.0 workplace design. Safety Science, 146, p.105561.

This is an accepted manuscript of an article published in Safety Science.

The final authenticated version is available online at: https://doi.org/ <u>10.1016/j.ssci.2021.105561</u>

Safety Science 146 (2022) 105561

Contents lists available at ScienceDirect



Safety Science

journal homepage: www.elsevier.com/locate/safety



A methodological framework to integrate motion capture system and virtual reality for assembly system 4.0 workplace design



Marco Simonetto^{a,*}, Simone Arena^b, Mirco Peron^a

^a Department of Mechanical and Industrial Engineering, Norwegian University of Science and Technology, S P Andersens V 3, 7031 Trondheim, Norway ^b Department of Mechanical, Chemical and Materials Engineering, University of Cagliari, Via Marengo, 2, Cagliari 09123, Italy

ARTICLE INFO

Keywords: Assembly Workplace Design Industry 4.0 Methodological framework Ageing

ABSTRACT

Nowadays new products are required more and more often. Since a certain product is assembled only on a specific Assembly System (AS), a new AS has to be designed every time a new product is developed. Similarly, since an AS is constituted by one or more workplaces, new workplaces need to be designed every time a new product is developed. However, this is not feasible considering the time- and resource-consuming AS workplace design procedures currently used (e.g., physical mock-ups and computer-aided systems). New solutions have thus emerged to accelerate the AS workplace design procedures, especially since the advent of Industry 4.0 (I4.0) technologies. Specifically, the combined use of motion capture (mocap) systems and Virtual Reality (VR) has been considered very promising, with many researchers showing its potential. However, to the best of the authors' knowledge, none of them suggests a clear methodology to follow when designing AS workplaces using the mocap system and VR. In this paper, we thus aim to fill this gap by developing a methodological framework that describes in detail the different steps to be followed. Moreover, the methodological framework has been developed in such a way that both productivity and Occupational Safety and Health (OSH) considerations are included. Furthermore, it encompasses the current ageing workforce scenario by explicitly including the ageing workforce's main characteristics (reduced flexibility and strength and greater experience of older operators). A simple but representative case study has then been carried out to demonstrate how to use the methodological framework and to prove its validity.

1. Introduction

Nowadays customers require new and customised products more and more often, and companies have hence to be able to easily follow these requirements in order to survive in today's competitive market. Companies thus have to be able to change their Assembly Systems (ASs) in a fast and economic way to cope with the short life-time of the products, i. e., Assembly Systems (ASs) are required to be flexible and reconfigurable (Battaïa et al., 2018). This, together with the optimisation of the AS workplaces (defined according to Cavatorta and Dipardo (2009) as (i) the workstation where the assembly tasks are executed, (ii) all the equipment that is necessary to execute the tasks, and (iii) all the systems necessary to store the components that are not convenient to store directly in the workstation), is fundamental to assemble new products at low costs, short time to market, and high reliability of deliveries (Battnii et al., 2011; Holubek and Ružarovský, 2014; Porta et al., 2020, Arena et al., 2021). Moreover, due to the presence of human operators in AS workplaces, their optimisation has to be carried out with respect not only to productivity but also to Occupational Safety and Health (OSH). Human operators are, in fact, fundamental to guarantee the high flexibility required by ASs (Makris et al., 2016), and it is thus important to promote their comfort to improve their OSH, especially their wellbeing. This can be achieved by considering the right ergonomic interventions during the workplace design process. These interventions can involve some adjustments in the AS workplace (Kim and Junggi Hong, 2013) and it is crucial that they also consider the current ageing of the workforce (OECD, 2015). Indeed, the ageing of the operators could have a negative impact on the AS productivity (Strasser, 2018): assembly tasks require physical efforts to be executed, but ageing is known to reduce physical capabilities. For example, Peng and Chan (2019) reported that the musculoskeletal capacity decreases by 20% after the age of 60. Therefore, it is necessary to limit the drawbacks associated with the decreased physical capabilities of the ageing workforce, and this can be done with the proper design of the AS workplace (Truxillo et al., 2015;

* Corresponding author.

https://doi.org/10.1016/j.ssci.2021.105561

Received 24 May 2021; Received in revised form 15 October 2021; Accepted 21 October 2021

0925-7535/© 2021 The Author(s). Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

E-mail address: marco.simonetto@ntnu.no (M. Simonetto).

Calzavara et al., 2020). In fact, an effective workplace can lead to a decrease in the physical burden of the human operators, thus reducing the ergonomic risks and improving the OSH (Roper and Yeh, 2007). This, in turn, results in increased productivity, better working conditions, and operators' wellbeing in an AS (Eswaramoorthi et al., 2010).

The design of these workplaces can be facilitated by the advent of Industry 4.0 and its technologies (Burggräf et al., 2019). The term Industry 4.0 is used to indicate the fourth industrial revolution that the industrial environment is currently experiencing (Wang, 2016). This revolution is moving companies toward a gradual and constant automation of traditional manufacturing practices (Schwab, 2016) and ASs are involved in this revolution, introducing the concept of Assembly System 4.0 (AS4.0) (Bortolini et al., 2017; Cohen et al., 2019). In particular, in the AS4.0 the new technologies of Industry 4.0 can be used both to enable better performance (Battini et al., 2020; Fager et al., 2020; Peron et al., 2020; Simonetto et al., 2020; Arena et al., IN PRESS) and to design their workplaces (Dolgui et al., 2021). Indeed, regarding this last point, the combined use of the motion capture (mocap) system and Virtual Reality (VR) has been reported to improve the AS workplace design compared to the currently used AS workplace design procedures (e.g., physical mock-ups and computer-aided systems) with respect to both productivity and operators' wellbeing (Faccio et al., 2017; Battini et al., 2018). Specifically, a mocap system allows the creation of a digital copy of the operator in real time thanks to "a virtual representation of the skeleton and its movements" (Bortolini et al., 2020), and this facilitates and improves ergonomic assessments, since AS workplace designers have access to the exact evolution over time of the position and orientation of the operator's different limbs in a common reference system (Oyekan et al., 2017). VR, instead, gives the opportunity to create a three-dimensional environment where users can interact efficiently with three-dimensional objects in real time using their natural senses and skills (Riva, 2002). The adoption of a mocap system and VR during the design phase of an AS makes it possible to overcome the main limitations of the currently used AS workplace design procedures, i.e., subjective and time-consuming assembly time measurements and ergonomic assessments (Battini et al., 2011; Battini et al., 2014). The combined use of these two technologies, in fact, enables fast and reliable assembly time measurements and ergonomic assessments (see Section 2.1 for more details).

It is hence clear that the potentialities associated with the use of these technologies are enormous, but it is still unclear how to maximise the associated benefits. In fact, to the best of the authors' knowledge, a clear description of the methodology that needs to be followed when designing using the mocap system and VR is still missing, and this represents one of the main reasons why "their adoption within real industrial environments is still very limited" (Prabhu et al., 2016). In this work, to fill this gap, we developed a methodological framework that AS workplace designers can follow when using mocap and VR. Specifically, the methodological framework herein developed gives a central role to the operator. In more detail, the operator is included in the methodological framework by considering his/her main characteristics, i.e., (i) physical strength and joint mobility and (ii) experience. This is extremely important in consideration of the current ageing of the workforce. In fact, a reduction in physical strength and joint mobility and an increase in experience is known to be associated with the ageing workforce (Börsch-Supan and Weiss, 2016; Roda et al., 2019; Peng and Chan, 2020). Characteristic (i) is considered in the ergonomic assessments, where, referring to the work of Wolf and Ramsauer (2018), we modified the ergonomics index Rapid Entire Body Assessment (REBA). Dealing with (ii), then, the experience of the ageing workforce is considered in the design phase: thanks to the experience they gained over the years, they can provide advice that can simplify and speed up the design procedure (Di Pasquale et al., 2020). The methodological framework has then been successfully applied to a simple but representative case study to prove its validity and to demonstrate its use.

The rest of the paper is organised as follows. In Section 2 we present a

narrative literature review dealing with the design of AS workplaces (Section 2.1) and thereafter the impact of the ageing workforce on productivity and OSH performance in AS workplaces (Section 2.2). Specifically, in this work we focus only on operators' wellbeing as OSH performance. Section 3 then presents the methodological framework developed herein. In Section 4 the case study is described and discussed. Finally, Section 5 concludes the work.

2. Literature review

In this section, a narrative literature review is conducted in order to have an overview of the evolution of the AS workplace design procedures (Section 2.1) and of the impact of the ageing workforce on productivity and OSH performance in these workplaces (Section 2.2). Specifically, as in the rest of the work, we focus only on operators' wellbeing as OSH performance.

2.1. Design of the assembly system workplace

AS workplace design, defined by Launis et al. (1996) as the "process and activity which leads to the birth of the workplace", has been subject to substantial changes over the last years. From what we will refer to as the "traditional AS workplace design procedure", i.e., the construction of the physical workplace mock-up proposed by Das and Sengupta (1996), in fact, new design procedures have been developed, especially thanks to the technological developments over the years, aiming to overcome the main limitations of the traditional AS workplace design procedure (i.e., time- and resource-consuming).

Based on the pioneering work of Warnecke and Haller (1982) and Bauer and Lorenz, (1985), Zha and Lim (2003) suggested overcoming these limitations using computer-aided systems. Specifically, they developed an algorithm that optimised the AS workplace design in a virtual environment, hence eliminating the need to create several physical mock-ups. Similarly, Udosen (2007) studied the possibility of optimising the AS workplace design thanks to a computerised heuristic. Through a case study, they demonstrated the possibility of optimising the AS workplace with respect to the cycle time in a faster and easier way compared to the traditional approach. However, the use of computer-aided systems became unsatisfactory when the urge to include also ergonomic considerations in the AS workplace design emerged.

The continuously increasing flexibility required by the market over the last years has in fact rendered the use of human operators in AS fundamental. The operators' capability to easily adapt to changes in production has in fact been considered essential in order to cope with the market's characteristics. Therefore, the concept of the humancentred AS workplace has started to arise (Giacomin, 2014; Caputo et al., 2019), pointing out the necessity to include ergonomic considerations in AS workplace design (Neubert et al., 2012; Sišková and Dlabač, 2013). Worthy of mention in this perspective is the work of Battini et al. (2011), who developed a methodological framework to optimise the AS with respect to both productivity and operators' wellbeing.

However, despite some improvements, the computer-aided systems have not been fully capable of adapting to these needs. In fact, although much has been done from the initial procedure where a physical mockup was needed to evaluate the operators' wellbeing in the computeraided designed AS workplace through photos and/or videos of the assembly operations, some limitations still exist (Naddeo et al., 2014). Specifically, although in their latest versions computer-aided systems make it possible to study both productivity and operators' wellbeing directly in the virtual environment (Feldmann and Junker, 2003), hence overcoming the limitations of using physical mock-ups and photos and/ or videos of the assembly operations for the ergonomic assessments (Realyvásquez-Vargas et al., 2020), the results are not fully representative of the reality. In fact, the ergonomic assessments are carried out on a virtual operator which does not represent the reality, since its movements and anthropometrical data differ from those of the operator for whom the AS workplace is being designed.

Recently, the combined use of the mocap system and VR has emerged as a new AS workplace design procedure that can overcome these limitations (Jayaram et al., 2006; Di Pardo et al., 2008). Specifically, this AS workplace design procedure consists in creating first the virtual environment in which the operator is immersed through the use of VR. The operator is then equipped with a mocap system that allows the creation of the operator's digital twin, i.e., a digital copy of the operator that is a truthful representation of the reality. In this way, the operator can virtually perform all the activities that he would normally do during his job, but without the need for a physical mock-up. Moreover, the ergonomic assessments are carried out on the operator's digital twin, which has the same anthropometric data and makes the same movements. Furthermore, it is also possible to carry out the ergonomic assessments in real time (Vignais et al., 2013; Battini et al., 2014).

Some researchers have thus been attracted by the potentialities offered by the combined use of the mocap system and VR for AS workplace design, and they have started investigating these



Fig. 1. Methodological framework. The dashed arrows indicate the steps subjected to possible changes during the iterative process.

potentialities in more detail (Peruzzini et al., 2017; Vosniakos et al., 2017; Michalos et al., 2018; Caputo et al., 2017; Battini et al., 2018). Peruzzini et al. (2017), for example, evaluated the AS workplace design procedure in a steel pipe manufacturer using either the mocap system and VR or a computer-aided system, and they reported that the outcomes of the AS workplace design procedure were substantially improved using the former thanks to the more accurate ergonomic assessments that could be obtained. Similar results were obtained also by Vosniakos et al. (2017) for the assembly of small to medium aircraft wings, and by Caputo et al. (2017) for assembly tasks in a lab environment. Michalos et al. (2018) then further proved the efficacy of this design procedure, focusing on the assembly of a robotic wrist. Specifically, they demonstrated that through the combined use of the mocap system and VR it was possible to easily redesign the robotic wrist AS workplace and to improve its performance in terms of the distance walked and task execution times. Finally, Battini et al. (2018) further investigated the potentialities of this design procedure, suggesting the possibility of coupling the mocap system and VR with a heart rate monitoring system in order to include also fatigue considerations in the AS workplace design.

However, despite the increasing interest in the use of the mocap system and VR for AS workplace design, to the best of the authors' knowledge a clear and detailed procedure to follow when designing the use of these technologies is still missing. In this work, we thus aim to fill this gap by developing a methodological framework that the AS workplace designer can follow, allowing full exploitation of the potentialities associated with the combined use of the mocap system and VR. Moreover, with the methodological framework we aim to overcome another limitation of the current design procedures, which, in fact, give only limited consideration to the operators. The different physical strength and joint mobility that can characterise the operators (due to their different ages, for example) are not considered in the ergonomic assessments, and their different levels of experience are not considered in any way in the current design procedures either. In the methodological framework herein developed we overcome these limitations. Specifically, the operators' physical strength and joint mobility are included in the ergonomic assessments, while their experience is used for useful advice concerning the AS workplace layout (e.g., the position of components and/or tools).

2.2. Ageing in the assembly system workplace

It is widely known that the world population is ageing, with the most developed countries having the highest share of older people (United Nations, 2018). This has great repercussions on the labour market, which is naturally experiencing an ageing workforce due to the necessity of increasing the retirement age (Hanson and Lindgren, 2020). This represents a key issue in some industrial sectors, especially in manufacturing. In fact, in a sector such as manufacturing where repetitive movements are often required, the decline of the operators' physical functions as the years go by (e.g., reduced musculoskeletal force, flexibility and motion capability) certainly represents a threat (Shephard, 2000; Ilmarinen, 2001; Bouma, 2013; Cote et al., 2014; Bures and Simon, 2015; Truxillo et al., 2015; Peng and Chan 2019; Orrù et al., 2020; Peng and Chan, 2020). For example, Roper and Yeh (2007) reported that, as a consequence of the decline of the operators' physical functions, musculoskeletal disorders (MSDs) are one of the most commonly found health risks among ageing workers. Similar observations can be found also in Jones et al. (2013). Specifically, Landau et al. (2008), analysing the assembly operations in the automotive industry, reported that lumbar spine and head-neck-shoulder symptoms are the most frequent MSDs that ageing workers suffer from, in agreement with Qin et al. (2014). It is hence clear that, to reduce the occurrence of MSDs, the ergonomic risks need to be reduced. To do so, Balderrama et al. (2015), for example, suggested assigning ageing workers to appropriate job positions, where the work demands match their physical

abilities. Similarly, Kovalski-Trakofler et al. (2005) suggested that ageing workers should avoid activities involving extreme joint movements, excessive force and highly repetitive movements since they involve high ergonomic risks.

Moreover, Varianou-Mikellidou et al. (2020), confirming the results of Roper and Yeh (2007) and Pennathur et al. (2003), reported that the decline of the operators' physical functions with increasing age affects also their efficiency. Compared to their younger colleagues, in fact, ageing workers are reported to be slower in their reactions and movements, as well as to require longer resting times after physical efforts (Zieschang and Freiberg, 2011). This, in turn, decreases their productivity. Göbel and Zwick (2009), in fact, reported that the productivity decreases after the age of 40. Similar results were also obtained by Alessandri et al. (2020) and by Abubakar and Wang (2019). Specifically, the latter reported that the decrease in productivity starts already at the age of 38, with an increase of 1% in the average task execution time every year after that age. However, Börsch-Supan and Weiss (2016) reported the opposite findings. In fact, evaluating truck assembly, they reported that the productivity profile of individual workers was increasing until the age of 65, and they explained their results by observing that older workers were able to compensate for the decline of their physical functions thanks to the experienced gained over the years. In fact, although ageing workers were found to make slightly more errors than their younger colleagues, they made hardly any severe errors due to their increased experience. Similar results were found also by Roosaar et al. (2019) and Kim (2019), who showed that ageing workers remained at least as productive as their younger colleagues thanks to their experience.

From what is reported above, there clearly emerges a need to consider explicitly the age of the operators in ergonomic and productivity-related considerations, especially when designing the AS workplace. For example, acting on the components location, extreme joint movements could be avoided, as well as limiting the negative impact of the ageing workers' slower movements. In this work, we developed a methodological framework for the workplace design that explicitly considers the age of the operator, including both the negative and positive aspects (reduced physical functions and increased experience, respectively), as better described in the previous section.

3. Methodological framework

In this section the methodological framework developed to guide an AS workplace design that includes both ergonomic and productivityrelated considerations is reported (Fig. 1) and explained in detail. Specifically, the AS workplace design procedure considered deals with the combined use of the mocap system and VR. Before moving forward, it is worth mentioning that the methodological framework is meant to be used during the detailed design phase, and it has been developed in such a way that it can be combined with already existing frameworks for AS design. For example, the framework developed herein can be used for carrying out steps 5 to 8 of Battini's framework (Battini et al., 2011).

3.1. Step 1 - Input collection

The first step is the collection of the input parameters. Since, as mentioned earlier, the methodological framework developed herein is meant to be used in the detailed design phase, we consider the known outputs of the preliminary design phase, and therefore the (i) AS layout configuration, (ii) cycle time (paced/un-paced), (iii) workstation type (i. e., open/closed, parallel/serial, two-sided, ...), and (iv) automation level (i.e., percentage of automation, type of equipment, ...) are known. Moreover, the product characteristics that influence ergonomics and safety (e.g., component weights) are also known in this design phase. Furthermore, the operators who will work on the AS workplace under development are generally known, and thus also their anthropometric data and age.

3.2. Step 2 - Workplace design

Once the inputs have been collected, the AS workplace design can be carried out. In this step, contrary to the traditional AS workplace design approach where physical mock-ups are needed, the workplace is designed virtually, using a 3D modelled environment. Many different alternatives can hence be easily developed and assessed, without the need to build a physical mock-up for each alternative. This, as already stated before, brings great benefits in terms of time and cost reduction.

During this phase, it is fundamental that the AS workplace designers are assisted by experienced operators. In fact, as we saw from the literature review, operators are affected by a gradual decrease of both physical and cognitive abilities with increasing age (Shepard, 2000; Bouma, 2013; Bures and Simon, 2015), and experienced operators, thanks to the experience gained over the years, can provide useful advice to the AS workplace designers to take these aspects into consideration. For example, to ensure their wellbeing, older operators might be required to make fewer movements, move fewer heavy objects, and do fewer complex activities during their daily assembly activities. Experienced operators, then, based on their knowledge, can advise the AS workplace designers on the proper position of, e.g., components and equipment to avoid poor work postures and unnecessary movements. Moreover, experienced operators can suggest to the AS workplace designers whether (i) the operators need to be supported by new equipment (e.g., lifter, automatic screwdrivers, collaborative robots, etc.) and (ii) the environmental conditions (e.g., lighting) need to be adapted or not, based also on their age. Moreover, in this phase, it is crucial that the AS workplace designers and the experienced operators adopt a holistic approach, where they consider productivity and operators' wellbeing as two complementary aspects, not as two separate entities. The AS workplace layout influences, in fact, both the productivity and operators' wellbeing through the whole assembly process: for example, the positioning of a component (or tool) in a certain location with the aim of making the assembly tasks faster might have a negative impact on the operators' wellbeing, and vice versa. Moreover, it is equally important that the AS workplace designers and the experienced operators do not consider each task as single, separate entities, but as a part of a wider aspect. Sometimes, in practice, some trade-offs are necessary to reach the overall objectives: the productivity and/or safety of an assembly task might need to be penalised to reach the desired goals in terms of productivity and operators' wellbeing for the whole assembly process.

It is worth noting that in this step, although many software tools are available to model a 3D environment, only those software tools that can interact with the mocap system and with VR can be used (e.g., Siemens JackTM). The interaction with the mocap system is, in fact, a crucial aspect since it makes it possible to obtain the real movements of the operator and to accurately replicate them in the 3D modelled environment.

3.3. Step 3 - Data collection

Once the virtual workplace is built, the collection of the data can take place. In this step, an operator needs to be equipped with the mocap system and with VR. Thanks to the combined use of the mocap system and VR, he will be immersed in the 3D modelled environment, being able to move and to interact with it. In this way he can carry out the assembly process virtually, without the need for any physical mock-up. While the virtual assembly process is being carried out, the mocap system allows the recording of the assembly operations (from now on we will refer to this as "mocap-based recording"), as well as the physical data (e.g., body joint angles, body segment orientation and positions) that will be used in Step 4 to evaluate the ergonomics. For the sake of clarity, from now on we will refer to these physical data as "ergonomicsrelevant data", and they are available for each frame of the mocap-based recording (the frame rate can usually be decided beforehand). It will be clear in the description of Steps 4 and 5 that this represents a very

important aspect.

Before moving to the description of Step 4, it is worth mentioning that at the beginning of Step 3 a crucial activity is needed to be able to collect useful and reliable data. A digital twin of the operator carrying out the virtual assembly process needs to be created before carrying out the virtual assembly process. In this way, we ensure that the anthropometric data of the operator's digital twin corresponds to those of the operator who will perform the virtual assembly process. This is fundamental for obtaining accurate and reliable ergonomic assessments. In this step, we recommend that the operator chosen to carry out the virtual assembly process is representative in terms of the anthropometric data of the operators who will work on the AS workplace under development. Considerations about the age of the operator chosen to carry out the virtual assembly are reported in Step 4.

3.4. Step 4 - Data analysis

Once the data are collected, they need to be analysed with respect to productivity and ergonomics. The productivity can be evaluated by means of time-related parameters (e.g., task execution times), which in the following we will refer to as "productivity KPIs", while the ergonomics can be evaluated using ergonomic indexes (RULA, REBA, NIOSH, OWAS, etc.). During the data analysis step, in order to optimise the outcomes, the AS workplace designers are required to divide the whole assembly process into the different assembly tasks that constitute it. Thus, each assembly task has to be associated with its own data (mocap-based recording and ergonomics-relevant data). In this way, the AS workplace designers can evaluate the productivity KPIs and the ergonomics index scores for each assembly task. The former can be evaluated through the mocap-based recording of the assembly: the task execution time, for example, can be evaluated based on the length of the mocap-based recording of the task under consideration. For the ergonomic scores, instead, the AS workplace designers have to consider the ergonomics-relevant data of each frame that constitutes the assembly task under consideration. Through simple formulas, these data can be converted into the ergonomic scores in a fast way (the process can be automated by using EXCEL, MATLAB or any other similar tool). For example, considering the REBA index, the relative angle between the neck and the trunk corresponds to the "neck score" in the REBA index. By doing so, it is hence possible to evaluate for each task the average and the distribution of the ergonomic scores (i.e., considering again the REBA index, how frequently is the REBA score equal to 1, to 2 and so on). Moreover, by analysing jointly the mocap-based recording of each task and the REBA scores of each frame constituting that specific task, it is possible to determine the subtasks that involve high ergonomic risks and to link them to the AS workplace layout, hence facilitating the AS workplace design improvement procedures. While doing this, the AS workplace designers can also note, for each task, what we will refer to as a value-added ratio, i.e., the ratio between the time spent on necessary activities and the total task execution time. In this way, it is easier to identify which tasks are characterised by highly unproductive activities (i.e., activities that do not add value for the assembly operations, e.g., moving, travelling, etc.), hence having a negative impact on the productivity KPIs, and to link them to the AS workplace layout.

During the ergonomic assessments, it is crucial that the age of the operator is explicitly considered. However, this is still overlooked in the literature, and, to the best of the authors' knowledge, only Wolf and Ramsauer (2018) proposed a solution for this. In detail, they proposed to modify the ergonomic scores by means of age-based multipliers, and they suggested evaluating these age-based multipliers by means of mathematical formulas. Specifically, they suggested obtaining these mathematical formulas by interpolating literature data about the decline in the ability under consideration (muscular strength, aerobic capacity, etc.) with respect to age. For example, they were interested in an age-based multiplier for strength reduction in order to include age considerations in the KIM ergonomics index. To evaluate this age-based

Safety Science 146 (2022) 105561

Table 1

Execution time, average REBA	score and	value-added	ratio fo	r each	task i	n the
as-is configuration.						

Task	Task execution time (s)	Average REBA score		Value-added ratio	
		Age = 27	Age = 60	(%)	
1	19.0	3.40	3.82	62.2	
2	7.2	2.30	3.51	60.3	
3	14.5	3.51	3.88	62.0	
4	72.6	4.63	5.23	54.5	
5	87.5	4.47	6.65	37.8	
6	62.0	4.50	5.40	67.6	
7	6.3	3.42	4.42	77.1	
8	18.0	4.36	4.96	41.5	
9	19.4	3.91	5.33	29.3	
10	75.8	4.29	4.84	60.4	
11	62.5	5.03	5.96	69.3	
12	22.2	2.60	3.12	40.6	
13	16.2	1.76	2.14	73.2	
14	56.1	4.51	5.25	48.0	
15	59.1	3.49	4.36	43.4	
16	132.1	3.94	4.78	43.0	
17	28.1	3.90	4.32	41.7	
Total	758.6	4.12	4.97	51.2	

multiplier ($f_{strength}$), they developed the mathematical formula reported in Equation (1) by fitting the data available from the literature regarding the changes in the muscular strength with respect to age.

$$f_{strength}(\%) = 0.00058 \cdot age^3 - 0.08478 \cdot age^2 + 3.24439 \cdot age + 62.92006$$
(1)

It is worth noting that the values of $f_{strength}$ provided by Equation (1) are in line with the age-based multipliers of the REFA method and with those provided by ISO 11228 and EN 1005.

Building on this approach, we suggest in Equation (2) the age-based multiplier for joints flexibility reduction ($f_{flexibility}$), which is needed when the ergonomic indexes contain assessments on joint flexibility (e. g., allowable joint angle limits in the REBA index). To obtain the mathematical formula reported in Equation (2), we fitted the joints flexibility reduction data reported in Table 1 of Wolf and Ramsauer's work (2018).

$$f_{flexibility}(\%) = -0.00019 \cdot age^3 + 0.034286 \cdot age^2 - 2.538095 \cdot age + 145$$
 (2)

By considering these age-based multipliers in the ergonomic assessments, we can ensure that the age of the operator is explicitly considered (see Appendix A for more details on how to include the age-based multipliers in the REBA index). Specifically, we strongly recommend following a precautionary approach, where the age of the oldest operator who will work on the AS workplace under development is used in these formulas.

It is not strictly necessary that the operator used for the virtual assembly is the oldest operator who will work on the AS workplace under development, and it can also be a younger colleague. The choice depends on the following considerations and a trade-off decision might be needed. On the one hand, if the oldest operator is used as the operator for the virtual assembly, any potential decrease in productivity associated with the increase of age is included. However, if the technological skills of the oldest operator are limited, it might take too much time to train him sufficiently to carry out a virtual assembly process that is representative of the reality. On the other hand, if a younger colleague is used for the virtual assembly, the chances of low technological skills are limited, but any potential decrease in productivity associated with the increase of age will get lost. The choice depends therefore on the workforce characteristics (in terms of technological skills) and on the assembly process under consideration (i.e., whether the assembly process is affected timewise by the age of the operator or not). However, whatever the choice is, the ergonomic assessments have to be modified with the age-based multipliers calculated with the age of the oldest



Fig. 2. Pump assembled in the case study.

operator.

3.5. Step 5 - Ergo-productivity satisfaction

The productivity KPIs and the ergonomic scores obtained from the previous step serve as input in this step, the ergo-productivity satisfaction. This is a decision step, where the AS workplace designers have to decide whether the productivity KPIs and ergonomic scores are satisfactory or not (with respect to the company's requirements, legal requirements, etc.). If they are, this represents the final AS workplace design, if not, then the user has to go back to step 2.

In the latter case, the workplace designers and the experienced operators can draw on the results of step 4 to improve the AS workplace design by modifying its layout. Thanks to those results, in fact, the workplace designers and the experienced operators know which are the tasks that decrease productivity and operators' wellbeing (i.e., the tasks characterised by low value-added ratios and high ergonomic scores, respectively). Moreover, they also know which subtasks are responsible for these criticalities and how they are linked to the AS workplace layout. Again, it is important that, when redesigning the AS workplace, the AS workplace designers and the experienced operators adopt a holistic approach, i.e., they (i) do not focus only on the critical tasks and (ii) do not consider productivity and operators' wellbeing as two separate entities, but as two complementary aspects. This is fundamental, since focusing the AS workplace improvements only on one single task and on one single aspect (i.e., productivity or ergonomics) per time would not lead to the best results achievable, since the modification of one task has impacts also on other tasks, sometimes worsening them. Once an improved design has been identified, the workplace designers can move again through steps 3-5. It should be noted that the operator's digital twin does not need to be created (or modified) every time the iteration takes place, unless the operator carrying out the virtual assembly operation changes.

4. Case study and discussion

To demonstrate how to use the methodological framework developed herein and to prove its validity, we used a simple but representative case study carried out at the Logistic 4.0 Lab at NTNU. Specifically, in the case study we applied the methodological framework for the redesign of the AS workplace of a medium/big size pump (Fig. 2), which is constituted of 25 components assembled through 17 main tasks (information about the components and the different tasks can be found in Appendix B).

The case study was carried out as follows. First, a second-year Ph.D. student working on AS4.0 designed the initial workplace, which from now on we will refer to as the "as-is configuration". In this phase, the AS

Safety Science 146 (2022) 105561

workplace was built physically in approx. 10 h, and the Ph.D. student decided the AS workplace layout, e.g., the location of components and tools (wrenches and manual and automatic screwdriver), sub-assemblies' feedings, etc., by himself, i.e., without any support from more experienced personnel. It is worth noting that since the focus of the case study is only on the methodological framework, we considered the pump to be assembled in a single workplace. Although this does not faithfully reproduce the real industrial scenario where the assembly process is carried out among several AS workplaces (i.e., in an assembly line), we decided to consider a single workplace to facilitate the understanding of how to apply the methodological framework.

Then, we gathered all the information necessary to carry out the redesign of the AS workplace according to the methodological framework, i.e., (i) which are the tasks that negatively affect the productivity and the operators' wellbeing, and (ii) which are the subtasks that are responsible for these criticalities and their link to the AS workplace layout. The redesign of a current workplace can, in fact, be considered equivalent to the case where the first virtual design does not satisfy the ergo-productivity requirements and a second iteration is needed: as during the second iteration the methodological framework requires that information, in the same way they are needed in the case of a redesign.

To obtain that information it was hence necessary to carry out the assembly process and to collect and analyse the data according to Step 3 and Step 4 of the methodological framework, respectively. The assembly process was carried out by the Ph.D. student mentioned above (to whom we will refer as the "case-study operator" in the following), who was trained for two hours the day before the Data Collection Step in order to increase his assembly skills. During the Data Collection Step, the casestudy operator was equipped with a mocap system to obtain the mocap-based recording and the ergonomics-relevant data. The mocap system used was the inertial motion capture system developed by Synertial, which consisted of 29 inertial measurement units (IMUs) that, thanks to an advanced compensation system, guaranteed to obtain very accurate ergonomics-relevant data. Specifically, the IMUs being placed on a full body suit, these data covered the full body, including also the hands (15 IMUs on the body and 7 on each hand). The mocap system was connected to a personal computer via a WIFI connection (this was possible because all the sensors communicate with a small portable multi-processing unit that in turn communicated with the personal computer), and the required data were obtained by using the Synertial SynDash software. It should be noted that to obtain reliable ergonomicsrelevant data the case-study operator's digital twin was needed (see the previous section for more details), and to do so we used Synertial AutoCal.

The outcome of the Data Collection Step consisted of a 758.6-seconds mocap-based recording of the assembly process constituted by 45,519 recording frames (and hence ergonomics-relevant data). These data were then analysed according to the Data Analysis Step of the methodological framework. Following the procedure of the Data Analysis Step, we divided the mocap-based recording of the whole assembly process into the 17 different assembly tasks. In this way, we were able to evaluate the productivity KPIs and the ergonomics index scores for each assembly task. Specifically, we considered the tasks execution times and the REBA index as the productivity KPI and ergonomics index, respectively. The execution time and the average REBA score of each task are reported in Table 1. It is worth noting that the age of the case-study operator (27 years) was explicitly considered in the ergonomics assessment by means of the age-based multipliers $f_{strength}$ and $f_{flexibility}$, which assumed a value of 1 and 0.98, respectively. Moreover, for the purpose of showing the importance of considering the age of the operators in the ergonomic assessments, in Table 1 we report also the (fictitious) case where the age of the case-study operator was 60 years ($f_{strength}$ and $f_{flexibility}$ equal to 0.77 and 0.75, respectively). As can be seen, this has a marked impact on the ergonomic assessments, and this can be clarified even better if the distribution of the ergonomic scores is considered (see



Fig. 3. Detail of the virtual assembly: operator equipped with the mocap system and VR (a), (b) virtual operator (c) view of the assembly operation in VR.



Fig. C1. REBA scores (average and distribution) for each task considering the real age (blue histograms) or a higher fictitious age (orange histograms) of the casestudy operator in the ergonomic assessments of the as-is configuration.

Safety Science 146 (2022) 105561



Fig. C.1 in Appendix C). Therefore, as suggested in the methodological framework, in the following, we will consider the age of the case-study operator equal to 60 years when dealing with any ergonomic consideration. Moreover, as reported in the *Data Analysis Step*, we analysed jointly the mocap-based recording of each task and the REBA scores of each frame constituting that specific task to identify the subtasks that

present a high ergonomic risk (i.e., REBA score equal to or higher than 8) and how these are linked to the AS workplace layout. Furthermore, while doing so, we evaluated the value-added ratio for each task (which are also reported in Table 1) and we determined the unproductive activities responsible for low value-added ratios, as well as their link to the AS workplace layout.

It was hence possible to identify the tasks that reduced the productivity (i.e., tasks with low value-added ratios) and the operators' wellbeing (i.e., tasks with high ergonomic risks), which are:

- For productivity: tasks 5, 8, 9, 12, 14, 15, 16 and 17
- For operators' wellbeing: tasks 4, 5, 6, 7, 9, 11, 14 and 16

Moreover, we identified also the subtasks responsible for these criticalities and we linked them to the AS workplace layout, finding that they were mainly due to the location of the components.

After collecting this information, we were then able to proceed with the redesign of the AS workplace according to the methodological framework developed herein. In the following, we will describe in detail the procedures carried out in the different steps of the methodological framework. Specifically, since we applied the methodological framework for a redesign (that we said before to be equivalent to a second iteration) we will discuss only Steps 2–5.

4.1. Step 2 – Workplace design

In this Step, the AS workplace was redesigned virtually using Siemens JackTM. Specifically, we adopted Siemens JackTM to model the 3D environment because it can interact with the mocap system and with the VR available in the Logistic 4.0 Lab (Synertial mocap and HTC VIVETM, respectively). The redesign procedures were carried out by a postdoctoral fellow in digital production and logistics systems, together with a full professor in industrial logistics. The former assumed the role of the AS workplace designer, modelling the 3D environment, while the latter assumed the role of the experienced operator, providing advice to the AS workplace designer concerning the AS workplace layout. Specifically, to provide meaningful advice, the information obtained from the analysis of the as-is configuration, in which there were tasks decreasing the productivity and the operators' wellbeing, as well as subtasks that were responsible for these criticalities and their link to the AS workplace layout, was crucial. Having ascertained that the inadequate location of certain components was the main cause of the low value-added ratios and high ergonomic risks of the activities reported above, the full professor was able to identify some improvement actions. Specifically, when providing his advice, the full professor adopted the holistic approach mentioned in the methodological framework, considering the repercussions that repositioning components and tools (and modifying the AS workplace layout in general) to increase the value-added ratio of a certain task could have on the ergonomic risk level of that task (and vice versa) and on the productivity and operators' wellbeing of the whole assembly process. Some examples of advice were that

- adjusting the height of the workstation could reduce the ergonomic risks in tasks 4 and 14, with no impact on the productivity (or on that task or the whole assembly process);

- the relocation of the components necessary to reduce the ergonomic risks in tasks 6, 7 and 11 could slightly affect the productivity (slightly higher task execution time); however, relocating these components allows an increase in the overall productivity of the assembly process since the repositioning of other components is then possible.

Based on these and other suggestions from the full professor, the workplace was then virtually designed in approx. 5 h.

4.2. Step 3 – Data collection

This step was already described in detail when dealing with the as-is configuration, but its main parts will now be summarised.

The assembly process was carried out virtually by the case-study operator equipped with the mocap system and with the VR available in the Logistic 4.0 Lab (i.e., Synertial mocap and HTC VIVETM, Safety Science 146 (2022) 105561

Table 2

Execution time, average REBA score and value-added ratio for each task in the redesigned configuration.

Task	Task execution time	Average REBA score		Value-added ratio	
	(s)	Age = 27	Age = 60	(%)	
1	20.9	3.48	3.92	56.7	
2	8.3	3.62	3.84	52.4	
3	12.9	4.37	4.67	70.0	
4	70.7	4.22	4.78	56.0	
5	52.3	4.27	5.12	63.3	
6	69.5	3.88	3.97	60.3	
7	6.9	3.35	3.84	70.6	
8	13.5	2.79	2.97	55.3	
9	9.3	3.10	4.17	61.0	
10	78.1	3.19	3.87	58.6	
11	64.3	4.06	4.13	67.4	
12	13.1	3.35	3.75	68.9	
13	17.3	3.54	3.91	68.6	
14	38.2	3.58	3.87	70.5	
15	49.3	3.48	4.17	52.1	
16	98.0	3.57	4.10	57.9	
17	19.0	3.74	3.80	61.5	
Total	641.6	3.74	4.24	60.6	

respectively) (Fig. 3). It is worth recalling that the case-study operator was trained for 2 h the day before collecting the data in order to familiarise himself or herself with the VR and to learn how to interact with the 3D modelled environment.

The assembly process was recorded through the mocap system, which allows recording of the assembly operations and the ergonomics-relevant data. Specifically, the mocap-based recording was 641.6 s long and consisted of 38,493 recording frames (and hence ergonomics-relevant data).

It is worth noting that to collect useful and reliable data it is necessary to have the digital twin of the case-study operator in order to ensure that the anthropometric data of the physical and virtual operator correspond. However, this time it was not necessary to develop it again since it was already available from the assessment of the as-is configuration.

4.3. Step 4 – Data analysis

The outcome of Step 3 (a 641.6-seconds mocap-based recording of the assembly process constituted by 38,493 recording frames and as many sets of ergonomics-relevant data) was then divided into the 17 different assembly tasks and analysed in terms of task execution times and the REBA index (the chosen productivity KPI and the ergonomics index, respectively) (Table 2), lasting approx. 2 h. Again, to further emphasise the importance of considering the age of the operators in the ergonomic assessments through the age-based multipliers, we report in Table 2 the average REBA score for each activity considering two ages of the case-study operator, i.e., the real one (27) and the fictitious one (60). More details of the ergonomic assessments can be found in Fig. C.2 in Appendix C.

4.4. Step 5 - Ergo-productivity satisfaction

In this step, according to the methodological framework, the productivity KPI and the ergonomic scores have to be compared with the company's requirements (or any other requirement) to evaluate whether the AS workplace design is satisfactory or not. However, since our case study was not carried out in collaboration with any company but in the Logistic 4.0 Lab in order to illustrate the use of the methodological framework and to prove its validity, we did not have any benchmark values. Therefore, we simply evaluated whether the

Safety Science 146 (2022) 105561



Fig. C2. REBA scores (average and distribution) for each task considering the real age (blue histograms) or a higher fictitious age (orange histograms) of the casestudy operator in the ergonomic assessments of the redesigned configuration.

Safety Science 146 (2022) 105561



redesigned AS workplace performed better than the as-is configuration with respect to both productivity and operators' wellbeing (a comparison of the as-is and redesigned configuration is reported in Table 3). From the analysis of the task execution times it emerges that, although some tasks are characterised by higher execution times (tasks 1, 2, 6, 7, 10, 11 and 13), the total assembly time in the redesigned configuration is lower than in the as-is configuration (758.6 vs 641.6 s). The methodological framework developed herein, in fact, allowed us to identify the unproductive activities (e.g., travelling to pick the right tool, moving to pick the right component, etc.) and their causes (i.e., many components and tools were stored in locations which were not optimal). This information was then used by the experienced operator (i.e., the full professor in industrial logistics in our case study) to provide advice on how to eliminate and/or reduce some of these unproductive activities (hence the higher value-added ratios in the redesigned configuration). However, while giving this advice, the experienced operator had to make some trade-off decisions, penalising the execution times of some tasks to favour the total assembly time. This is part of the holistic approach mentioned in the methodological framework, where each task is not considered as a single entity, but as a part of a wider context.

Moreover, since the holistic approach jointly considers productivity and ergonomics, the overall operators' wellbeing was also increased. In fact, although some tasks are characterised by a higher average REBA score in the redesigned configuration (tasks 1, 2, 3, 12 and 13), the overall average REBA score decreased. Furthermore, no tasks with high ergonomic risks (i.e., REBA score equal to 8 or higher) were present in the redesigned configuration (Figs. C.1 and C.2 in Appendix C).

It is worth reiterating the importance of considering the age of the operator in the ergonomic considerations. As can be seen from Tables 1-3, considering different ages of the case-study operator through the age-based multipliers had a marked impact on the REBA scores: for example, the average REBA score of task 5 in the as-is configuration moved from 4.47 to 6.65 by considering the age of the case-study operator equal to 60 instead of 27 years. The effects of considering the case-study

Table 3

Tasks execution times, REBA average scores (both considering the age of the case-study operator equal to 27 years and 60 years) and value-added ratios in the as-is and redesigned configuration.

Task	Task execu	Task execution time (s)		Average REBA score				Value-added ratio (%)	
		Age = 27		Age = 60					
	As-is	Redesigned	As-is	Redesigned	As-is	Redesigned	As-is	Redesigned	
1	19.0	20.9	3.40	3.48	3.82	3.92	62.2	56.7	
2	7.2	8.3	2.30	3.62	3.51	3.84	60.3	52.4	
3	14.5	12.9	3.51	4.37	3.88	4.67	62.0	70.0	
4	72.6	70.7	4.63	4.22	5.23	4.78	54.5	56.0	
5	87.5	52.3	4.47	4.27	6.65	5.12	37.8	63.3	
6	62.0	69.5	4.50	3.88	5.40	3.97	67.6	60.3	
7	6.3	6.9	3.42	3.35	4.42	3.84	77.1	70.6	
8	18.0	13.5	4.36	2.79	4.96	2.97	41.5	55.3	
9	19.4	9.3	3.91	3.10	5.33	4.17	29.3	61.0	
10	75.8	78.1	4.29	3.19	4.84	3.87	60.4	58.6	
11	62.5	64.3	5.03	4.06	5.96	4.13	69.3	67.4	
12	22.2	13.1	2.60	3.35	3.12	3.75	40.6	68.9	
13	16.2	17.3	1.76	3.54	2.14	3.91	73.2	68.6	
14	56.1	38.2	4.51	3.58	5.25	3.87	48.0	70.5	
15	59.1	49.3	3.49	3.48	4.36	4.17	43.4	52.1	
16	132.1	98.0	3.94	3.57	4.78	4.10	43.0	57.9	
17	28.1	19.0	3.90	3.74	4.32	3.80	41.7	61.5	
Total	758.6	641.6	4.12	3.74	4.97	4.24	51.2	60.6	

operator's age are even more evident if Fig. C1–C2 in Appendix C are considered: many tasks that did not involve high ergonomic risk when the case-study operator was considered young (age = 27 years) became high-risk when he was considered old (age = 60 years). Therefore, as reported in the methodological framework, the ergonomics assessment should always include the age of the operator. But the highest age should be considered when designing an AS workplace not only because it allows consideration of the fact that some activities might not be of high ergonomic risk for young operators, but because they might become so when the age increases.

5. Conclusion

In this paper, we answered the need for a clear description of the methodology that needs to be followed when designing an AS workplace using the mocap system and VR. The combined use of the mocap system and VR has, in fact, emerged as a breakthrough in AS workplace design, since it makes it possible to overcome the main limitations of the currently used AS workplace design procedures. For example, the combined use of the mocap system and VR does not require the large amount of time and resources needed by the traditional AS workplace design procedure (i.e. the construction of a physical workplace mockup) to optimise the AS workplace design, since several AS workplace designs can be easily and rapidly created and tested digitally. Moreover, the use of the mocap system and VR can also overcome the main limitation of computer-aided systems (i.e. ergonomic assessments not fully representative of the reality) since the mocap system allows the creation of a digital copy of the operator that is a truthful representation of the reality. However, prior to our work, it was still unclear how best to use them for AS workplace design in order to fully exploit their enormous potentialities, especially when aiming to maximise the AS workplace design considering both the productivity and the operators' wellbeing. The methodological framework developed herein solved this issue, providing five simple steps that can be followed when designing an AS workplace with the mocap system and VR. Moreover, the methodological framework makes it possible to consider the current labour market, since it provides the possibility to include ageing workers and their associated benefits and drawbacks in the AS workplace design procedure. Furthermore, the methodological framework allows both the productivity and operators' wellbeing to be maximised considering a holistic approach.

The validity of the methodological framework has been proved by means of a simple but representative case study. The methodological framework was successfully applied to the redesign of the AS workplace of a medium/high size pump. Specifically, the task assembly times were reduced by around 15%, and the ergonomic risks were also reduced from high to medium. Fundamental for these achievements was the detailed data analysis step described in the methodological framework, which made it possible to easily identify the tasks that were critical for the productivity and for the operators' wellbeing. Moreover, the possibility to design the workplace virtually halves the time spent on building the AS workplace (the digital AS workplace was built in approx. 5 h, while the physical one in approx. 10 h).

However, to prove its validity further, the methodological framework needs to be applied in other case studies, where real industrial applications, bigger sample sizes (meaning the number of operators considered) and the repetitiveness of the data are considered. Nevertheless, the potentialities of the methodological framework have been shown to be considerable, and they will be further investigated in the future in order to overcome the main limitations just described.

CRediT authorship contribution statement

Marco Simonetto: Investigation, Conceptualization, Writing – original draft, Software, Methodology. **Simone Arena:** Methodology, Conceptualization, Writing – review & editing. **Mirco Peron:** Conceptualization, Data curation, Formal analysis, Investigation, Methodology, Software, Writing – original draft.

Declaration of Competing Interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Appendix A

Here we will show how to include the age-based multipliers *f*_{strength} and *f*_{flextbility} in the ergonomic assessment. We will limit our demonstration to the REBA index, but the same procedure can be applied to any other ergonomic index containing allowable joint angle limits.

REBA is an ergonomic index used to evaluate the ergonomic risks of the whole body, assigning a score to the following body regions: neck, trunk, legs, upper and lower arms, and wrists. The former three are evaluated in the so-called body segment section A, while the latter three in the so-called body segment section B. The scores of the two body segments are modified considering additional adjustments and then combined to obtain the ergonomic risk. Specifically, the assessment of the body segment section A consists of six steps:

1. Locate neck position,

- 2. Locate trunk position,
- 3. Locate legs position,
- 4. Look-up posture score in Table A,
- 5. Add force/load score,
- 6. Score A,

similarly to the body segment section B:

- 7. Locate upper arm position,
- 8. Locate lower arm position,
- 9. Locate wrists position,
- 10. Look-up posture score in Table B,
- 11. Add coupling score,
- 12. Score B

The age-based multipliers need to be considered in Steps 1–3, 5, 7–9, specifically $f_{strength}$ in Step 5, while $f_{flextbility}$ in the others. Considering first $f_{strength}$, Step 5 in the traditional REBA corresponds to the following:

 $\begin{cases} ifload < 11lbs. = +0\\ ifload 11to 22lbs. = +1\\ ifload > 22lbs. = +2 \end{cases}$

Step 5 in the REBA, modified to include age considerations by introducing the age-based multiplier fstrength, corresponds to:

 $\begin{array}{l} load < (11 \bullet f_{strength}) lbs. = +0 \\ load (11 \bullet f_{strength}) to (22 \bullet f_{strength}) lbs. = +1 \\ load > (22 \bullet f_{strength}) lbs. = +2 \end{array}$

Similarly for $f_{flexibility}$. Considering as an example Step 1, the traditional REBA is:

```
\begin{array}{l} load < (11 \bullet f_{strength}) lbs. = +0 \\ load (11 \bullet f_{strength}) to (22 \bullet f_{strength}) lbs. = +1 \\ load > (22 \bullet f_{strength}) lbs. = +2 \end{array}
```

Modifying Step 1 to include age considerations through the age-based multiplier $f_{flexibility}$ leads to:

 $\begin{array}{l} neckangle < 0^{\circ} = +2\\ 0^{\circ} \leq neckangle \leq 10^{\circ} \cdot f_{flexibility} = +0\\ 10^{\circ} \cdot f_{flexibility} < neckangle \leq 20^{\circ} \cdot f_{flexibility} = +1\\ neckangle > 20^{\circ} \cdot f_{flexibility} = +2 \end{array}$

Similarly for Steps 2, 3 and 7-9.

Appendix B

In Table B1, the 25 different components constituting the pump used in the case study are gathered, while in Table B2 the different tasks are described.

Table B1

Components constituting the pump. Tank



Engine



Valve



Hydraulic cover



Long circular screw



Pressure gauge



Square nuts



Fan cover







Nut type 2



Electrical housing cover







Tank support



Small screws



Ноор



Metallic tube



Long shank screws



(continued on next page)

Table B1 (continued)



Table B2

Tasks description.	
Task number	Task description
1	Assembly of the pressure gauge on the metal tube
2	Assembly of the metallic disc on the electrical components
3	Assembly of the square gasket on the electrical components
4	Assembly of the metallic tube on the electrical components (using Long shank Phillips screws)
5	Assembly of the electrical components on the engine (using Phillips head screws)
6	Assembly of the electrical housing cover on the electrical components (using long shank screws)
7	Assembly of the internal grey ring on the engine
8	Assembly of the valve on the hydraulic cover
9	Assembly of the <i>rubber O-ring</i> on the engine
10	Assembly of the hydraulic cover (with the valve in it) on the engine, using the hoop (and the long circular screw and nut type 2)
11	Assembly of the fan cover on the engine (with small screws)
12	Assembly of the metallic tube on the hydraulic cover
13	Assembly of the caps on the metallic tube and on the hydraulic cover
14	Assembly of the tank on the tank support (with two of the square nuts)
15	Assembly of the pump base on the tank (with two of the square nuts)

Assembly of the metallic tube on the tank

Assembly of the engine on the pump base (with square screws)

Appendix C

16

17

Fig. C1 reports the REBA scores (average and distribution) for each task. To show the importance of considering the age of the operators in the ergonomic assessments, we carried out these assessments considering both the real age of the case-study operator (27 years, blue histograms) and a higher, fictitious age (60 years, orange histograms).

Fig. C2 reports the REBA scores (average and distribution) for each task considering both the real age of the case-study operator (27 years, blue histograms) and a higher, fictitious age (60 years, orange histograms).

Abubakar, M.I., Wang, Q., 2019. Key human factors and their effects on human centered assembly performance. Int. J. Ind. Ergon. 69, 48–57.

- Alessandri, G., Truxillo, D.M., Tisak, J., Fagnani, C., Borgogni, L., 2020. Withinindividual age-related trends, cycles, and event-driven changes in job performance: a career-span perspective. J. Bus. Psychol. 35 (5), 643–662.
- Arena, S., Roda, I., Chiacchio, F., 2021. Integrating modelling of maintenance policies within a stochastic hybrid automaton framework of dynamic reliability. Appl. Sci. 11 (5), 2300.
- Arena, S., Florian, E., Orrù, P.F., Sgarbossa, F., Zennaro, I., A novel Decision Support System for managing predictive maintenance strategies based on Machine Learning approaches, Safety Science, 2021, Article in press.
- Balderrama, C., Flores, J., Maldonado, A., 2015. The effects of age to meet the standard production rate. Work, 51(4), pp.827-837.Batraïa, O., Otto, A., Sgarbossa, F., Pesch, E., 2018. Future trends in management and
- Battaia, O., Otto, A., Sgarbossa, F., Pesch, E., 2018. Future trends in management and operation of assembly systems: from customized assembly systems to cyber-physical systems. Omega 78 (July), 1–4. https://doi.org/10.1016/j.omega.2018.01.010. Battini, D., Faccio, M., Persona, A., Sgarbossa, F., 2011. New methodological framework
- Battini, D., Faccio, M., Persona, A., Sgarbossa, F., 2011. New methodological framework to improve productivity and ergonomics in assembly system design. Int. J. Ind. Ergon. 41 (1), 30–42.
- Battini, D., Finco, S. and Sgarbossa, F., 2020. Human-Oriented Assembly Line Balancing and Sequencing Model in the Industry 4.0 Era. In Scheduling in Industry 4.0 and Cloud Manufacturing (pp. 141-165). Springer, Cham. Battini, D., Persona, A., Sgarbossa, F., 2014. Innovative real-time system to integrate
- Battini, D., Persona, A., Sgarbossa, F., 2014. Innovative real-time system to integrate ergonomic evaluations into warehouse design and management. Comput. Ind. Eng. 777, 1–10.
- Battini, D., Calzavara, M., Persona, A., Sgarbossa, F., Visentin, V. and Zennaro, I., 2018. Integrating mocap system and immersive reality for efficient human-centred workstation design. *IPAC-PaperSOLINE*, 51(11), pp.188-193.
- Bauer, W., Lorenz, D., 1985. New Methods of Ergonomic Work Design. In Toward the Factory of the Future. Springer, Berlin, Heidelberg, pp. 897–902.
- Bortolini, M., Faccio, M., Gamberi, M., Pilati, F., 2020. Motion Analysis System (MAS) for production and ergonomics assessment in the manufacturing processes. Comput. Ind. Eng. 139, 105485. https://doi.org/10.1016/j.cic.2018.10.046.
- Bortolini, M., Ferrari, E., Gamberi, M., Pilati, F., Faccio, M., 2017. Assembly system design in the Industry 4.0 era: a general framework. IFAC-PapersOnLine 50 (1), 5700–5705.
- Börsch-Supan, A., Weiss, M., 2016. Productivity and age: evidence from work teams at the assembly line. The Journal of the Economics of Ageing 7, 30–42.Bouma, H., 2013. Accommodating older people at work. Gerontechnology 11 (4), 489–492.
- Bures, M., Simon, M., 2015. Adaptation of production systems according to the conditions of ageing population. MM Sci. J. 2015 (02), 604–609.
- Burggräf, P., Kampker, A., Kreisköther, K., Adlon, T., Riegauf, A., Dorn, B., Casla, P., Marguglio, A. and Donvito, G., 2019, July. Enabling Smart Workplaces by Implementing an Adaptive Software Framework. In International Conference on Applied Human Factors and Ergonomics (pp. 116-127). Springer, Cham.
- Calzavara, M., Battini, D., Bogataj, D., Sgarbossa, F., Zennaro, I., 2020. Ageing workforce management in manufacturing systems: state of the art and future research agenda. Int. J. Prod. Res. 58 (3), 729–747.
- Caputo, F., Greco, A., Egidio, D.A., Notaro, I. and Spada, S., 2017, July. A preventive ergonomic approach based on virtual and immersive reality. In International Conference on Applied Human Factors and Ergonomics (pp. 3-15). Springer, Cham.

Confidence on Appreciational rectors and Egonomics (p. 9-19), optings., Came Caputo, F., Greco, A., Fera, M., Macchiaroli, R., 2019. Digital twins to enhance the integration of ergonomics in the workplace design. Int. J. Ind. Ergon. 71, 20–31.

- Cavatoria, M., Dipardo, M., 2009. Improving the ergonomics of the workplace to enhance productivity and safety. In Indoor Work Living Environ. Heal. Saf. Perform., chapter 8, pages 293–306.
 Cohen, Y., Naseraldin, H., Chaudhuri, A., Pilati, F., 2019. Assembly systems in Industry
- Cohen, Y., Naseraldin, H., Chaudhuri, A., Pilati, F., 2019. Assembly systems in Industry 4.0 era: a road map to understand Assembly 4.0. Int. J. Adv. Manufact. Technol. 105 (9), 4037–4054.
- Cote, M., Kenny, A., Dussetschleger, J., Farr, D., Chaurasia, A., Cherniack, M., 2014. Reference values for physical performance measures in the aging working population. Hum. Factors 56 (1), 228–242.
- Das, B., Sengupta, A.K., 1996. Industrial workstation design: a systematic ergonomics approach. Appl. Ergon. 27 (3), 157–163.
- Di Pardo, M., Riccio, A., Sessa, F., Naddeo, A. and Talamo, L., 2008. Methodology development for ergonomic analysis of work-cells in virtual environment (No. 2008-01-1481). SAE Technical Paper.
- Di Pasquale, V., Miranda, S., Neumann, W.P., 2020. Ageing and human-system errors in manufacturing: a scoping review. Int. J. Prod. Res. 58 (15), 4716–4740.
- Dolgui A., Sgarbossa F., Simonetto M., 2021. Design and management of assembly systems 4.0: systematic literature review and research agenda. International Journal of Production Research. DOI is: 10.1080/00207543.2021.1990433.
- Eswaramoorthi, M., John, M., Rajagopal, C.A., Prasad, P.S.S., Mohanram, P.V., 2010, Redesigning assembly stations using ergonomic methods as a lean tool. Work 35 (2), 231–240.
- Faccio, M., Gamberi, M., Piana, F. and Pilati, F., 2017. Motion analysis system for the ergonomic assessment of manufacturing and assembly manual activities. Proceedings of the Proceedings of XXII Summer School "Francesco Turco", Palermo, Italy, pp.13-15.
- Fager, P., Calzavara, M., Sgarbossa, F., 2020. Modelling time efficiency of cobotsupported kit preparation. Int. J. Adv. Manufact. Technol. 106 (5-6), 2227–2241.
- Feldmann, K., Junker, S., 2003. Development of new concepts and software tools for the optimization of manual assembly systems. CIRP Ann. 52 (1), 1–4.

Giacomin, J., 2014. What is human centred design? Des. J. 17 (4), 606-623.

- Göbel, C., Zwick, T., 2009. Age and productivity-Evidence from linked employer employee data. ZEW-Centre for European Economic Research Discussion Paper, (09-020).
- Hanson, T., Lindgren, P.Y., 2020. No country for old men? Increasing the retirement age in the Norwegian armed forces. Def. Peace Econ. 1–26.
- Holubek, R., Ružarovský, R., 2014. The methods for increasing of the efficiency in the intelligent assembly cell. In Applied Mechanics and Materials (Vol. 470, pp. 729-732). Trans Tech Publications Ltd.
- Ilmarinen, J.E., 2001. Aging workers. Occupational and environmental medicine, 58(8), pp.546-546.
- Jayaram, U., Jayaram, S., Shaikh, I., Kim, YoungJun, Palmer, C., 2006. Introducing quantitative analysis methods into virtual environments for real-time and continuous ergonomic evaluations. Comput. Ind. 57 (3), 283–296.
- Jones, M.K., Latreille, P.L., Sloane, P.J., Staneva, A.V., 2013. Work-related health risks in Europe: are older workers more vulnerable? Soc. Sci. Med. 88, 18–29. https://doi. org/10.1016/j.socscimed.2013.03.027.
- Kim, J., 2019. Aging Workforce, Firm Productivity and Labor Costs in Korea: Are Older Workers Costly to Firms? Asian Econ. J. 33 (2), 115–142.
- Kim, S.E., Junggi Hong PhD, A.T.C., 2013. Ergonomic interventions as a treatment and preventative tool for work-related musculoskeletal disorders. International Journal of Caring Sciences, 6(3), p.339.
- Kowalski-Trakofler, K.M., Steiner, L.J., Schwerha, D.J., 2005. Safety considerations for the aging workforce. Saf. Sci. 43 (10), 779–793.
- Landau, Kurt, Rademacher, Holger, Meschke, Herwig, Winter, Gabriele,

Schaub, Karlheinz, Grasmueck, Marc, Moelbert, Ingo, Sommer, Michael, Schulze, Jens, 2008. Musculoskeletal disorders in assembly jobs in the automotive industry with special reference to age management aspects. Int. J. Ind. Ergon. 38 (7-8), 561–576.

Launis, M., Vuori, M., Lehtelä, J., 1996. Who is the workplace designer?—Towards a collaborative mode of action. Int. J. Ind. Ergon. 17 (4), 331–341.

- Makris, Sotiris, Karagiannis, Panagiotis, Koukas, Spyridon, Matthaiakis, Aleksandros-Stereos, 2016. Augmented reality system for operator support in human-robot collaborative assembly. CIRP Annals – Manufact. Technol. CIRP 65 (1), 61–64. https://doi.org/10.1016/j.cirp.2016.04.038.Michalos, George, Karvouniari, Anna, Dimitropoulos, Nikolaos, Togias, Theodoros,
- Michalos, George, Karvouniari, Anna, Dimitropoulos, Nikolaos, Togias, Theodoros, Makris, Sotiris, 2018. Workplace analysis and design using virtual reality techniques. CIRP Ann. 67 (1), 141–144.
- Naddeo, A., Cappetti, N., Califano, R., Vallone, M., 2014. Manual assembly workstation redesign based on a new quantitative method for postural comfort evaluation. In Applied Mechanics and Materials (Vol. 459, pp. 368-379). Trans Tech Publications Ltd.
- Neubert, N., Bruder, R., Toledo, B., 2012. The charge of ergonomics-A model according to the influence of ergonomic workplace design for economical and efficient indicators of the automotive industry. Work 41 (Supplement 1), 4389–4395.
- OECD, 2015. OECD Employment Outlook 2015. OECD Publishing, Paris. Orrů, P.F., Zoccheddu, A., Sassu, L., Mattia, C., Cozza, R., Arena, S., 2020. Machine learning approach using MLP and SVM algorithms for the fault prediction of a centrifugal pump in the oil and gas industry. Sustainability 12 (11), 4776.
- Oyekan, John, Prabhu, Vinayak, Tiwari, Ashutosh, Baskaran, Vinubha, Burgess, Mark, Mcnally, Rob, 2017. Remote real-time collaboration through synchronous exchange of digitised human–workpiece interactions. Fut. Generat. Comput. Syst. 67, 83–93.
- Peng, Lu, Chan, Alan H.S., 2019. A meta-analysis of the relationship between ageing and occupational safety and health. Saf. Sci. 112, 162–172.
- Peng, L., Chan, A.H., 2020. Adjusting work conditions to meet the declined health and functional capacity of older construction workers in Hong Kong. Safety science 127, 104711.
- Pennathur, Arunkumar, Contreras, Luis Rene, Arcaute, Karina, Dowling, Winifred, 2003. Manual dexterity of older Mexican American adults: a cross-sectional pilot experimental investigation. Int. J. Ind. Ergon. 32 (6), 419–431.
- Peron, M., Sgarbossa, F., Strandhagen, J.O., 2020. Decision support model for implementing assistive technologies in assembly activities: a case study. Int. J. Prod. Res. 1–27.
- Peruzzini, Margherita, Carassai, Stefano, Pellicciari, Marcello, 2017. The benefits of human-centred design in industrial practices: re-design of workstations in pipe industry. Procedia Manuf. 11, 1247–1254.
- Prabhu, Vinayak Ashok, Song, Boyang, Thrower, John, Tiwari, Ashutosh, Webb, Phil, 2016. Digitisation of a moving assembly operation using multiple depth imaging sensors. Int. J. Adv. Manufact. Technol. 85 (1–4), 163–184.
- Porta, M., Arena, S., Fiori, V.L., Pau, M., Orrù, P.F., 2020. Assessment of ergonomic risk in industrial field: Analysis of the operators' body postures through wearable sensors. Proceedings of the XXV Summer School "Francesco Turco", 259899.
- Qin, Jin, Lin, Jia-Hua, Bucholz, Bryan, Xu, Xu, 2014. Shoulder muscle fatigue development in young and older female adults during a repetitive manual task. Ergonomics 57 (8), 1201–1212.

Realyvásquez-Vargas, A., Arredondo-Soto, K.C., Blanco-Fernandez, J., Sandoval-Quintanilla, J.D., Jiménez-Macías, E., García-Alcaraz, J.L., 2020. Work standardization and anthropometric workstation design as an integrated approach to sustainable workplaces in the manufacturing industry. Sustainability 12 (9), 3728.

- Riva, Giuseppe, 2002. Virtual reality for health care: the status of research. Cyberpsychol. Behav. 5 (3), 219–225.
- Roda, I., Arena, S., Macchi, M., Orrù, P.F., 2019, September. Total Cost of Ownership Driven Methodology for Predictive Maintenance Implementation in Industrial Plants. In IFIP International Conference on Advances in Production Management Systems (pp. 315-322). Springer, Cham.

- Roosaar, Liis, Masso, Jaan, Varblane, Urmas, 2019. Age-related productivity decrease in high-waged and low-waged employees. Int. J. Manpower 40 (6), 1151–1170. Roper, Kathy O., Yeh, Daniel C., 2007. Ergonomic solutions for an aging workforce.
- J. Facilities Manage. 5 (3), 172–178. Shephard, Roy J, 2000. Aging and productivity: some physiological issues. Int. J. Ind.
- Ergon. 25 (5), 535–545. Schwab, K., 2016. The Fourth Industrial Revolution. Crown Publishing Group, New York.
- Simonetto, M., Peron, M., Fragapane, G., Sgarbossa, F., 2020. In: October. Digital Assembly Assistance System in Industry 4.0 Era: A Case Study with Projected Augmented Reality. Springer, Singapore, pp. 644–651.
- Sišková, V., Dlabač, J., 2013. Environmental management system as the part of the methodology for optimization of assembly workplaces. International Journal of Mathematical Models and Methods in Applied Sciences. Strasser, H., 2018. The "art of Aging" From an Ergonomics Viewpoint - Wisdoms on age.
- Strasser, H., 2018. The "art of Aging" From an Ergonomics Viewpoint Wisdoms on age Occup. Ergon. 13 (S1), 1–24.
- Truxillo, D.M., Cadiz, D.M. and Hammer, L.B., 2015. Supporting the aging workforce: A review and recommendations for workplace intervention research.
- Udosen, U.J., 2007. Computer Aided Construction, Evaluation And Improvement Of A Workplace Assembly Layout Using CADWORK. In Advanced Materials Research (Vol. 18, pp. 353-358). Trans Tech Publications Ltd.
- United Nations, 2018. World Population Ageing 2017 [WWW Document]. URL https:// www.un.org/en/development/desa/population/theme/ageing/WPA2017.asp (accessed 11.2.20).

- Varianou-Mikellidou, Cleo, Boustras, Georgios, Nicolaidou, Olga, Dimopoulos, Christos, Anyfantis, Ioannis, Messios, Paris, 2020. Work-related factors and individual characteristics affecting work ability of different age groups. Saf. Sci. 128, 104755. https://doi.org/10.1016/j.ssci.2020.104755.
- Vignais, Nicolas, Miezal, Markus, Bleser, Gabriele, Mura, Katharina, Gorecky, Dominic, Marin, Frédéric, 2013. Innovative system for real-time ergonomic feedback in industrial manufacturing. Appl. Ergon. 44 (4), 566–574.
 Vosniakos, G.-C., Deville, J., Matsas, E., 2017. On immersive virtual environments for
- Vosniakos, G.-C., Deville, J., Matsas, E., 2017. On immersive virtual environments for assessing human-driven assembly of large mechanical parts. Procedia Manuf. 11, 1263–1270.
- Wang, K., 2016. Intelligent predictive maintenance (IPdM) system-industry 4.0 scenario. WIT Trans. Eng. Sci. 113, 259–268.
- Warnecke, H.J., Haller, E., Warnecke, H.J., 1982. Computer-aided planning of manual assembly workplaces. CIRP Ann. 31 (1), 371–375.
- Wolf, M., Ramsauer, C., 2018, August. Towards an Age-Differentiated Assessment of Physical Work Strain. In Congress of the International Ergonomics Association (pp. 189-205). Springer, Cham.
- Zieschang, H., Freiberg, S., 2011, May. ERGONOMIC DESIGN OF
- WORKPLACES_PREVENTION FOR ALL GENERATIONS. In Contemporary Ergonomics and Human Factors 2011: Proceedings of the international conference on Ergonomics & Human Factors 2011, Stoke Rochford, Lincolnshire, 12-14 April 2011 (p. 299). CRC Press.
- Zha, X.F, Lim, S.Y.E, 2003. Intelligent design and planning of manual assembly workstations: a neuro-fuzzy approach. Comput. Ind. Eng. 44 (4), 611–632.

Paper 6

Simonetto, M., and Saporiti, N., (under review). Traditional Kanban, e-Kanban or Digital Twin: a Decision Support System for material management solution. Industrial Management & Data Systems.

This paper is under review and is not included in NTNU Open



ISBN 978-82-326-5608-0 (printed ver.) ISBN 978-82-326-6498-6 (electronic ver.) ISSN 1503-8181 (printed ver.) ISSN 2703-8084 (online ver.)

